Exploring Teachers' Perceptions on History Public Examinations Performance in Five High Schools in Maseru



 \mathbf{BY}

KAIZER FELIX NONYANA -201703391

Submitted in fulfilment for the Master of Arts in Education (History Education)

Supervisor: Dr. Walter Sengai

August 2024

DECLARATION

I, Kaizer Felix Nonyana, do hereby declare that this dissertation is my own original work and it has never been previously submitted to any other academic institution. I, further, declare that the work presented will not be submitted to any other academic institution for consideration for a Masters' degree award. Aside from that, no information from other academics is included in the dissertation, and where it is, appropriate citation is supplied.

Signature	KeNbryperk	
Date	12/09/2024	

SUPERVISOR'S STATEMENT

This dissertation has been submitted with my approval.

Signature	MSo
Date	12/09/24

AKNOWLEDGEMENTS

I would like to express my sincere gratitude and appreciation to the following people for their support and contribution to the completion of this research:

Above all, I would like to express my heartfelt gratitude to Almighty God for your protection, grace, mercy, loving, caring, and faithfulness in providing me with the strength, brainpower, and motivation essential to complete this research. Dear God, I give you thanks and appreciation for all of your blessings, knowledge, and opportunity to complete this dissertation. You were always by my side during the academic difficulties and tribulations. You provided me with deeper, steadfast, unlimited grace, love, and mercy, and I am extremely thankful for this support from you, dear God. My whole idea of completing this dissertation would not have been feasible without you, God.

I would like to express my sincere gratitude to my supervisor, Dr. Walter Sengai, for your invaluable expertise, guidance, and contributions to the quality and completion of this work. I thank you for your constructive comments that gave this dissertation its present form, and your guidance and outstanding mentorship have enhanced my academic growth. I will always acknowledge your dual role as not only a supervisor but also as a caring and loving parent. I acknowledge your perseverance and continuous support throughout the entire study, and I have learned a lot from you.

Thirdly, I would like to articulate my special gratitude to Professor 'Maboleba Kolobe, a professional editor in the Department of English from Faculty of Humanities who worked tirelessly to edit this dissertation for language. Correspondingly, I extend my gratitude to Dr. Tawanda Mataka and Dr. Musa Adekunle Ayanwale for the deeper research knowledge you have imparted to me and your guidance and motivation.

I also extend my gratitude to my pastors, Pastor Rapelang Mpheele and 'Matlhohonolofatso Mpheele, for your constant support with your prayers, advice, encouragement, emotional support, and material support you provided me to complete my dissertation. I also appreciate your consistent and wholehearted parenting for me. I further extend my gratitude to my beloved parents, Mr. Nonyana and Ms. 'Maseinama Nonyana, for their guidance, financial and emotional support

in order for me to complete this work. I also acknowledge your encouragement and your sacrifice for me to allow me to complete this work.

Furthermore, I extend my gratitude to my two best friends, Tsietsi Khopholi and Tsibela Selemela, for their firm financial and emotional support during financial crises and hard times. You have been with me with your encouragement while I was about to give up on this work due to difficulties. I also extend my gratitude to Mr. Tumisang Damasane, Mphekeleli Besele, Atang Thulo, and Thabang Khalema, whom we used to discuss academic matters that influenced my perceptions and beliefs about research.

I would also like to convey my heartfelt gratitude to all my 2022 classmates who supported me through their motivations, emotional support and advice. My special gratitude goes to my colleague, Maletsema Mosole, who lent me her laptop to complete my dissertation because of my broken laptop when I was about to complete my dissertation.

I, further, extend my gratitude to the administration and staff of the Department of Education for ensuring the smooth completion of this dissertation through your contributions in seminar presentations and the provision of an introductory letter for me to generate data. Finally, my gratitude extends to my participants for your willingness to participate in this study by sharing your knowledge, opinions, and experiences. Thank you for your persistent participation which made this dissertation to be completed. Therefore, I am grateful to each and everyone of you for helping me to finish this dissertation. I pray that you all continue to receive many blessings from the almighty God.

ABSTRACT

The main purpose of this study was to explore teachers' perceptions on performance in History public examinations. Moreover, the study intended to explore the views of teachers on the status of History in the national curriculum, considering learners' performance in History public examinations. The study was further aimed at investigating how History teachers perceive the factors influencing the performance of History learners in public examinations. Therefore, understanding the status of History in the national curriculum and factors influencing History learners in public examinations can help raise awareness among educational stakeholders, including History teachers, on how to deal with the unacceptable status of History and the negative factors influencing the performance of History learners. The study was to further explore teachers' suggestions on mitigation of History learners' performance in public examinations. In order to achieve the main purpose of the study, the research employed the interpretivism paradigm as the theoretical lens of this study. Further, the study employed a qualitative study and a case study design. Five schools and seven History teachers were selected using the purposive sampling technique. Open-ended questionnaires, field notes, and semi-structured interviews were used to generate data. The study further used both manual inductive and deductive thematic analysis to analyse data. The findings of this study uncovered that History is performed poorly in public examinations. The findings further revealed that despite the benefits of History in the national curriculum, this poor performance may result in an unacceptable status for History, which may lead to the subject being completely removed from the syllabus. The semi-structured interviews discovered the following major factors influencing the performance of History teachers: improper training of History teachers on how to teach, unclear communication between Examination Council of Lesotho and History teachers, and incompetence of History teachers. As a result, the study recommends training in-service History teachers on how to teach, mark, assess learners, and use technology in the teaching of History. The study also recommends that teachers should give learners more frequent assessments and immediate feedback and correct them to enhance their performance in History public examinations.

LIST OF ABBREVIATIONS

CAP Curriculum and Assessment Policy

COSC Cambridge Overseas School Certificate

ECOL Examination Council of Lesotho

ICT Information, Communication and Technology

JC Junior Certificate

MOET Ministry of Education and Training

NCDC National Curriculum Development Centre

PSS Personal, Spiritual and Social

Table of Contents

DECLARATION	i
SUPERVISOR'S STATEMENT	ii
AKNOWLEDGEMENTS	iii
ABSTRACT	v
LIST OF ABBREVIATIONS	vi
LIST OF TABLES	xi
CHAPTER ONE	1
THE PROBLEM AND ITS SETTING	1
1.1 Introduction	1
1.2 Background to the study	1
1.3 Statement of the problem	6
1.4 Research objectives	7
1.5 Research questions	7
1.6 Significance of the study	8
1.7 Definition of terms	9
1. 8 Delimitation of the study	10
1.9 Limitation of the study	10
1.10 Theoretical framework	11
1.11 Chapter outline	12
CHAPTER TWO	13
LITERATURE REVIEW	13
2.1 Introduction	13
2.2 The status of History in the national curriculum	13
2.3 Factors influencing performance of History learners in public examinations	15
2.3.1 Teachers' teaching experience	18
2.3.2 School type	19
2.3.3 Teachers' professional qualifications	20
2.3.4 Teachers' mastery of subject matter	20
2.3.5 Content coverage	21
2.3. 6 Revision materials	21

	2.3.7 Lesson preparation	22
	2.3.8 Coaching in the teaching of History	24
	2.3.9 Drilling in the teaching of History	25
	2.3.10 Relationship with the principals, other teachers and learners	26
	2.3.11 Assessment methods	27
	2.3.12 English as a medium of instruction	28
	2.3.13 Teaching resources	29
	2.3.13.2 Library resources	30
	2.3.13.3 Technology	30
	2.14 Computer laboratories	31
	2.3.15 Motivation of teachers and learners	32
	2.3.16 Support from school management	33
	2.3.17 Family job types and income	34
	2.3.18 Examination system	35
	2.3.19 Curriculum alignment	35
	2.3.20 Health of learners in the learning of History	36
	2.3.21 The attitude and behaviour of History teachers	36
	2.3.22 Attitude and behaviour of History learners	37
	2.3.23 Congested classrooms	38
	2.4 Mitigation of History learners' performance in public examinations	39
	2.5 Empirical Review	44
	2.3.1 Justification of gap for this study	46
	2.4 Conclusion	47
СН	APTER THREE	48
RE	SEARCH METHODOLOGY	48
	3.1 Introduction	48
	3.2 Research Paradigm	48
	3.3 Research approach	49
	3.4 Research design	50
	3.5 Population and sampling procedure	51
	3.6 Research instruments	51

3.6.1 Open-ended questionnaires	52
3.6.2 Semi-structured interviews	52
3.6.3 The field notes	54
3.7 Data analysis	55
3.8 Trustworthiness of the research	58
3.8.1 Credibility	58
3.8.2 Transferability	59
3.8.3 Dependability	60
3.8.4 Confirmability	60
3. 9 Research ethics	61
3.9.1 Voluntary participation	62
3.9.2 Informed consent	62
3.9.3 Anonymity	62
3.9.4 Confidentiality	63
3.9.5 Potential for harm	63
3.9.6 Results communication	63
3.10 Conclusion	64
CHAPTER FOUR	65
DATA PRESENTATION AND ANALYSIS	65
4.1 Introduction	65
4.2 Demographic profile of participants	65
4.3 Presentation of findings on the objectives	67
4.4 Data presentation and analysis on open-ended questionnaires	68
4.4.1 Status of History in the national curriculum of Lesotho	68
4.4.2 Benefits of History in national curriculum	69
4.4.3 Current number of History learners	70
4.4.4 General performance of History learners in public examinations of 2023	72
4.4.5 The reasons for the poor performance of History learners in public examinations	74
4.5 Data presentation and analysis on semi-structured interviews	78
4.5.1 Motivation of History teachers in the teaching of History	78
4.5.2 Relationship with principals, other teachers and learners	80

4.5.3 Lesson preparation	82
4.5.4 Use of teaching aids	84
4.5.5 Use of Teaching Strategies	85
4.5.6 Assessment methods	87
4.5.7 Immediate feedback	89
4.5.8 Correction from feedback	91
4.5.9 Challenges encountered by History learners at school and in learning	92
4.5.10 Mitigation on performance of History learners in public examinations	94
4.6 Conclusion	96
CHAPTER FIVE	98
SUMMARY OF FINDINGS, DISCUSSIONS, CONCLUSIONS AND RECOMMENDATIONS	98
5.1 Introduction	98
5.2 Summary of the findings	98
5.3 Discussion of findings	100
5.3.1 Status of History in national curriculum	100
5.3.1.1 The reasons for the poor performance of in public examinations	101
5.3.2 Factors influencing performance of History learners in public examinations	102
5.3.3 Use of teaching aids and resources in the teaching of History	104
5.3.4 Use of teaching strategies in the teaching of History	104
5.4 Conclusions	108
5.5 Limitations of the study	109
5.6 Implication of the study	109
5.7 Recommendations	110
5.8 Recommendation for future research	110
REFERENCES	112
APPENDIX A	137
APPENDIX B	139
APPENDIX C	140

LIST OF TABLES

Table 1: Demographic Profile of Participants	66
Table 2:Benefits of History in National Curriculum	69
Table 3:Current Number of History learners	70
Table 4:General Performance of History learners in their Public Examinations of 2023	72
Table 5:Motivation of History teachers in the Teaching of History	78
Table 6:Relationships with Principals, Other Teachers and Learners	80
Table 7:Lesson Planning and Scheming	82
Table 8:Use of Teaching Aids	84
Table 9:Use of Teaching Strategies	85
Table 10:Assignments and Weekly Tests	87
Table 11:Immediate Feedback on Assignments and Weekly Tests	89
Table 12:Ways in which Corrections from Feedback can Improve the Performance of History learners	s91
Table 13:Challenges Encountered by History learners at School and in Learning	92
Table 14:Solutions to the Challenges	94

CHAPTER ONE

THE PROBLEM AND ITS SETTING

1.1 Introduction

The performance of learners in History public examinations is a crucial aspect to be considered by educational stakeholders. The aim of this chapter, which functions as the introduction to this study, is to give a justification for the purpose of the study. Therefore, the main purpose of this study is to explore teachers' perceptions on performance in public examinations in three high schools in Maseru. In order to achieve the purpose of the study, this chapter consists of background to the study, a statement of the problem, research objectives, and research questions. The study further consists of the significance of the study, definition of terms, delimitation of the study, limitations of the study, theoretical framework, and chapter outline. Below is the background to this study.

1.2 Background to the study

The academic performance of learners in a certain discipline is very important because it may determine their academic success or failure in that particular discipline (Narad & Bilkees, 2016). More often than not, academic performance may either be satisfactory or unsatisfactory, depending on the ability of an individual learner. According to Singh (2021), satisfactory academic performance is when learners obtain higher grades in examination scores, whereas unsatisfactory performance can be described as when learners obtain lower grades in examination scores.

Apart from that, satisfactory academic performance may be defined as the high levels of academic achievements and successful completion of learning objectives (PubGenius Inc., 2024). On the contrary, Al-Zoubi (2015) defines unsatisfactory academic performance as a performance that falls below the expected standard, as adapted by the examiner and other educational stakeholders. In assertion, Drugs.com (2024) agrees that unsatisfactory academic performance occurs when the grades of learners steadily and rapidly dwindle over time.

Based on the findings of Adedoyin (2013), it can be noted that satisfactory academic performance acts as the foundation of learners' careers and may determine their successful development. As identified by Narad and Abdullah (2016), satisfactory academic performance of learners signifies

their enhanced career visions and a protected future. In alignment with this, Singh (2021) asserts that the satisfactory academic performance of learners is crucial for securing good jobs, a better career, and a quality life. If that is the case, satisfactory public examination performance can provide learners with opportunities to achieve professional and economic success.

However, Kaur (2019) accentuates that the professional success of learners is largely attributed to the continuous mediation of teachers, who are key factors in their learners' academic achievement. In concurrence with Kaur, Johnson (2017) affirms that teachers play a crucial role in motivating learners and creating a supportive environment that enhances their learning outcomes. In other words, it would appear that the success or failure of learners is attributed to their teachers, who act as curriculum implementers and facilitators during the teaching and learning process. Therefore, it is in this regard that History teachers may have better perceptions of the performance of their learners in public examinations than any other stakeholders.

Above all, it is important to define "perception, academic performance," and "public examinations" as essential terms used in this study. According to Amodu (2006), perception refers to the process of gathering and interpreting information from the world around us. Similarly, Keengwe et al. (2016) define perception as the ability of a human being to understand or interpret information to add meaning. In congruence with McDonald (2021), the study takes perception as an individual or group's unique way of understanding a phenomenon, which involves processing stimuli, memories, and experiences. Further, StudySmarter (2024) defines perception as the process of selecting, organising, interpreting, becoming aware, and producing stimuli into a meaningful and coherent worldview. Seemingly, perception is the process of understanding or interpreting the world around us, particularly in the academic performance of History learners on public examinations in this case.

Other than that, Lamas (2015) defines academic performance as the measure of the reflective and reactive abilities of learners, indicating their learning outcomes based on specific course objectives. Furthermore, the Education Policy on Ballotpedia (2023) adds that academic performance refers to the evaluation of the achievement of learners in various academic subjects. With that in mind, the literature suggests that academic performance can be evaluated through grade point average (GPA), standardised tests, and the high school graduation rate. Among other things, the Asia Development Bank (2017) affirms that academic performance can be evaluated

through classroom tests, national assessments, and public examinations, as defined in detail in the following paragraph.

According to Bernard (2009), public examinations may be defined as high-stakes standardised tests used by external agencies such as the Examination Council of Lesotho, in the context of Lesotho, to select learners for higher education. In support, Rind and Mari (2019) assert that public examinations refer to the systems that are used by the government universally in order to evaluate the abilities of learners for the next level of education. Therefore, it would appear that the above authors defined public examinations as summative assessments written by learners for their certification from one level of education to the next level of education, which in this case would be from high school (secondary) level to tertiary level.

Taking that into consideration, the Ministry of Education and Training (MOET), through Lesotho Curriculum and Assessment Policy (CAP) and Lesotho Basic Education Curriculum Policy (LBECP), mandates formative assessment (diagnostic and continuous), monitoring educational progress, and summative assessment (selection and certification) methods to evaluate educational and curriculum achievements (MOET, 2009; MOET, 2021). Fundamentally, Ismail et al. (2022) emphasise the importance of formative assessment, such as quarterly tests, in determining the success of summative assessment (a public examination) in selecting learners for higher education. As a result, it would seem that public examinations of learners are very important in determining their academic and professional progression.

Moreover, the 2009 Curriculum and Assessment policy mandates summative assessments at grades 10 (Junior Certificate) and 12 (Senior Secondary), respectively, but this does not happen as planned because the current public examinations are conducted once at grade 11. Bearing that in mind, Limene (2018) suggests that reducing written public examinations for learners may negatively impact their academic performance. On the contrary, the Ministry of Education and Training (2021) through LBECP recommends a rescheduled public examination for learners at the end of grade 12 due to CAP challenges.

What is important, Adedoyin (2013) emphasises the benefits of public examinations, including selection of suitable candidates, control over school curriculum, and provision of certificates after program completion. Taking that into consideration, in high schools, learners are taught different subjects, which may assist them in selecting suitable careers based on their interests (Lamas, 2015).

Aiming to achieve its objectives, the Ministry of Education and Training, through its curriculum frameworks (CAP and LBECP), has included History as an elective subject in the curriculum under Personal, Spiritual, and Social (PSS) (Social Sciences), where learners can make their choice (MOET, 2009; MOET, 2021).

On the face of it, throughout adolescence and adulthood, learners consistently engage in societal and national activities. A typical example is one of the 18-year-old and above youths who are allowed to participate in local government and general elections in Lesotho. This assertion is supported by the National Assembly Electoral Act of 2011, section 5, subsection (1) which outlines that individuals aged 18 and above in Lesotho are eligible to register for elections and vote (Government of Lesotho, 2011). In the opinion of Lamas (2015), it is at these stages where learners have the right to express their views about their society and the country they live in. Keeping that in mind, teaching History as part of the high school curriculum may be very crucial during these periods.

In conformity with Julina (2020), it should be noted that learners are future leaders who need to be instilled with knowledge to be responsible, active, participatory, and well-informed decision-makers in the society they live in and their entire nation. For that reason, it appears that History is a suitable subject that may impart learners with such enlightening and emancipatory knowledge. On that account, Sabbowa and Ng'ambi (2020) highlight that to impart this knowledge to learners in the 21st century, effective teaching and learning must be practised.

However, it appears that the performance of History in public examinations is a global issue that requires global attention. The problem with the subject is that the majority of schools and learners perform it unsatisfactorily (Ntaben, 2006; Fru, 2015). Evidently, studies conducted in Asia, Europe, America, and Africa show poor performance in History due to negative attitudes towards the subject among learners, a lack of resources, and rote learning (Nyamwembe, Ondigi, & Kiio, 2013; Chong & Yeo, 2016; Samuelsson, 2018; Juliana, 2020). Contextually, the Examination Council of Lesotho (ECOL) reported that from 2020 to 2023, most schools and learners performed below par in public examinations in the History subject (ECOL, 2020; ECOL, 2021; ECOL, 2022; ECOL; ECOL, 2023).

As an extra, the Examination Council of Lesotho's statistics from 2020 to 2023 show a consistent decline in the performance of high school History learners on Lesotho's public examinations

(ECOL, 2020; ECOL, 2021; ECOL, 2022; ECOL, 2023). For instance, Examination Council of Lesotho (2022) examination results reveal that out of 60 History schools, only three performed well, with the majority performing unacceptably. On account of that, the Examination Council of Lesotho (2022) reported that the majority of schools in Lesotho struggled with History, with learners obtaining lower grades like F, G, and U. However, these schools with unsatisfactory performance have few learners achieving higher grades (A* to C) and average grades (D to E) (ECOL, 2022). This implies that, even though most of the schools perform unsatisfactorily on History public examinations, some individual learners in these schools excel.

Another essential point is that the unsatisfactory performance of History learners on high school public examinations is reportedly outpacing their satisfactory performance (ECOL, 2022; ECOL, 2023). This low performance among History learners has raised concerns among educational stakeholders, who view the subject as irrelevant and unsuitable for the 21st century (Okechukwu, 2023). According to Naicker (2020), some people indicate that learning History may be a waste of time as it is not useful in everyday life. By appearance, these raised concerns about History public examinations have led to public outcry, with some parents even discouraging their children from studying it due to poor performance (Ntabeni, 2006; Fru, 2015a).

Several similar studies on public examinations performance of high school learners were conducted. First, Fisher (2023) conducted a study in 2023 in America on the perceptions of teachers about learners' low performance on the Advancement Placement English Language and Composition exams. Secondly, Lekhetho (2013) conducted a study on the stakeholder's perspectives on strategies that can improve learners' low performance. Thirdly, Lebata (2014) conducted a study on the low performance of learners in biology. Additionally, Lekhetho (2021) conducted another study on factors that influence the performance of high-achieving secondary schools in Lesotho.

Furthermore, Ntabeni (2006) conducted a study on the status of History teaching, learning, and examination results in Lesotho, which focused on the low performance of external learners (both the Cambridge Oversees School Certificate (COSC) and the Junior Certificate (JC). Besides, Ntabeni (2007) conducted another study on History teaching, learning and Junior Certificate examination results in Lesotho, 2000–2006. Observably, most of the studies conducted focused on

non-historical subjects, except Ntabeni's 2006 and 2007 studies, which identified essay writing skills and inadequate teacher training as primary factors causing poor performance.

Fundamentally, the public examinations written by external learners are considered to be a reflection of quarterly examinations as they measure learning improved through quarterly examinations (Ismail et al., 2022). Therefore, performance in public examinations appears to be influenced by quarterly examination performance factors (Lekhetho, 2021). For that matter, there is a need for the educational development stakeholders to address quarterly examination factors affecting the performance of History learners in their public examinations. Nonetheless, it seems that there is a limited study conducted specifically to hear from the voices of teachers about the status of History on performance in public examinations. As a result, it is against this background that the study aims to explore teachers' perceptions on performance in History public examinations in three high schools in Maseru district.

1.3 Statement of the problem

History as a subject in Lesotho is seemingly performed poorly among other subjects across the entire nation. In support, the Examination Council of Lesotho reports that the performance of History learners in public examinations from 2020 to 2023 fluctuates annually (ECOL, 2020; ECOL, 2021; ECOL, 2022; ECOL, 2023). Resultantly, the issue with the frequent change seems to be falling below average performance, causing a shift from one level of low performance to another. In appearance, the overall performance of History learners in the public examinations in 2023 was significantly lower than in 2022, 2021, and 2020. This assertion is proven by the History Examiner's report of 2022, which showed a decrease in the performance of both History papers 1 and 2 compared to the other aforementioned previous years (ECOL, 2022).

Additionally, the Minister of Education and Training's speech released on 17th January 2024, on Lesotho General Certificate in Secondary Education results revealed that History's performance was unsatisfactory, ranking among well-performed subjects like Sesotho (58%), English (26%), Physical Science (22%), and Mathematics (11.2%) (Lesotho News Agency, 2024). With that being said, the lower public examination performance of History learners appears to be an escalating issue that warrants further attention. For that reason, failure to consider it may put History subject at risk of being removed from the syllabus (Fru, 2015).

However, it seems that there are limited studies on the perceptions of teachers on why the majority of schools and grade 11 History learners perform poorly in public examinations. On the grounds of this, this study seeks to understand teachers' perspectives on why some schools and learners perform better in History, while most perform unsatisfactorily in the same subject. This knowledge gap on the performance in public examinations of History learners in Lesotho's high schools presents a challenge for educational stakeholders responsible for improving and maintaining the public examination performance of grade 11 History learners.

Therefore, it is against this problem that the current study intends to explore teachers' perceptions on performance in History public examinations in three high schools in Maseru. The study further seeks to understand and inform policies and strategies to improve the public examination performance of grade 11 History learners by mitigating negative factors and maintaining positive ones.

1.4 Research objectives

The main research objective: To explore teachers' perceptions on performance in History public examinations in three high schools in Maseru.

Sub-research objectives

- 1. To explore the views of teachers on the status of History in national curriculum considering learners' performance in public examinations.
- 2. To investigate how History teachers, perceive the factors influencing performance of History learners in public examinations.
- 3. To explore History teachers' suggestions on mitigation of History learners' performance in public examinations.

1.5 Research questions

The main research question: What are teachers' perceptions on performance in History public examinations in three high schools in Maseru?

Sub-research questions

1. What are the History teachers' views on the status of History in national curriculum considering learners' performance in public examinations?

- 2. How do History teachers perceive the factors influencing performance of History learners in public examinations?
- 3. What are History teachers' suggestions on mitigation of History learners' performance in public examinations?

1.6 Significance of the study

The study may be significant to the school management for the introduction of new teaching and learning policies and the adjustment of the existing ones. Importantly, these policies may promote learners' performance on History public examinations. In addition to this, the findings of the study may be significant to the teachers, as they may help them improve their teaching and learning strategies as well as their content knowledge. Over and above, the study may be significant to the Ministry of Education and Training for improving the education system of Lesotho to meet the current needs of History learners. This may not only benefit the country but its citizens by paying more attention to History subject which may be very important in moulding the future leaders of this country to be effective leaders who may make informed decisions.

Apart from that, with the help of the findings from this study, the Ministry of Education and Training may improve the training of qualified teachers through several workshops. The study may similarly be significant to the National University of Lesotho and the Lesotho College of Education, which are the tertiaries that train student teachers in Lesotho. For this reason, with the aid of this study, these tertiaries may introduce new policies and adjust the existing ones to improve the training of trainee teachers, which may in turn improve their pedagogic content knowledge.

Besides, this study may contribute to existing knowledge, and the findings from this study may also be significant to scholars who may use this study as a reference. Considering that, the study may contribute to the existing knowledge on the perceptions of teachers on History public examination performance and factors that may affect History learners' public examination performance. The study may also be significant to the National Curriculum Development Centre (NCDC) for the amendment of History curriculum policy where necessary for the benefit of the learners. With that in mind, the recommendations from this study may enlighten the abovementioned stakeholders about the importance of considering the public examination performance of grade 11 History learners.

1.7 Definition of terms

Academic performance: This may be defined as the evaluation of the achievement of learners in different academic subjects.

Curriculum: It is defined as a plan guided and directed by the school for selecting content and organising learning experiences for the purpose of changing and developing the behaviours and understandings of learners.

Motivation: It is a process that stimulates the individuals to actions which may help them to accomplish certain goals.

Demotivation: It is the state of mind characterised by a lack of interest in enjoyment in the process of teaching and learning.

Grade: It refers to the measurement of learners' performance in public examinations or tests.

Examination score: This refers to the numerical representation of learners' performance on an examinations or tests.

Expectancy: It is the belief that an effort leads to better performance.

Examination Council of Lesotho: It may be defined as a body that is responsible for assessing learners' abilities in public examinations for their certification from level of knowledge to another level of knowledge.

Instrumentality: It refers to the positive value of reward for acceptable performance.

History: It is a socials science subject which focuses on the study of change and development over time and it covers all aspects of human society.

High school: It refers to the school that offers secondary education to learners.

Perception: It refers to the ability of human beings to interpret information from the world around us to add meaning.

Public examinations: It may be defined as summative assessments written by learners for their certification from one level of education to the next level of education.

Quality education: It is the kind of education that provides learners with abilities to become economically productive, develop sustainable livelihoods, contribute to peaceful and democratic societies and enhance individual well-being.

Satisfactory academic performance: It is when learners obtain higher grades in examination scores.

Unsatisfactory academic performance: It is the performance that falls below the standard.

1. 8 Delimitation of the study

This study was conducted in five high schools in Maseru district of Lesotho. Two participants were selected from the first school, one participant was selected from the second school, another participant was selected from the third school, the other two participants were selected from the fourth school and a remaining participant was from the fifth school. The scope of the study was delimited to the History schools that were underperforming and well performing in the last four years. Among these four schools, one school performed well in 2022 and one school improved its History performance in 2023 as compared to other previous years. However, performance was not as good as expected according to the Examination Council of Lesotho and the Ministry of Education and Training standards.

1.9 Limitation of the study

The fact that the study was based on the performance in History public examinations, it was difficult to access the learners to get their views. That is why the study ultimately focused on the teachers' perceptions on performance in History public examinations. However, in future, the researcher will conduct a study on the learners' perceptions on performance in History internal examinations so that History learners' views will be accessed. Apart from that, the findings of this study revealed that the evolving marking by Examination Council of Lesotho plays a very significant role in poor performance of History learners in their public examinations due to the fact that History teachers are not even trained on how to mark History. Having considered that, another limitation was the difficulty in evaluating the effectiveness of History teachers in marking the subject hence there will be another future study to be conducted on this matter.

1.10 Theoretical framework

The study is underpinned by the Expectancy Theory (Process Theory), which is also a motivational theory. The theory was developed by Victor Vroom in 1964 with the aim of analysing and interpreting individual behaviour in learning, decision-making, attitude formation, and behaviour (Vroom, 1964; Lokman et al., 2022). Based on the ideas of Day (2020), this theory stresses that individuals possess a diverse array of goals, which can be fuelled by specific expectations. The theory was used in this study to explore whether History teachers have certain expectations regarding learners' performance in public examinations.

Basically, the Expectancy Theory consists of three principles: expectancy, instrumentality, and valence, which emphasise the positive value of reward for acceptable performance (Vroom, 1964). By definition, expectancy is the belief that an effort leads to better performance, while instrumentality is the trust that achieving good performance will lead to the expected outcome (Rehman, Sehar, & Afzal, 2019). On the other side, valence is defined as the unique value one places on the expected outcome (Day, 2020). Considering that, the suitability of this theory in this study may be beneficial for History teachers, as diligent efforts may enhance the performance of grade 11 History learners in their public examinations.

Moreover, the Expectancy Theory suggests that people's behaviour is primarily driven by their willingness to choose from a variety of options (Lokman et al., 2022). In addition to that, Rehman, Sehar, and Afzal (2019) emphasise that an individual will always select the behaviour that may maximise their satisfaction and minimise their stress. On the face of it, good performance by History learners in their public examinations can satisfy their teachers, while poor performance can dissatisfy them and cause stress among teachers. On the basis of that, motivation is one of the crucial aspects that may boost teachers' satisfaction and reduce their stress levels.

Therefore, the Expectancy Theory was used in this study to understand whether History teachers are motivated or demotivated to work hard in order to produce quality public examination results for grade 11 History learners. The theory was further helped the researcher explore the driving forces behind motivation and demotivation in individual teachers. The theory was used to assess the effectiveness of History teachers in selecting optimal opportunities and teaching methods to enhance the performance of grade 11 History learners in their public examinations (Day, 2020).

In essence, History teachers with high motivation are dedicated to imparting their learners with the necessary knowledge for their successful performance in public examinations. Another essential point is that History teachers who are motivated can significantly impact the motivation of their learners in the process of teaching and learning (Mhonyiwa, 2014). In the same way, motivated learners may exert effort to excel in their public examinations. In the similar manner, Rehnman, Sehar, and Afzal (2019) emphasise the importance of motivated learners for successful learning. On the other hand, low-motivated or demotivated teachers may not work hard, leading to unsatisfactory performance, hence the importance of the Expectancy Theory.

1.11 Chapter outline

The research is divided into five chapters being chapters one, two, three, four and five as well as the references. Chapter one provides background to the study, statement of the problem, research questions, research objectives, significance of the study, definition of terms, delimitation of the study, and limitation of the study, theoretical framework which explains the theory that underpin this study as well as chapter outline. The chapter two further presents the themes and empirical review of relevant literature and the knowledge gap to be filled by this study.

Chapter three includes methodology specifically paradigm of the study, research approach, design and instruments. The chapter further consists of data analysis technique, sampling procedure, trustworthiness of the research as well as ethical considerations. Chapter four consists of data presentation and analysis. In this chapter, data is presented through the use of tables. Further, chapter five includes summary of findings, discussions, conclusions and recommendations. Lastly, references of the cited literature are provided at the end.

CHAPTER TWO LITERATURE REVIEW

2.1 Introduction

The chapter presents a review of the related literature on the main themes derived from the research objectives and research questions of the study. These are the status of History in the national curriculum of Lesotho, factors influencing performance of History learners in public examinations and mitigation of History learners' performance in public examinations. Equally important, the researcher reviewed related studies that are relevant to this current study.

2.2 The status of History in the national curriculum

Historically, Chatterjee (2020) argues that the English word "history" has its roots in the Greek word "historia," which translates to inquiry and knowledge gained through investigation." Nevertheless, the word "history" comes from the German word "Geschichle", which means an important account of historical events. In general, History is the study of the past, covering all aspects of civilisation, human thoughts, deeds, and sufferings, and their impact on present events and changes (Chatterjee, 2020). According to Ajala (2003), History serves as a tool for fostering peaceful coexistence in society by imparting norms and values to learners in order to promote healthy human existence. History, according to Radha (2021), is the study of historical events, their causes and consequences and why and how they occurred.

Based on the findings of Nguringa and Cheprasow (2023) History refers to the study of evaluating and interpreting historical knowledge, lifestyles, and events. Nyarko (2023) defines History as the study of societal change and development over time and space. In addition, History, as defined by History Today (2024), is the study of people, actions, decisions, interactions and behaviour. As a result, the present study notes that History is the study of people, their actions, decisions, societal change and development as well as how people's actions, decisions, societal change and development affected the past, the present and the future.

The Ministry of Education and Training (2009) outlines Lesotho's secondary education aim as equipping high school learners with the necessary knowledge, skills, and attitudes for effective democratic participation and social activities. As stipulated in the History syllabus, History can equip learners with knowledge and skills for effective democratic processes and social activities,

instilling patriotism and nationality (Ministry of Education and Training (MOET), 2020). Another essential point is that the Curriculum and Assessment Policy (CAP) and Lesotho Basic Education Curriculum Policy (LBECP) frameworks aim to provide learners with moral and religious education, promoting social and cultural acceptable character development, co-operation and service to others (MOET, 2009; MOET, 2021). This is in alignment with one of the History aims being to promote understanding of Basotho's diverse social and cultural values, emphasising tolerance as a prerequisite for peace, stability, and national unity (MOET, 2020). To all appearances, the alignment of secondary education national curriculum goals with high school History syllabus proposes that History could significantly contribute to Lesotho's national curriculum for national development.

As identified by Wasserman (2010), History introduces learners to their heritage through teaching them about their society's origins, development, functioning and future state. In doing so, History promotes Basotho heritage, potentially fostering a sense of identity and culture essential for nation building (Fru, 2015b). Furthermore, Van Eeden (2012) argues that History enhances critical thinking skills in learners by evaluating evidence, distinguishing facts from complex information, and providing protection against propaganda. In support, Fru (2015a) asserts that History fosters responsible, reflective, and active citizens who can make informed decisions about societal issues at both local and global levels. Seemingly, History plays a significant role in the national curriculum of Lesotho as it improves critical thinking skills in learners, protects people against propaganda, and promotes identity, culture, and active citizens who make informed decisions specifically for nation building.

However, it seems that History is at risk of being removed from the curriculum due to unsatisfactory performance of learners in public examinations. Ntabeni (2006) confirms that the subject is experiencing a decline due to its unpopularity within the social sciences group. In addition, Fru (2015b) echoes that in Lesotho, History appears to be underappreciated in the 21st century. Ntabeni (2007) and Fru (2015a) argue that the high failure rate in History, the government's failure in promoting History and stigmatising History teachers and learners are contributing factors to the decline of History in the national curriculum of Lesotho. However, the subject itself is not a problem but the problem may emanate on how other educational stakeholders

including History teachers and learners perceive it. If they view it as a useless subject, they will approach it with that mentality and ultimately this may result in poor performance.

The literature further stresses that the status of History in in the national curriculum is unsatisfactory due to poor performance of learners in public examinations (Ntabeni, 2006). For instance, Fru (2015a) argues that after-lunch timetables lead to a regular neglect of History, which has a poor impact on learners understanding of the subject because of the excessive fatigue that follows from eating, playing, and occasionally intense sun. Therefore, learners may get disengaged from their studies and fail to comprehend crucial historical information, which could have a detrimental effect on their performance in public examinations.

For that reason, the consequences may result into overly dull and interesting subject reputation. That is why, according to Ntabeni (2006), many schools are not offering the History subject, leading to a decrease in in the number of History learners. As a result, the study by Ntabeni (2007) indicates that both learners and parents have a negative attitude towards the History subject, with some parents even discouraging their children from pursuing it. Notably, not only stigmatisation of the subject contributes to bad status of History but also negative attitude of other educational stakeholders such as parents, learners and even some teachers contribute to the bad status of the subject.

2.3 Factors influencing performance of History learners in public examinations

This section presents the factors influencing performance of History learners in public examinations. Generally, there are many factors influencing performance of History learners in public examinations. Therefore, the literature reviewed in this study reflects plenty of factors that are divided into internal (school), external, and personal factors. In this study, internal factors include classroom congestion, textbooks, libraries, computer laboratories, teaching aids, revision materials, information, communication and technology (ICT) (internet effectiveness, cellphones), assessment (homework, assignment, weekly tests and quizzes) and curriculum alignment (Mtwathwana et al., 2014).

The other internal factors are content coverage, relationship with principals, other teachers, and learners, support from school management, school type (private and government), coaching, English proficiency as medium of instructions, drilling, in-service training of teachers and teachers' role. In the context of this study, teachers' role is divided into professional development, pedagogical relationship between teachers and their learners, and mastery of subject matter (Limene, 2018). On the contrary, external factors consist of family education, family job types, family finance, parenting style, and parental involvement. Apart from that, in this study, personal factors include learners' behaviour and attitude, teachers' behaviour and attitude, and peer pressure. Below is the detailed review of each factor.

Primarily, Mtwathwana et al. (2014) identifies inadequate teaching and learning resources, libraries, laboratories, classroom congestion, teaching aids, and quality teaching as key factors influencing learners' performance in public examinations. Similarly, Mhonyiwa (2014) includes teaching resources like textbooks as contributing factors to performance of learners in public examinations. Additionally, the same scholar adds other factors influencing learner' performance in public examinations such as professional development, teachers' mastery of subject matter, teachers' experience, in-service training of teachers, motivation, English proficiency as a medium of instruction and school management and inspection. Similar to Mhonyiwa (2014), Annua-Sekyi and Nti (2015) consider motivation and support from school management as the key factors contributing to performance of learners in public examinations. They further mention factors such as pedagogical relationship between teachers and their learners, and assessment for learning. Concerning the issue of motivation, Oscan (2021) emphasises that teachers must be motivated, and the school environment itself must be motivating in order to enhance the performance of History learners.

Akkessa and Dhufera (2015) identify factors such as family education, and job types, which influence the economic status of learners' families and ultimately their performance in public examinations. Oscan (2021) categorises the same factors under economic conditions. What is more, Wachira (2021) pin-points factors such as availability of resources (textbooks), teaching environment, and learners' negative attitudes as contributing factors to performance of History learners in public examinations. More importantly, Mhonyiwa (2014), Mtwathwana et al. (2014),

Annua and Nti (2015) and Wachira (2021) find textbooks to be a similar factor. However, Mtwathwana et al. (2014) focused on lack of textbooks whereas Wachira focused on the availability of textbooks and Mhonyiwa (2014) together with Annua-Nti (2015) concentrated on how both availability and insufficient resources affect the performance of learners under physical environment.

Again, Limene (2018) identifies various factors affecting learners' performance in public examinations, including instructional resources, home-based factors (parental involvement, family socio-economic issues, parents' level of education), teacher-related factors, learners' attitudes, and school-based factors, including school environment and facilities. Regarding instructional resources, Sengai and Mokhele (2022) highlight the importance of their accessibility in the teaching of History because they help learners to conceptualise the subject matter and this may enhance the performance of History learners in public examinations. In line with Limene's (2018) findings, Asif et al. (2020) identify home-based factors such as family education, parental characteristics, and household income as significant determinants of learners' performance in public examinations. They further mention teacher qualification similar to the one found by Mhonyiwa (2014) and Annua-Nti (2015).

In addition, Mani et al. (2020) classify the above-mentioned factors influencing learners' performance into internal and external factors. According to their findings, internal factors include learners' proficiency in English, class congestion, assessments, textbooks, teachers' role in the class, technology and examination system. Further, the external factors consist of family problems, work and financial, social and other domestic problems. Seemingly, Mani et al.'s (2020) study is more encompassing as it covers most of the identified factors that may affect the performance of History learners in their public examinations.

On the other hand, Iquabal et al. (2022) classify factors that influence learners' academic performance into personal (individual, social and domestic) and school factors. They further subdivided individual factors into individual behaviour, self-reflection and motivation. The authors also categorises social factors into social interactions, online social media, communication and peer influence. They finally classify domestic factors into home environment, family income,

parents' education, family stress, and family support. Similarly, Briones et al. (2021) highlight factors such as parenting style, learner characteristics, internet effectiveness, lack of motivation as significant influences on performance of learners in public examinations. They further unpack parenting style into supportive, strict and democratic styles of parenting. Below is the explanation of the reviewed factors used in the context of this study.

2.3.1 Teachers' teaching experience

Teachers' teaching experience is one of the factors influencing the performance of History learners in public examinations. Milkie and Warner (2011) argue that teachers' teaching experience positively influence learners' academic performance. For instance, Mhonyiwa (2014) claims that teachers with over 5 years of teaching experience positively affect learners' performance in their public examinations, while those with less than 5 years of teaching experience have a negative impact. Bolapinwa et al. (2020) validate the findings of Mhonyiwa by revealing that in Nigeria, schools that have more teachers with over 10 years of teaching experience achieve better results than schools that have more teachers with below 10 years of teaching experience. In agreement, Oguta (2022) affirms that teachers' experience plays a crucial role in enhancing learners' success in public examinations because experienced teachers possess greater confidence, a deeper understanding of subject matter, and implement the curriculum effectively. Therefore, History teachers with more experience may enhance the performance of History learners, while those with less teaching experience may negatively impact their performance in public examinations because the experienced ones may have more exposure to historical content and knowledge while less experienced ones may have more exposure to historical content and knowledge while less experienced ones may not.

Nevertheless, Limene (2018) challenges findings of the preceding scholars by arguing that learners of experienced teachers with 5-15 years of teaching experience often perform below public examination expectations as discovered in Namibia. This was supported by the study of Musau and Abebe (2015) which demonstrates that teachers' experience does not significantly contribute to the high performance of learners in public examinations. They cite Kenya where 17 fresh graduates who have less teaching experience improved the performance of learners than 18 inservice teachers with more than 5 years of teaching experience. As a result, the argument suggests that teachers' teaching experience alone may not significantly improve learners' performance in their public examinations but it will also depend on the effort of individual teachers. Further, as

per Musau and Abebe (2015), performance of History learners in public examinations may be improved by the career growth of teachers. This implies that the more career growth of the History teachers, the more will be the improvement of the performance of History learners in their public examinations.

2.3.2 School type

In the context of this study, school type is classified into two categories being private schools and public schools or government school. Mhonyiwa (2014) found that school type affects performance of learners in public examinations, with private schools performing better than government schools. As to the support offered by Monserate (2018), learners who attend private schools are in a better position than those who attend government or public schools because they have more access to computers and the internet. In assertion, Canziz et al. (2016) concur that attending private school boosts learners' academic performance in their public examinations since these learners are intrinsically motivated to learn there. Based on the ideas of the aforementioned scholars, it is clear that private school History learners perform better than public school learners due to increased motivation and access to resources like computers and internet.

Hahn et al. (2014) corroborate the aforementioned claim by stating that learners in private schools perform better than those in public schools since the former is superior in terms of preparing learners for tertiary education. Hurry (2016), on the other hand, challenges the notion that learners' performance in public examinations is determined by the type of school they attend. Rather, performance is driven by a number of factors, such as family income, teacher qualifications, and school management. Hurry appears to be right hence it is not always the case that learners' performance in History is determined by their school type. In the context of Lesotho, for example, there are numerous government schools that do better in History than private ones. Although private schools outperform public schools, it would seem that performance of History learners in public examinations may not only be attributed to school type alone but it may also be attributed to other several factors such as teacher qualifications and school management.

2.3.3 Teachers' professional qualifications

Professional teachers and their development may affect the performance of History learners in public examinations. As identified by Mhonyiwa (2014), teachers' professional development encompasses their understanding of the subject matter, curriculum, and methods of how learners learn. Bernard's (2009) study shows that some teachers are not subject-specialists whereas some are under qualified. In the opinions of Mani et al. (2020), unqualified teachers often struggle with effective use of teaching resources, accepting learners' responses, dominating classrooms, addressing incorrect responses, and explaining concepts effectively. Therefore, it seems that such teachers may negatively affect performance of History learners in public examinations, indicating the need for schools to engage high qualified History teachers. On the contrary, Briones et al. (2021) suggest that high qualified teachers possess a greater understanding of their subject area's pedagogical content. In appearance, Briones et al.'s (2021) findings are accurate, as highly qualified teachers may possess sufficient content to be effectively transmitted to their learners. For that reason, qualified teachers may improve the performance of History learners in public examinations.

2.3.4 Teachers' mastery of subject matter

More importantly, teachers' mastery of subject matter is among the factors that affect performance of History learners in public examinations. The teachers' mastery of subject matter is categorised into knowledge of content, pedagogical content knowledge and general pedagogical content knowledge (Mhonyiwa, 2014). This implies that History teachers must effectively teach various subject concepts to learners as per Shing et al. (2015). Contrary to this, the literature indicates that some teachers who lack mastery of certain topics may struggle with content and pedagogical skills (Heeralal & Dhurumraj, 2017). Moea (2022) argues that teachers' lack of pedagogical skills is attributed to their unclear understanding of the curriculum due to their exclusion in curriculum planning and their lack of in-service training. With that being said, it seems that teachers' poor mastery of subject matter may be due to their lack of pedagogical skills, which may not only emanate from misunderstanding of the curriculum but also from their lack of in-service training through workshops. This may contribute to lower performance of History learners in their public examinations since teachers struggling with content knowledge may not deliver historical content effectively to their learners.

2.3.5 Content coverage

Content coverage is another factor influencing performance of History learners in public examinations. Content for education includes all of the procedures, knowledge, skills, attitudes, abilities and values needed to accomplish the learning objectives listed in lesson plans and programs (Smowltech, 2024). According to Smowltech (2024), effective and appropriate content promotes learning, accessibility, and adaptability by allowing learners to effectively interact with their teachers. Nakhanu (2012) claims that although learners who cover the syllabus early outperform those who do not, most of the schools only cover a portion of the curriculum and never teach certain topics, in most of the schools. According to the findings of the aforementioned scholars, it appears that learners who complete the syllabus early in the year do better in their public examinations since they dedicate their more time on revision, even though some schools do not cover the syllabus due to various challenges such as shortage of teachers.

Wafulal et al. (2019) point out that late or non-content syllabus coverage negatively impact learners' performance, with in-depth and timely coverage leading to better performance of learners in their public examinations. It is in this regard that Peterson et al. (2020) argue that teachers must view content as a collection of concepts, competences, and facts to be deliberately selected rather than a list of topics to be covered. This means that History teachers must teach not only to cover content but also to instil broad knowledge in learners for their effective learning. On the other hand, Oketch et al. (2013) debate that content coverage influences learners' performance positively and significantly if content coverage is the only predictor. For that reason, teachers in low performing schools should not be held accountable; instead, the government should investigate non-school factors as they may be partially responsible for the persistent differences that may result in the unfavourable failing schools and few succeeding schools.

2.3. 6 Revision materials

Revision materials are some of the factors influencing performance of History learners in their public examinations. Based on the ideas of Cross et al. (2016), using sample exam papers can enhance question literacy, making exam questions clearer and easier to understand for learners. For this reason, this may enable History learners to understand examination questions and answer them correctly hence their higher passing rate. As specified by Onteng and Bruwer (2020), learners regard viewing and practising of past assessments to have an influence on their performance in

final examination. Learners utilise past assessments as a deductive tool in order to predict future areas that will be assessed as well as certain characteristics such as the level of difficulty of question papers and how to address them.

According to Wangdi and Zangmo (2023), the time used to revise the lesson taught in class engage learners in learning and has great impact on learners' performance in public examinations. Lopz et al. (2021) highlight that revision instruction increases learners' ability both to detect and correct substantive issues in the text and benefits the quality of writing of learners hence enhance their performance. As a result, it appears that revision materials may improve the performance of History learners in their public examinations as they help learners to be familiar with the questions that are asked in public examinations. Apart from that, most of the questions repeat themselves and revision materials may help them to be in an opportunity of getting such questions correctly hence the improvement of their performance. However, the literature highlights that it is not always the case that the questions are repeated and learners must not rely on revision materials only but should also read their notebooks and do their work on time (Lopz et al., 2021).

2.3.7 Lesson preparation

Besides, it would seem that teachers' lack of pedagogical skills might be attributed to lack of lesson preparations. Mumba (2019) suggests that lesson preparations can enhance learners' examinations performance by preparing lesson plans, scheming, selecting effective teaching methods, and monitoring prepared work by administrators. In addition, Strong (2021) emphasises that lesson planning facilitates the learning of learners by positively influencing their affective, cognitive, and psychomotor skills. On the other hand, it seems that History teachers without lesson planning can lead to confusion among the learners, hindering meaningful learning and good performance of their learners in public examinations (Khan et al., 2024). Therefore, the above scholars demonstrate that inadequate and improper lesson preparation may significantly contribute to poor performance of History learners whereas adequate lesson preparation may positively affect the performance of History learners in their public examinations.

Straessele (2014) stresses that the scheme of work clearly outlines the use of resources, class activities, and assessment to ensure that the course's learning aims and objectives are met. Further, Khan et al. (2024) affirm that the scheme of work assists school supervisors in assessing teachers'

efforts in achieving the syllabus objectives. Nonetheless, the findings of Mumba (2019) demonstrate that lack of preparation of scheme of work might lead to learners performing poorly academically since teachers may be teaching haphazardly without proper arrangement of teaching resources, and teaching aids. Additionally, the literature shows that teachers may also run the risk of excluding significant and often examined topics from the syllabus (Munba, 2019). Seemingly, proper lesson preparation through a well-structured scheme of work positively impacts History learners' performance in public examinations, as teachers ensure proper use of teaching resources, activities, and assessments unlike teachers without proper scheming who teach haphazardly. It would appear that teachers who teach in this way may improperly plan, ineffectively deliver the content to learners, and may skip important content which may appear in the final examinations thus contributing to the failure of History learners in their public examinations.

Evidently, Mhonywa (2014) found that in Tanzania, some of the teachers were frustrated by the challenging topics they teach their learners. Keeping that in mind, some teachers allow learners to work in groups without brainstorming previous topics, instead of preparing lesson presentations on some of the topics they teach (Iquabal et al., 2022). Apparently, these are the signs of ineffective teachers who Ntabeni (2006) regard as ineffective in preparing History learners for essay writing skills, leading to high failure rates on performance of History learners in public examinations. Bearing that in mind, it would seem that such kinds of teachers may inadequately prepare for their lessons prior to their teaching process.

The study of Aada (2020) reports that ineffective teachers without proper planning struggle to effectively facilitate learners from previous lessons to the current ones, leading to failure for learners to establish a connection between lessons. Consequently, this may affect the examination performance of History learners negatively. On the other hand, Limene (2018) asserts that knowledgeable teachers with proper lesson preparations effectively scaffold learners and connect previous topics to current ones. Besides, such teachers may even use teaching aids and various learner-centred teaching methods such as role-play, presentations and debate hence this may result in good performance of History learners in their public examinations (Alkalaki, 2021). Taking that into account, it appears that History teachers with well-structured lesson preparation may

positively affect the performance of their learners in public examinations as they may use teaching strategies and teaching aids properly.

The literature further suggests that selecting effective teaching methods can positively influence the performance of History learners in their public examinations (Muhoza et al., 2023). Munyaradzi (2013) suggests that combining teacher-centred and learner-centred teaching methods is the most effective approach for achieving optimal learners' outcomes. Even though Munyaradzi recommended using both of these teaching strategies in the teaching and learning process, learner-centred approaches promote better topic mastery than teacher-centred which centralise the flow of knowledge as one-way from a teacher to the learners. In support, Badar et al. (2022) emphasise the importance of learner-centred methods like demonstrations, stating that they lead to improved learners' performance. As a consequence of that, it would seem that interaction among learners and between learners and teachers can enhance their knowledge acquisition, leading to better performance in public examinations, emphasising the importance of learner-centred teaching methods.

2.3.8 Coaching in the teaching of History

As discovered by Ehsanipour and Zaccarellin (2017), coaching is defined as a tool for teachers to enhance their ability to assess the effectiveness of instructional strategies in supporting learning. As to the definition provided by Krat et al. (2018), coaching is a type of professional development program where teachers receive feedback from peers or coaches after they observe their instruction to assist them better. According to Trido (2021), coaching consists of a few fundamental components, including continuing support, non-evaluative feedback, and assistance in helping teachers gain a better knowledge. Therefore, it appears that coaching could be advantageous for both teachers and learners since it could enhance learning and encourage the retention of highly successful teachers, both of which would eventually boost learners' performance. It is for this reason that the literature suggests that both teachers and learners require to go through the process of coaching (Summer, 2011; Krat et al., 2018).

Based on the ideas of Kenny and Faunce (2004), learners who experience and enjoy coaching outperform learners who do not experience and enjoy coaching. Using the concepts from Summer (2011), instructional coaches assist teachers with lesson analysis, use of assessment data, lesson

planning, teacher observation and conferences, and lesson modelling. However, despite the fact that coaching may promote the performance of History learners, the literature further specifies that there are some learners who perform worse even when they are coached (Kennedy & Faunce, 2004). Therefore, it would appear that coaching History teachers and their learners may be beneficial as it may enhance the performance in History public examinations. However, coaching may require effective training of History teachers and regular monitoring to learners for it to be effective and beneficial to learners in public examinations.

2.3.9 Drilling in the teaching of History

According to Aini et al. (2020), drilling involves repeating what is heard after listening to the model which is given by the teacher or another learner in the classroom. Drilling is further described by Mayasari (2020) as an instructional strategy in which teachers model a word or sentence and learners repeat it in order to help learners recall and understand the aim. In accord, Rofiah (2022) says that drilling is a method that teachers employ while imparting new language concepts to their learners. During this process, learners listen to their teachers and mimic their phrases and words. It is apparent that drilling appears to be an instructional strategy used by teachers whereby learners are taught to repeat what they are taught by their teachers to help them understand the taught concept. Seemingly, understanding the concept may help History learners to perform well in their public examinations.

Due to Septibera's (2016) observation that drilling could increase learners' participation by allowing them to respond to the teachers, pay attention, listen to the teachers, repeat what they hear, and complete the task, it is possible that using drilling in the teaching of History may be advantageous to History learners. Mayasari (2020) reiterates that drilling enhances learners' learning and participation by providing unique information, as all learners need to learn from each other. According to the literature, drilling may help History learners become more proficient in vocabulary, communication, pronunciation, and may also encourage lengthier word memory, error correction, and conversation practice (Rofiah, 2022). In doing so, drilling may engage learners in learning, help them in understanding historical content, terminologies, and facilitate their retention of information, ultimately leading to better performance in public examinations.

On the other hand, the literature notes that teaching drilling is challenging since it calls for an extraordinary amount of the teachers' creativity, skills and effort (Mayasari, 2020). Besides,

Rofiah (2022) adds that drilling takes a lot of time, is monotonous unless learners are highly motivated, does not always help them understand the content being taught, and can prevent learners from applying their skills in the same situation if it relies too heavily on memorisation. Therefore, this may support the findings of Aini et al. (2020) who emphasise the need for drilling to be used in conjunction with action research cycles from cycle to cycle in order to enhance History learners' comprehension. If that is the case, understanding historical content could help History learners in performing well, just as practising oral presentations and memorisation exercises through drilling improve these skills.

2.3.10 Relationship with the principals, other teachers and learners

The relationship of History teachers with the principals, other teachers and learners appear to be one of the factors influencing learners' performance in History public examinations. Based on the ideas of Edgerson et al. (2006), teachers who have strong relationship with their principals inspire and motivate their learners to perform at the highest level in public examinations. This is because positive environments boost teachers' energy and productivity and help learners perform well on public examinations. Koso et al. (2019) state that principals' leadership responsibilities connected to the development of learners' character are restricted to the setting an example for learners, inspiring them to think critically, and offering daily guidance for learners' academic progress. In this regard, the relationship of the principals with both History teachers and learners should be cordial for the benefit of learners, and this may contribute to the creation of positive learning environments that may ultimately improve the performance of History learners in their public examinations.

Based on the existing literature, fostering great school cultures and guaranteeing learners performance require the presence of amicable, cordial relationship between the teacher and learners and between the teachers and principals (Better Leaders Better Schools, 2022). Accordingly, Edigbo (2024) states that cordial relationship between the teachers and the learners result in improved performance of learners in public examinations since it facilitates teaching and learning process. On the other hand, Emosivwe et al. (2023) show that while a friendly relationship between teachers and learners improves learners' performance in public examinations, most secondary schools do not have a friendly relationship between teachers and learners because most teachers punish their learners for misbehaviour, and some teachers are harsh with their learners in

the classroom. Consequently, as highlighted from the literature, harsh teaching styles may negatively impact History learners' performance in public examinations, as they may retreat and hide their abilities in such situations (Edigbo, 2024). It is therefore, against this matter that cordial relationship between the History teachers and their learners is required in the teaching of History in order to improve performance in History public examinations.

2.3.11 Assessment methods

Equally important, assessment methods like testing, assignments, homework and feedback can significantly influence the performance of History learners in their public examinations, either positively or negatively. This may depend on whether or not teachers regularly and effectively assess their learners through homework, assignments, quizzes, and classwork each week (Atabek et al., 2014). Moreover, Shin et al. (2017) add that this may also depend on whether teachers provide their learners with immediate feedback and do corrections with their learners or not. Considering that, Djarfour (2018) emphasises the importance of weekly feedback from tests, assignments, homework, and class activities as it aids learners in recognising and correcting their mistakes. This means that it is apparent that History learners' performance in public examinations is influenced by teachers' regular assessment and immediate feedback, with regular assessment and immediate feedback positively affecting performance, while rare assessment and delayed feedback negatively affect it.

According to Annua-Sekyi (2015), some teachers skip difficult topics, provide insufficient feedback, and provide destructive comments on learners' feedback, while others incompetently prepare for examinations. Therefore, according to Mani et al. (2018), the aforementioned factors symbolise that learners are partially taught. History learners may not perform well in their public examinations because the examinations are designed in accordance with the prescribed content to be covered (Iquabal, 2022); as a result, History teachers must provide learners with immediate feedback with constructive plans as recommended by the Lesotho educational policy frameworks (MOET, 2009; MOET, 2021).

Besides, Moghaddam's (2013) research suggests that weekly quizzes significantly enhance the performance of learners compared to those who do not participate in them. To validate this, Basolm and Johnson (2009) state that weekly quizzes generate extrinsic motivation for learners, as they

strive for good grades in their courses, leading to extensive preparation and effort not only in their weekly quizzes but also in their tests, home works, and assignments. Besides, the literature demonstrates that frequent tests expose learners to class material, helping them become familiar with examiners' expectations and test-wise, enabling them to identify questions for final examinations (Moghaddam, 2013). Therefore, it appears that frequent tests and quizzes may reduce learners' test anxiety, leading to enfeebling test anxiety during final examinations compared to those who take tests less frequently. Atabek et al. (2014) stress that practising and testing aid learners in retaining previously learned information hence this prepares for public examinations.

Djarfour (2018) proclaims that assessment helps in knowledge development of learners that enables them to see their errors and shortcomings, and enables them to work towards correction. Additionally, Oyinloyr and Imenda (2019) confirm that learners who adhere to the assessment teaching strategies outperform those who do not. According to Songsirisak and Jipranee (2019), homework and other forms of assessments have a significant impact on learners' performance because they force them to focus more intently on the materials and lessons, which may improve test results. In support, Haq et al. (2020) affirm that homework not only reinforces classroom instructions but also enhances understanding and keeps learners engaged in their studies at home. Therefore, it appears that frequent assessment with immediate feedback motivates learners intrinsically and extrinsically, while not only rare assessment but also over-loading assessment demotivates learners to learn.

2.3.12 English as a medium of instruction

English as a medium of instruction is another factor that influences performance of History learners in public examinations. As noted by Mlay (2010), History learners struggle to communicate well in English and lack confidence in using it. It is for this reason that teachers use code switching and code mixing in their teaching in order to help learners to understand the subject matter. This is supported by Mhonyiwa's (2014) study, which demonstrates that most teachers utilise their mother tongue for their teaching instruction to assist learners to understand the subject content. According to Patrick (2020), many high school learners struggle with language errors like unclear instruction, inappropriate word usage, and poor written language structure. However, the literature discovered that incompetence in instruction language is a problem not only for learners but also for teachers, as many lack fluencies to confidently deliver engaging lessons (Mhonyiwa,

2014; Patrick, 2020). Therefore, it seems that not only History learners but also their teachers face challenges with English usage, often delivering content in their mother tongue, highlighting the need for History teachers to use English for instruction and mother tongue for scaffolding learners only.

2.3.13 Teaching resources

According to Gogliotti (2020), teaching resources are important components of high-quality education. As identified by Organisation for Economic Cooperation and Development (2020), teachers need teaching resources in order to transmit education that is up-to-date and that is challenging and responsive to the needs of learners. Sengai and Mokhele (2022) further state that improved access to instructional resources can enhance teachers' effectiveness in the classroom by aiding History learners in conceptualising taught content and having a significant impact on their academic performance in their public examinations.

Contrary to this, Juliana (2020) emphasises that inadequate resources in high schools have a severe negative impact on both learners' learning and teachers' ability to teach a class. However, History teachers must be as innovative as they can in addressing the challenges of inadequate resources. According to Sengai and Mokhele (2022), it may be encouraged for teachers to create basic teaching resources so that learners are better acquainted with the material they would be covering in class. Seemingly, these inadequate resources may demotivate teachers, leading to stress and negative impact on the performance of History learners. In the context of this study, the discussed teaching and learning resources include textbooks, computer laboratories, library resources, and technology. The resources are discussed in detail below starting with textbooks to technology.

2.3.13.1Textbooks

Textbooks in schools may affect the performance of History learners in public examinations. Evidently, in Tanzania, Mhonyiwa (2014) exposes an extreme shortage of textbooks in government schools compared to private schools. In the opinions of Mwatwana et al. (2014), government schools face textbook shortages due to limited government funds, causing parents to rely on purchasing expensive textbooks for their children. In consideration of that, Lamas (2015) proposes that parents' inability to afford expensive textbooks for their children may lead to unsatisfactory performance of learners in public examinations. Resultantly, Makori and Onderi (2014) argue that shortage of textbooks influences teacher workload, pace of learning, and

negatively impact syllabus coverage, potentially leading to poor performance of History learners in their public examinations. Bearing that in mind, the absence of textbooks can negatively impact performance of History learners, while their availability and effective use can positively impact their performance in public examinations.

2.3.13.2 Library resources

Aside from textbooks, library resources may also affect the performance of History learners in their public examinations. According to Mhonyiwa (2014), library resources are crucial for teaching and learning, instilling a reading culture among learners. Mwathwana et al. (2014) exemplifies that high schools of Kenya often lack current textbooks, store prehistoric and unapproved materials in their libraries and some libraries lack textbooks. In a similar manner, Moea (2022) notes that in Lesotho, high schools lack libraries and textbooks. Therefore, it is without doubt that unavailability of libraries may negatively influence the performance of History learners' in their public examinations.

2.3.13.3 Technology

Technology is another factor that may influence the performance of History learners in public examinations. According to the literature, technological tools such as computers, projectors, internet and audio-visuals can be employed in the teaching of History to enhance the performance of History learners in their public examinations (Boadu et al., 2014). Monserate (2018) further expresses that teachers and learners' computer literacy positively affect the performance of learners in public examinations. To bolster this, Risda and Rey (2023) note that learners who receive instructions using technology outperform learners who receive instructions through traditional ways of teaching since technology improves learners learning and useful teaching practices. Additionally, Monserates (2018) highlights that learners attending private schools tend to possess greater self-efficacy and utilise technology more extensively, leading to improved performance in their public examinations. In light of this, technology may help History learners perform better in their public examinations because learners who are computer literate outperform those who are not.

However, Bingimlas (2009) identifies challenges in technology integration, including lack of accessibility, poor resource organisation, poor hardware, inappropriate software, insufficient

computers, lack of technical support, slow internet, lack of teachers training, malfunctioning computers, and outdated computers. Apart from that, Boadu et al. (2014) highlights other significant barriers to integrating technology in History teaching, including the unavailability of technological tools, insufficient time, and lack of administrative support. Ismai et al. (2020) further highlights the challenges of integrating technology in teaching including lack of teachers' relevant skills, lack of expert and inadequate administrative support. All these factors mentioned above may not only negatively affect the internal performance of History learners but also their performance in public examinations.

Besides, Dube et al. (2018) point out that poor training, negative attitudes and lack of ICT software are the challenges of integration of ICT in the teaching and learning. Kennedy (2023) states that the main obstacles to integrating ICT into the classroom are a lack of computers, projectors, lack of institutional internet access, lack of support for instructors, lack of training, a shortage of ICT experts, funding inflexibilities, and lack professional development. Even though technological tools may enhance the performance of History learners in their public examinations, integrating technology in History teaching is challenging due to lack of administrative support, lack of accessibility, poor resource organization, poor hardware, in-appropriate software, and lack of technical support, and less ICT experts. Factors such as teacher competence, training, malfunctioning computers, inadequate computers, old computers, unavailability of projectors, slow internet, lack of internet, lack of History teachers' training, insufficient time, and funding inflexibilities also hinder progress.

2.14 Computer laboratories

Literature reveals that use of computers in teaching and learning could address textbook and library shortages, improve performance of History learners in their examinations, as they perform similar tasks (Mhonyiwa, 2014). However, Limene (2018) reports that while some schools have computer laboratories in Namibia, the internet is often unavailable. Besides, Mani et al. (2020) highlight that some schools neglect to utilise existing computer laboratories and internet facilities, limiting teachers and their learners to look for updated knowledge. As far as anyone can judge, computers are very important because they may address the challenges of lack of textbooks. However, it seems that most of the schools that have computer laboratories often lack internet facilities as is

the case of Lesotho, potentially leading to lower performance of History learners in public examinations (Moea, 2022).

2.3.15 Motivation of teachers and learners

Motivation of both teachers and learners may also influence the performance of History learners in their public examinations. According to Liu et al. (2022), motivation is the driving force behind goal-directed academic activities and key to the good academic performance of high school learners. In that regard, Wilkesmann et al. (2021) suggest that teachers play crucial role in fostering the motivation of learners. Although teachers have a significant influence in motivating learners, learner and teacher motivation may be equally important in improving performance of History learners in their public examinations.

Correspondingly, Mudago (2017) asserts that teachers' motivation stems from their desire and commitment to achieve individual and school goals, which is reflected in their job performance and this can maintain learners' attention, behaviour, and provide their energy for task completion. Thoka (2020) argues that lack of self-motivation among the learners is the major cause of poor academic performance of learners in public examinations. Therefore, according to the above scholars, it seems that motivation can result in good performance of learners while demotivation can result in poor performance in their public examinations and teachers are key to motivation. Not only that, teachers' motivation seems to come from their dedication and desire to accomplish personal goals, which is evident in how well they perform at work.

Mruma's (2013) ideas suggest that teachers' motivation stems from their work environment, which in turn boosts their morale, enabling them to perform better than average. It is in this regard that their effort may ultimately and positively affect the performance of their learners in their public examinations. Nevertheless, Mhonyiwa (2014) states that government high school teachers are often demotivated to teach their specialist subjects due to factors like low salaries, inadequate resources, and lack of promotion whereas private school teachers are motivated by their high salaries and close supervision from heads of department and principals. As a result, it appears that school environmental factors such as resources may be the source of teachers' motivation and demotivation. Adequate and effective use of these factors may result in teachers' motivation, while inadequacy and ineffective use may result in teachers' demotivation.

In addition to that, Mudongo (2017) highlights that teachers' motivation stems from job satisfaction and career development opportunities like promotions, in-service training, workshops, seminars, and revision courses. For that reason, Annua-Sekyi and Nti (2015) argue that demotivated teachers do not truly teach, but rather instruct and assign tasks to their learners. Seemingly, job satisfaction and in-service training also contribute to teachers' motivation. Teachers who are motivated can have a positive impact on performance of History learners in their public examinations, while those who are demotivated can have a negative impact.

2.3.16 Support from school management

Support from school management may also affect the performance of History learners in public examinations. As highlighted by Nzoka and Orodho (2014), most school managements fail to adequately support their teachers and learners due to their inconsistent instructional supervision and uncoordinated guidance and counselling. This may affect the performance of History learners in their public examinations because teachers may not teach learners well enough to prepare them for public examinations. For example, there may be high rate of History teachers' absenteeism, and some may even skip important historical content which may appear in the public examinations. It is against this notion that Sengai (2021) posits that the responsibility of the head of department to guarantee that History is taught correctly in the classroom might result in the adoption of demanding programs for instructional leadership, which will help to enhance teaching methods in History classes to ultimately improve the performance of History learners in their public examinations.

Additionally, this support may be determined by the management styles used by the principals of the schools as demonstrated by Mani et al. (2020). These authors claim that autocratic principals often communicate ineffectively with teachers, preventing them from expressing their opinions on learners' performance in public examinations. The literature suggests that this ineffective communication of autocratic principals to teachers and learners suggests a lack of support from school management (Oscan, 2021). If that is the case, this kind of management style may demotivate teachers to teach and learners to learn and ultimately affecting the performance of History learners negatively in their public examinations. At the same time, this lack of support from school principals seems to be related to insufficient management training, which exposes

their shortcomings and may have a detrimental impact on History learners' and teachers' performance in History public examinations.

Furthermore, Sulan (2024) confirms that learners' performance is negatively affected by autocratic school principals who typically exhibit excessive strictness and harshness, discouraging their teachers from giving their best effort. Nonetheless, Oscan (2021) argues that principals with effective instructional leadership skills improve school system development by enhancing teachers' knowledge levels and enhancing learners' achievement. According to Senguo and Ilomo (2020), these kinds of principals work effectively for planning, motivating, and inspiring learners to put in a lot of effort in order to enhance their academic performance in public examinations. On the other hand, it would seem that supportive principals with effective leadership skills can significantly influence curriculum instruction, academic success, and performance of History learners in public examinations.

2.3.17 Family job types and income

According to Akessa and Dhufera (2015), the family job types affect learners' academic performance in public examinations either positively or negatively. They further argue that learners with working parents perform better than those with farming or not working parents. Asif et al. (2020) confirm that household income significantly and positively affects performance of learners in public examinations. Resultantly, it seems that the wealthy families can purchase necessary teaching resources like textbooks, smart phones, and data for History learners, promoting stress-free learning for learners hence improving performance of History learners in public examinations.

To the casual observer, learners from low socio-economic families often struggle to afford teaching resources, school fees, and may be expelled from schools, which can demotivate them and hinder their learning experiences. As a result, this creates additional stress on both learners and their parents thus negatively affecting performance of History in their public examinations (Oscan, 2021). To elaborate, Khan (2010) highlights that South Asian learners from low socio-economic backgrounds change school attendance dates based on their attendance and absence patterns as they become present on certain days and absent on certain days. Therefore, this may affect the performance of History learners in their public examinations negatively.

2.3.18 Examination system

Based on the findings of Limene (2018), the ministerial system of automatic promotion negatively influences the academic performance of completing learners. In support, Zhang and Huang (2022) assert that learners are promoted to the next grade without achieving the basic competencies of the current grade. James (2017) highlights the significant issue of learners appearing unprepared for the next grade due to this serious problem. Therefore, promotion may significantly influence the performance of History learners negatively in their public examinations, as promoted learners may be demotivated to work hard.

With reference to Lesotho, Moea (2022) comments that the introduction of the integrated curriculum brought the idea of promotion, preventing primary and some high school learners from repeating even if they failed. The literature further reveals that promotion often results in incompetent learners, making it challenging for high school teachers who lack elementary teaching skills (Moea, 2022). Additionally, James (2017) asserts that learners who are automatically promoted from previous classes are not prepared for the current class. The literature further indicates that promotion leads to lazy learners who do not put extra effort to pass to the next class putting more loads on high school teachers. Seemingly, high school teachers may struggle to cover anticipated content due to learners' unpreparedness in reading and writing. Some teachers may use a lecturing method, rush over content and skip another content causing a higher failure rate among History learners in public examinations (Limene, 2018).

2.3.19 Curriculum alignment

Phaenton (2017) highlights the confusion in teachers' interpretation of curriculum due to misalignment between curriculum designers and examiners, who often fail to express the curriculum's intention regarding practical work. According to Meij and Merx (2018), curriculum alignment at program-level, that is, the positive coherence between instruction, learning and assessment is crucial. In the opinions of Oscan (2021), in some schools, there are misalignment between the examination questions set and the marking schemes used. According to the literature, some of the misalignment are the results of exclusion of History teachers in curriculum design (Moea, 2022). That is why Sengai and Mokhele (2021) underscore that in order to effectively execute curriculum reforms, policy-makers should make an effort to include teachers in the process

as this may allow them to voice their concerns that are critical in the successful curriculum implementation.

The literature further stresses that there is also misalignment between the content of the syllabus and the examination questions set (Limene, 2018). In addition, Moea (2022) warns that unclear communication between curriculum designers and examiners as well as markers affect teachers negatively who in turn affect the performance of their learners negatively. Therefore, it would appear that misalignment of curriculum between designers and examiners may negatively affect the performance of History learners while proper curriculum alignment may positively affect the performance of History learners hence a need for alignment between planned and implemented curriculum. It is in this regard that Sengai and Mokhele (2022) accentuate that History teachers should be urged to use their creativity and utilising educational resources that support the themes they teach in each lesson, particularly when a new syllabus is implemented.

2.3.20 Health of learners in the learning of History

Health is another factor that influences performance of History learners in public examinations. According to Show et al. (2015), learners with poor health have higher probability of failure in their public examinations because poor health on children results into emotional, social, physical, and mental problems. In concurrence, Olusegun (2017) acknowledges that learners who frequently attend class due to their good health appear to perform better in their studies and are more successful than the ones who do not regularly attend class due to poor health. This poor health may emanate from poor nutrition, hunger, poor hygiene and lack of humanisation and it may ultimately hinder History learners from performing well in their public examinations because it demotivates them to learn. To add more, Sabir et al. (2021) confirm that learners' general well-being has significant impact on how well they perform in their public examinations. With that in mind, it would seem poor health of History learners may negatively affect their performance in public examinations while good health may positively affect their performance in public examinations.

2.3.21 The attitude and behaviour of History teachers

The attitude and behaviour of teachers also influence the performance of History learners in their public examinations. Based on the findings of Ullah et al. (2018), teachers who apply positive attitude and behaviour towards their learners motivate them in learning of the subject. In a similar

vein, Dar (2022) argues that positive teacher behaviour and interactions with learners strengthen the potential needed for enhanced performance of learners in public examinations, while on the other hand, harsh teachers can lead to demotivation of and poor performance of learners. As a result, it would appear that History teachers' pleasant behaviour and attitude could be highly important in motivating learners since it allows them to engage effectively with their teachers, which could improve learners' performance in their public examinations.

Additionally, Kahveci (2023) highlights the impact of teachers' attitude and behaviour on learners' academic, social, and emotional development, with positive interactions boosting comfort and interest, and oppressive ones leading to long-term sadness and failure in public examinations. Contrary to this, Baloglu (2009) explains that teachers act aggressively and authoritatively, with male teachers physically abusing and humiliating learners while some female teachers dress in a sexually aggressive manner. Further, Le Courtois (2024) warns that certain punishment methods and negative teacher feedback can harm the relationship between teachers and their learners, potentially leading to insecurity and demotivation. All these acts may destruct History learners and demotivate them to learn hence affecting their performance in public examinations. Therefore, positive attitude and behaviour of teachers may enhance learners' performance, while their negative attitude and behaviour may negatively impact History learners' performance in public examinations.

2.3.22 Attitude and behaviour of History learners

According to Simba and Kabuka (2016), general discipline characterised by obedience, politeness, orderliness, and social competence can influence the attitude and behaviour of learners that may ultimately influence their performance. This is because according to Dar (2022), indiscipline hinders learners' focus on educational goals, which require hard work, time management, and respect while disciplined learners are more likely to stay focused in their educational goals, and succeed academically. This is more so, because they are less likely to become involved in disciplinary cases that could take them away from their studies. Therefore, it would appear that such learners portray positive behaviour and attitude in learning and this may enable them to enhance their striving for academic success and eventually boost their performance in their public examinations.

According to Dar (2022), positive teacher-learner relationships have an influence on learners' attitude and behaviour in the classroom because happy teachers motivate them to have positive attitude towards the classroom learning and school as a whole. However, it is not always the case that learners portray positive attitude and behaviour in the classroom as sometimes they portray disruptive behaviour. By definition, Okeke et al. (2023) define disruptive behaviour as actions that hinder classroom facilitation and educational outcomes, often resulting from learners violating rules. To put it in simpler words, disruptive behaviour of History learners may not only affect their teachers negatively but also their performance in pubic examinations.

The literature also portrays that this type of behaviour could be detrimental to History teachers' well-being since it could be difficult for them to deal with disruptive learners' behaviour because it is unbearable and stressful, which lowers History teachers' psychological and cognitive well-being (Okeke et al., 2023). In this way, History learners' disruptive behaviour and attitude may demotivate their teachers and ultimately affect their performance in public examinations negatively as teachers may ineffectively transmit historical knowledge to them. Conversely, Kahveci (2023) debates that negative attitude and behaviour of teachers can lead to learners' negative attitude and behaviour ultimately resulting in subject failure due to feelings of sadness, hatred, anger, and oppression. From all appearances, teachers' attitude and behaviour significantly influence learners' attitude and behaviour with positive teachers motivating and developing positive learning attitude, while negative teachers demotivate learners, leading to high failure rates in public examinations.

2.3.23 Congested classrooms

Congested classrooms may also influence the performance of History learners in their public examinations. Mani et al. (2020), clarify that there are congested classrooms with inability to accommodate a large number of learners in high schools. According to West and Meier (2020), lack of teachers and inadequate classrooms are the key contributors of congested classrooms. These authors stress that teachers struggle with managing large classrooms with learners and efficiently revisiting stages, as they have limited time to attend to all learners. Consequently, based on the ideas of Mankgele (2023), classrooms with insufficient spaces are found to hinder learners effective learning and outcomes. Apparently, this may affect the performance of History learners in their public examinations because teachers may fail to address all the learners such as failure to

offer continuous assessment through several quizzes, tests, and exercises to maintain learners' commitment and understanding of the taught concept. For example, the teacher may not be able to address all learners' questions during learning, as they may need clarification on specific concepts.

2.4 Mitigation of History learners' performance in public examinations

Historically, Annua-Sekyi and Nti (2015) emphasise crucial role of pedagogical relationships between teachers and their learners in teaching and learning, as both are active participants in classroom settings. Nevertheless, in reality, this is not always the case as Mani et al. (2020) argue that the majority of teachers employ a banking approach system in which learners are passive recipients of knowledge rather than actively participating in the teaching and learning process. Moreover, the literature further shows that qualified teachers' qualities and effective use of teaching-learning resources significantly influence their pedagogical relationships with learners and their academic performance (Iquabal et al., 2022). Therefore, History teachers should foster a positive pedagogical relationship with their learners to enhance their academic learning performance in public examinations.

As emphasised by the literature, an incompetence in English as a medium of instruction is not only a problem to the learners but also to their teachers (Mhonyiwa, 2014; Patrick, 2020). According to Martirosyan et al. (2015), self-perceived English proficiency is influenced by motivation, self-efficacy, confidence, and a positive attitude towards English language so that teachers are better equipped to confidently and fluently deliver the lesson in the classroom. In light of this, Tian et al. (2022) advise teachers to steadily raise learners' English literacy levels by incorporating more teaching tasks, and activities, concentrating on developing learners' language proficiency, critical thinking, and learning methods during the teaching process. The implication here is that History teachers must not only concentrate on the correct historical content of learners but they must also concentrate on their correct use of English.

Support from school management is another factor influencing performance of History learners in their public examinations. In response to this challenge, Nzoka and Orodho (2014) propose enhancing leadership skills through exhaustive training of principals, enabling former learners to form organisations for government funding subsidies, and providing operational guidance and counselling. Further, Senguo and Ilomo (2020) suggest that school frontrunners should enhance their managerial practices, involve parents in decision-making processes, and accept constructive

ideas from learners to enhance academic performance. Jackson and Andala (2022) emphasise the importance of effective school management strategies and peer learning implementation in enhancing learners' performance. According to Sulain (2024), school administrators should refrain from autocratic management, giving teachers complete control over their workloads, instead, monitor them to ensure that they are following the rules and perform their work effectively. Apparently, school managers should improve their practices by involving parents in decision-making, embracing learners' constructive ideas, implementing effective management strategies, and offering guidance and counselling for History learners.

The issue of lack of resources need to be considered for better performance of History learners in their public examinations. It is in this regard that, Okongo et al. (2015) recommend that the government should allocate more funds for the purchase of teaching and learning resources, infrastructure development, teacher training, and recruitment. Juliana (2020) highlights the importance of community donations, the schools' prudent use of funds, and their efficient distribution of resources for the benefits of learners. Putting that into consideration, Organisation for Economic Cooperation and Development (2020) emphasises the necessity for sufficient, quality, and effective use of material resources by teachers and school management to enhance teaching and learning. Sengai and Mokhele (2022) also suggest that the government may set up resource centrers where History teachers can borrow educational resources that are not always available in classrooms. Against this background, it would seem that in order to mitigate the performance of History learners, the government, community, school management, and teachers should collaborate to ensure the adequate quality, and efficient use of resources for History learners to improve their performance in public examinations.

The use of technology in the teaching of History has been shown to improve the performance of History learners in public examinations. Boadu et al. (2014) suggest that successful technology integration in History education requires teachers' familiarity with various technologies, motivational packages from managements of the schools, technology resources from the Ministry of Education and Training, and in-service training sessions. Musau and Abebe's (2015) study suggests that in order to help History teachers use information and communication technologies to enhance learners' performance, the government and non-government organisations should fund provision of computers and internet infrastructure to schools.

Furthermore, Monserate (2018) emphasises the importance of schools providing adequate computer use opportunities for learners, allowing learners to use technology independently, frequently, and providing teachers advanced technology training. With that being said, it seems that the government, non-government organisations, and school management should collaborate to provide schools with subsidised technological resources and infrastructures for the benefit of History learners. Therefore, the Ministry of Education and Training should offer History teachers in-service training on the effective use of technology in the teaching of history.

In addition, motivation is a very important factor in influencing performance of History learners in public examinations. Fundamentally, Annua-Sekyi (2015) suggests that effective teaching and learning requires teachers to motivate their learners. In support, Iquabal et al. (2022) concur that motivated learners learn better than demotivated ones due to their ability to adapt their strategies based on workload. Additionally, Kokoro (2021) suggests that boosting teachers' motivation and effectiveness can be achieved through adequate salaries, fringe benefits, and in-service training. From all appearances, teachers who motivate their learners can foster a positive attitude, leading to better work ethic, punctuality, effective learning, and higher marks on performance of History learners in public examinations. Bearing that in mind, it would seem that teachers should be motivated so that they could motivate their learners in the teaching and learning.

Vroom's Expectancy Theory suggests that teachers should be intrinsically motivated and make the best choices to minimise stress and maximise their satisfaction (Vroom, 1964). Further, Kokoro (2022) highlights that motivating teachers is essential to achieving organisational goals and providing quality instruction. This is due to the fact that motivated teachers are unquestionably pleased; as a result, they are motivated to perform their jobs competently and with greater effort in order to produce high quality results. This is supported by the study of Wikiesmann et al. (2021). The study suggests that teachers should motivate their learners through the various applications of teaching aids and learners-centred methods. Therefore, the teaching and learning in the absence of resources appears to be advanced by motivated History learners who can choose from a wide range of options. These options may seem to address some of the barriers to instruction in order to deliver necessary and appropriate knowledge for use in the teaching and learning of History, which will improve the performance of History learners in their public examinations.

Moreover, Ullah et al. (2018) advise that History teachers should model positive behaviour and attitude towards their learners in the classroom, as this encourages their learners to study subject. This is in line with Iquabal et al.'s (2022), declaration that History teachers with positive behaviour, and attitude can positively influence their learners' performance in their public examinations, while those with negative behaviour and attitude negatively affect their performance in public examinations. In support, Briones (2021), echoes that learners with good behaviour, and attitude perform better than those without these aspects. Dar (2022) emphasises the importance of teacher and learner motivation in creating a positive classroom environment that promotes healthy interactions and goal achievement. Therefore, both History teachers and learners are urged to engage in good behaviour and attitude as positive attitude and behaviour can lead to better performance of History learners in their public examinations.

It appears that qualified and highly educated History teachers are deemed capable of providing good quality information to their learners. However, some highly educated and qualified History teachers may struggle to master subject content, necessitating teacher training through several workshops as suggested by Ntabeni (2006) and Fru (2015b). Along the same vein, Ene et al. (2022) suggest that educational administrators and governments should organise in-service trainings, workshops, and seminars for History teachers to improve their qualitative development experience that would enable them to enhance the performance of their learners in public examinations. In support, Peter et al. (2023) emphasise the need for workshops and seminars to address learners' persistent failure in public examinations.

In the opinions of Asif et al. (2020), professional teachers often involve learners in learning through discussion, independent learning, exercise writing, and appropriate response writing. Further, Oscan (2021) acknowledges that professional teachers, unlike unprofessional ones, actively work to enhance their teaching and learning resources, monitor learning, and effectively explain complex concepts. Actually, Peter et al. (2023) suggest prioritising the employment of qualified teachers by the government to enhance their experience and improve learners' performance in public examinations. Therefore, it could seem that the efforts and contributions of these teachers in the teaching and learning of History might promote performance of learners in their public examinations. For that reason, professional teachers should further work hard, monitor learning, and explain difficult concepts so that they can be comprehensible to History learners.

Furthermore, according to Oscan (2021), improved family education levels, good income from good paying job and learners' good health positively affect learners' academic success and attitude towards learning, as educated, working parents and learners with good health often have a positive outlook towards school. To support this, Argon and Kiyici (2012) concur that educated families can explain complex concepts to their children. Emanuel (2017) emphasises the need for parental sensitisation to improve children's psychological status, reduce stigma, and address poverty as a root cause of poor performance in public examinations. Resultantly, parents' education level can effectively affect the performance of History learners in their public examinations, suggesting their collaboration with teachers to utilise their knowledge for academic assistance.

Apparently, peer pressure could affect academic performance of History learners in their public examinations. In conformity with that, Alzoubi (2021) posits that friends' influence is more potent than that of families due to their shared age and environment over specific periods. In order to address the challenge of peer pressure, Simba and Kabuka (2016) suggest the practice of disciplinary in the school environment, as schools with a strong disciplinary climate tend to have higher improvements in learners compared to those with less discipline. In alignment with this, Dagogo (2020) concedes that disciplined learners are able to pass their examinations than undisciplined learners. The aforementioned scholars seem right, as many History learners may struggle in their public examinations due to peer pressure from negative friends. Nonetheless, good friendship among disciplined History learners who frequently collaborate and support each other can significantly enhance their performance in public examinations.

According to Phaeton (2017), curriculum can be problematic since things do not always go as planned due to potential discrepancies between curriculum designers' expectations and actual classroom activities. Basic principles of curriculum alignment, as proposed by Ziebell and Clarke (2018), stipulate that learners' knowledge should align with their teachings and assessments to determine their achievement level. Meij and Merx's (2018) findings emphasise the importance of ensuring alignment between the planned (intended), implemented (enacted) and assessed curriculum. Therefore, it seems that there should be clear communication among curriculum designers, teachers, examiners, and markers to align examination questions set with content and marking schemes. It is for this reason that Sengai and Mokhele (2021) argue that policy-makers

should endeavour to include teachers in curriculum reforms in order to address their concerns, as they are essential to the successful implementation of curricula.

2.5 Empirical Review

Ntabeni (2006) conducted a qualitative study about the status of History teaching, learning, and examination results in Lesotho, 2000-2004, providing insight for teacher education. The study focused on the unsatisfactory status of History teaching and learning in Lesotho. The study utilised five-year documentations of examination results for data analysis. It observed a declining number of schools and learners that still offer History and unsatisfactory performance of learners as the two indicators of unsatisfactory status of history.

Ntabeni (2007) conducted another qualitative study about History teaching, learning, and Junior Certificate examination results in Lesotho, 2000-2006: Implications for teacher education. The study used a seven-year documentation of Junior Certificate History examination results. It revealed that unsatisfactory public examination performance of History learners in Junior Certificates is due to lack of essay writing skills, possibly due to insufficient teacher training of both qualified teachers and student teachers.

Another related study was carried by Lekhetho (2013). He conducted a qualitative study in Lesotho on stakeholder perspectives on strategies that can improve learner performance in school-leaving examinations in Lesotho. The study involved learners, teachers, deputy principals, school board members, and school inspectors as participants with data generated through questionnaires. The 25 schools were selected through stratified random sampling, while the learners were selected through random sampling, and the data was analysed using descriptive statistics. Unlike the studies of Ntabeni (2006) and (2007) that recommend in-service training for teachers, the Lekhetho's (2013) study suggests that enhancing academic performance requires collaboration, English Language culture, regular class attendance, purposeful leadership, discipline, and learner selectivity.

Again, Lebata (2014) conducted a qualitative study using semi-structured interviews and document analysis as the research instruments. The study analysed the performance of Biology 5090 learners in selected high schools in Lesotho, using General System Theory (GST). Similar to the findings of Ntabeni's (2006) and (2007) on History learners, Lebata 's (2014) study revealed that learners in Biology 5090 perform poorly due to factors like lack of resources, demotivation,

and ignorance of instructions, in-service teacher training, low classroom attendance, and inappropriate teaching strategies.

Additionally, Mhonyiwa's (2014) qualitative study used semi-structured interviews, focus group discussion, observation and documentary analysis to investigate factors influencing examination performance in commercial subjects in Tanzania. The study of Mhonyiwa (2014), unlike Ntabeni (2006) and (2007), Lekhetho (2013), and Lebata (2014), revealed that the qualifications of teachers significantly influence learners' performance in their public examinations. The study further showed that unqualified teachers, heads of department, and principals negatively affect the performance of learners due to lack of leadership skills and inclusive content knowledge, affecting delegating tasks to other learners.

Furthermore, Mwatwhana et al. (2014) conducted a qualitative study with the use of descriptive design in Kenya on the impact of History teaching/learning resources on learner performance. The study discovered that most Kenyan schools lack adequate History resources like textbooks, teachers' guides, maps, game facilities, and stationary, while minorities have sufficient resources. Further, the study discovered that few libraries have shortage of textbooks and journals due to insufficient funds for new publications. Apart from that, it was found out that most of the schools lack libraries due to financial constraints and school managements' prioritisation of learning and teaching resources. Therefore, unlike the other empirical reviewed studies, Mwathwana et al.'s (2014) attributed unsatisfactory performance of History learners in public examinations to the limited supply of resources.

Limene (2018) also conducted a mixed method study on examining factors influencing academic performance of grade 12 learners in English Language. The study showed that all interviewed teachers were young, aged 25-36, with one teacher being 37 or older. Despite the young and energetic teachers, there was no significant improvement in the performance of grade 12 learners. These teachers could have improved performance of grade 12 learners because according to Wanyama (2013), young teachers in most of the schools contribute positively to the performance of their learners.

Mani et al. (2020) carried out a qualitative study about assessing the factors affecting learners' academic performance in secondary schools in Kenema City. The participants were different teachers from different subjects including history. Differently from the other empirical reviewed

studies, the findings of Mani et al. (2020) demonstrated that teachers were using and mastering English as a language of instruction. However, the study revealed that despite teachers' mastery of English and school efforts to improve the English proficiency of learners, most learners struggle with English as a medium of instruction, as they are unable to speak and write English effectively. Therefore, learners' inability to master English led to significant loss of concepts in their lessons. For instance, one of the History teachers claimed that learners struggle to distinguish between the effects and reasons of the scramble for Africa due to their lack of English proficiency.

Fisher's (2023) qualitative study in the United States of America utilised semi-structured interviews with eight teachers, employing phenomenological design and thematic analysis for data analysis. The study was on the teachers' perceptions of learner performance on the Advanced English Language and Composition exam in Department of Defense Schools. The study revealed that various instructional and population factors affecting learners learning include reduced instructional time, lack of preparation, absenteeism, and reliance on computer-assisted software, COVID-19, and learners' lack of material knowledge.

2.3.1 Justification of gap for this study

The purpose of this study was to explore teachers' perceptions on performance in History public examinations in five high schools. For that matter, the study aimed to address three types of gaps identified in the empirical review which are knowledge gap, empirical gap, and theoretical gap. The studies reviewed lacked a comprehensive understanding of the perspectives of teachers regarding the public examinations' performance of History learners in Lesotho. Most studies reviewed were not specifically focused on the History subject, instead, they were about factors affecting learners' internal performance and were conducted outside the context of Lesotho. However, Ntabeni (2006) and (2007) conducted similar studies on performance in public examinations of History learners, focusing on the status of History teaching and learning.

Even though Lekhetho's (2013) study did not directly address History but instead concentrated on how learners' poor performance on public examinations may be improved, it is nonetheless comparable to this current study. Besides, the majority of the reviewed studies on performance in public examinations were attributed to other subjects such as commercial subjects and English Language, while some were not subject specific but general. A typical example includes the studies of Lekhetho (2013), Mhonyiwa (2014), Limene (2018) and Fisher (2023).

More importantly, most of the participants from the reviewed studies were parents, teachers, learners, and head of departments, deputy principals and principals. It is noted that previous reviewed studies did not employ Expectancy Theory and this implies that there are inconsistencies, limitations, and gaps regarding teachers' perceptions on performance in History public examinations in Lesotho as highlighted in the above paragraphs. Therefore, Expectancy Theory was employed in this study to help in addressing the identified gaps from the reviewed studies.

2.4 Conclusion

This chapter described the status of History in the national curriculum regarding the performance of learners. The chapter further explained the factors influencing the performance of History learners in public examinations. The factors were explained in detail on how they affected the performance of History learners both negatively and positively. Furthermore, the chapter provided information on how performance of History learners can be mitigated. Finally, the chapter reviewed empirical studies and was able to provide justification for the gap in the earlier studies.

CHAPTER THREE RESEARCH METHODOLOGY

3.1 Introduction

The purpose of this study was to explore teachers' perceptions on performance in History public examinations. The focus of this chapter is on the detailed elaboration of research methodology with reference to research paradigm, research approach, research design and population and sampling procedure in this study. The chapter further focuses on the research instruments used in this study, and how they were used to generate data. Apart from that, the chapter explains the process of data analysis in this study with the use of both inductive and deductive thematic analysis. Finally, the chapter discusses the trustworthiness of this study and its research ethics employed in the process of data generation.

3.2 Research Paradigm

The philosophical lens of this study was interpretivism paradigm. The central purpose of this paradigm in this study was to understand the subjective world of human experience (Gemma, 2018). Therefore, the paradigm in this study was used to help me to make meaning of data through intellectual and cognitive processing which were informed by my interaction with History teachers (Shard, 2018). This paradigm enabled me to interact with History teachers as my participants and hear their views about the performance of History learners in public examinations. Borrowing the views of Creswell (2018), this paradigm was used to help me understand and interpret the views and contextual meaning of the participants on performance in History public examinations. Therefore, interpretivism paradigm was used to understand teachers' viewpoints as well as their interpretations on performance of History learners in their public examinations.

Furthermore, the ontological assumption of this paradigm is that there are multiple perspectives on one incident and social reality, that is, people interpret and make their own meaning differently (Scotland, 2012). In this study, the paradigm was used to help me to acquire multiple views from the History teachers on the performance of History learners in public examinations with the focus on the status of History in national curriculum, factors influencing performance of History learners in their public examinations and mitigation on performance of History learners. It is for this reason that different History teachers from different high schools were selected to hear their different perspectives.

Fundamentally, I was engaged with teachers through dialogue, recording, questions, and listening during the data generation period. The adoption of the interpretivist paradigm helped me gain a thorough understanding of how History teachers perceive performance in History public examinations through their facial expressions, behaviours emotions and differing opinions. I was able to note down all these in my field notes to be used during interpretation. This enabled me to describe the reality of what actually happened and ultimately gain deep insight of the phenomenon. Again, it allowed me to reach conclusion regarding teachers' perceptions on the real status of History in national curriculum considering performance of History learners, major factors influencing performance of History learners and how poor performance of History learners in public examinations could be mitigated. However, the data generated by this paradigm is less likely to be generalised due to its independence on specific context and viewpoints.

3.3 Research approach

The study employed the qualitative approach. Qualitative research may be defined as a researcher's position and interpretive practices, aiming to make the world visible through their observations (Dentin & Lincoln, 2011). Further, Hamersley (2018) highlights that qualitative research involves generating and analysing non-numerical and unstructured data, emphasising the role of subjectivity in understanding concepts, opinions, or experiences about various naturally occurring cases. The study explored the differing perspectives of teachers across five schools regarding the performance of History learners in their public examinations, requiring interpretation by both participants and me and which is in support of Cohen et al. (2018) ideas that highlight ontological multiple realities in interpreting a situation by all parties involved in qualitative research.

Epistemologically, Creswell (2013) accentuates that qualitative research behaviour and data are socially situated, context-related, context-dependent, and context-rich, all influenced by the context in which they are conducted. In support, Cohen et al. (2018) assert that qualitative research consists of assumptions and interpretive contexts that inform the study of research problem in order to address the meaning which individuals assign to a social problem. As a result, in this study, I utilised a qualitative approach in order to understand, describe, and explain the multiple differing interpretations on the status of History in national curriculum regarding performance of History learners in their public examinations, its sources, consequences as well as its mitigation from the perspectives of different teachers from five high schools in Maseru district. This aligns with the

words of Cohen et al. (2018) which indicate that qualitative researchers concentrate on the subjective explanations, opinions, and interpretations of phenomena by participants.

Eventually, this study incorporated teachers' voices, researcher's reflection, a comprehensive description of History learners' performance in public examinations, a call for change, and a contribution to literature on the problem. In addition, the qualitative approach was used to gain a comprehensive understanding of performance in History public examinations from the perspectives of both the participants and myself. Cohen et al. (2018) emphasise the importance of qualitative approach in providing in-depth understanding of meanings, actions, non-observable and observable phenomena, attitudes, intentions, and behaviours. Therefore, qualitative approach assisted me to gain in-depth understanding of performance in History public examinations through History teachers' meanings, actions, behaviours, attitudes, intentions, and non-observable and observable sensations.

3.4 Research design

This study used the case study design which is defined as a design that enables investigation of a phenomenon in its environmental context using a diversity of data sources (Baxter & Jack, 2010). Similarly, McCombie's (2023) adds that the case study design is a detailed study of a specific subject, such as a person, group, place, event, organisation or phenomenon in order to describe, compare, evaluate and understand different aspects of a research problem.

Essentially, the five schools in Maseru district of Lesotho were used as multiple case studies in their multiple different experiences on performance in History public examinations. These cases were used in order to gain concrete, contextual and in-depth knowledge from the perspectives of teachers on performance of History learners in their public examinations as a problem in those five environments. The use of these five schools aimed to comprehend the variations and similarities among these cases, thereby enabling the discovery of contrasting outcomes. This helped me to gain a comprehensive understanding of teachers' perceptions on performance in History public examinations.

However, the process of generating data using the case study design was time-consuming for me due to the large amount of data that required sifting. By giving me erroneous and incomplete answers to the questions I asked, the participants also impacted the results. It took me a long time to confirm the information they gave me and had to ask them to provide it again. Notwithstanding

several constraints of the case study design the design enabled me to thoroughly delve into perceptions of History teachers on performance in History public examinations. The use of case study design further helped me to provide rich insights into the underlying factors influencing performance of History learners from the perspectives of different teachers from different schools.

3.5 Population and sampling procedure

Population refers to the entire group that a researcher needs to draw conclusions from whereas the sampling technique may be defined as the actual list of individuals that the sample will be drawn from (McCombie's, 2023). The target population of this study was seven secondary school History teachers and it was selected using purposive non-probability sampling technique. Purposive sampling is a sampling technique whereby participants are selected deliberately in order to provide important information that cannot be obtained from other choices (Taherdoost, 2020).

Therefore, in this study, History teachers were purposively selected because they have in-depth knowledge about public examinations performance of History learners as Cohen et al. (2018) state that purposive sampling is used to access knowledgeable people. Additionally, satisfactory and unsatisfactory performing schools were also purposively selected to compare their performance in public examinations of History learners. The study aimed to examine factors behind these performances, teaching and learning approaches, and how these approaches affect learners' performance and potential mitigation strategies on performance of History learners in public examinations. Purposive sampling, being qualitative in nature, helped me to explore the nuances of History teachers' experiences, opinions, and actions. Consequently, it enhanced my comprehension of how well History learners perform in public examinations as perceived by experienced and knowledgeable History teachers.

3.6 Research instruments

Data generation instruments are the tools that are used to measure, generate and analyse data related to the interests of the researcher's study (Duke University Libraries, 2023). In alignment with this, Insight (2022) affirms that research instruments are the tools used to measure, acquire and analyse data from subject around the research topic. This study used open-ended questionnaires, semi-structured interviews, and field notes to gather data on teachers' perspectives on the performance of History learners in their public examinations. The instruments are discussed in detail in the following paragraphs, respectively.

3.6.1 Open-ended questionnaires

Questioner Software (2022) defines open-ended questionnaires as free-form surveys that allow respondents to answer in open-text format, based on their complete knowledge and understanding. In addition, Shout (2020) adds that open-ended questionnaires are questions that allow participants to provide free-form responses in their own words, drawing on their own experiences. In this study, History teachers were freely responded to the questions about the status of History in the national curriculum and how History learners perform in public examinations based on their experience, knowledge and comprehension. Thus, as a consequence of the public examination performance of History learners, open-ended questionnaires were used to allow History teachers to express their thoughts, feelings and experiences in detail (Gharehbagh et al., 2022). This research generation tool was addressing research question one which was "What are the History teachers' views on the status of History in the national curriculum considering learners' performance in public examinations?"

In this study, open-ended questionnaires helped me to gather deep insights of data about the status of History from participants because participants elaborated on their experiences, thoughts, and feelings with provision of more examples. This was strengthened by my probing of participants through asking more questions so that they could elaborate more on the targeted responses especially where they provided short answers on the questions asked. On the other hand, this became a disadvantage to me because some of the responses of my participants deviated from the primary focus as some of the participants provided irrelevant information to the questions asked due to the freedom in answering with own words.

3.6.2 Semi-structured interviews

Cho (2018) defines semi-structured interviews as a qualitative data collection tool that combines pre-determined open questions for speedy discussion, allowing interviewers to explore specific themes and responses further. In addition, George (2023) adds that semi-structured interviews are a data collection technique that consistently asks questions that are not arranged within a pre-determined thematic framework. Therefore, in order to guide the conversation with the participants, I prepared a set of predetermined questions on the factors influencing performance of History learners in their public examinations and on the mitigation of poor performance of History learners in their public examinations.

The purpose of using semi-structured interviews in this study was to uncover teachers' perceptions, beliefs, values and emotions about performance in History public examinations and providing valuable insights into their subjective experiences. Semi-structured interviews were used in this study because they give comparable and reliable data as well as a room to ask follow-up questions because of the combination of structured and unstructured questions (Ruslin et al., 2022). In doing so, I was able to acquire in-depth information on research question 2 which was about perceptions of teachers on the factors influencing History learners' performance in public examinations and research question 3 which focused on the suggestions on mitigation of History learners' poor performance in public examinations.

More importantly, semi-structured interviews assisted me to explore specific areas of interest which in this case was teachers' perceptions on factors influencing performance of History learners in public examinations and mitigation of History learners' poor performance in public examinations. Therefore, due to its adaptability and flexibility, semi-structured interviews also assisted me in exploring unanticipated information and capturing intricate information. With relation to the performance of History learners in public examinations, the instrument also assisted me in facilitating a cooperative and co-constructive knowledge-building process by reflecting the complexity of teachers' lived experiences in the process of teaching.

I was able to probe History teachers further and go deeper into their narratives by using semistructured interviews, which enabled me to uncover hidden meanings and intricate details about performance of History learners in public examinations. This is because the open-ended format of these interviews encouraged detailed explanations, first-hand accounts, and contextual insights, which helped me to fully comprehend how teachers felt about performance in History public examinations. With that being said, I did not only hear teachers' thoughts about the performance of History learners in public examinations, but I also perceived their expressions, observed their feelings, and I was able to draw conclusions about their perceptions.

Probing participants, however, became disadvantageous to me as it took up a lot of time, and some of them ended up getting upset at me because they felt I had taken too much of their time and hindered some of their commitments. Some of the participants provided short answers because they seemed tired of being probed. For that reason, I ultimately decided to omit some of the

insightful probing questions in order to save participants from being held for an extended amount of time.

3.6.3 The field notes

According to Creswell (2013), the qualitative research method encourages researchers to take field notes to enhance data and provide rich context for analysis. Phillippi and Lauderdale (2018) define field notes as researcher's private, personal thoughts, ideas and queries regarding their observations and interviews. Additionally, field notes may also be defined as a record of activities observed and informal discussion from the field (Deggs & Hernandez, 2018). Further, the University of Southern California (2024) defines field notes as a researcher's note-taking during a field study, recording observations, behaviours, activities, and other features. Therefore, this study utilised field notes to document emerging issues, ideas, questions, and potential challenges encountered during data generation.

As identified by Neuman (2011), the type of field notes used in qualitative data collection consists of jotted data and direct observation written immediately after leaving the field. Field notes also consists of inference which involve reflecting social relationships, emotions and meanings, analysis which includes methodological strategies and theoretical notes, interview notes which involves the information about interview location and interviewee and personal journal which includes personal feelings and emotional reactions. Therefore, with the use of field notes, I wrote what I observed which is related to the performance of History learners at the five schools and what I observed during interviews. The observation that I included in the field notes was the matter of four selected schools having computer laboratories and one school having no computer laboratory.

In the field notes, I also recorded the expressions of teachers when talking about lack of History resources in their schools. One of the teachers even cried as she emphasised the issue of how difficult it is to be a one History teacher teaching all learners from grade 8 to grade 11. This crying implies that the participant was really overwhelmed with the workload and that affects her negatively. Therefore, scarcity of History teachers may result into overloading of work to some teachers and this may affect the performance of History learners negatively in public examinations.

According to Deggs and Hernandez (2018), field notes help in the creation of reasoned records and bring forth insightful qualitative data to ensure a complete understanding of the research

settings and phenomenon. In support, Phillippi and Lauderdale (2018) assert that field notes aid in creating a comprehensive, auditory, and detailed description of interviews' data, thereby enhancing participants' comprehension of the phenomenon. Therefore, using field notes helped me to attain complete understanding of the location of the five schools where data was generated and I also attained complete understanding about the perceptions of teachers on performance in History public examinations.

3.7 Data analysis

Thematic analysis was used to analyse data in this study. According to Braun and Clarke (2006), thematic analysis is a credible method for identifying and reporting themes (patterns), experiences, meanings and the realities of participants in data. Furthermore, Braun and Clarke (2012) define thematic analysis as a systematic method for identifying, organising, and providing insight into patterns of meaning across a dataset. In support, Crosley (2021) defines thematic analysis as the study of analysing the patterns and themes in dataset to uncover the underlying meaning. Similarly, Caulfield (2023) explains thematic analysis as a method of qualitative data analysis that involves closely examining data to identify common themes, ideas, topics, and patterns of meaning that come up repeatedly. Therefore, in the context of this study, manual thematic analysis was used to analyse data on teachers' perceptions on performance in History public examinations. The analysis was aimed to uncover underlying meaning about the status of History in national curriculum concerning the performance of History learners, factors influencing performance of History learners.

According to Braun and Clarke (2006), there are two approaches to thematic analysis which are inductive and deductive approaches. Inductive analysis is defined as a process of coding the data without trying to fit it into the analytic pre-conditions of the researcher whereas deductive thematic analysis is driven by the researcher's theoretical or existing knowledge (analytical interest) and the researcher comes to the data with some preconceived expected themes to be found reflected there Caulfield (2023).

In contrast to deductive thematic analysis, which is guided by the researcher's theoretical or prior knowledge and proceeds with the assumption that certain themes will be reflected on data, inductive thematic analysis is the process of coding the data and extract themes from participants' responses (Caulfield, 2023). Therefore, in this study, I employed both a manual inductive and

deductive thematic analysis because themes were derived from the existing knowledge and the responses of the participants by grouping the identical themes together. In other words, themes were not only derived from existing knowledge but they were also derived from the emerging responses of participants. I used a manual thematic analysis because it was accessible to me unlike software analysis which was difficult for me to access.

In this study, using both deductive and inductive thematic analysis was beneficial. Deductive approach helped me with a structured approach to analyse data about teachers' perceptions on performance in History public examinations. It also helped me to compare similar findings across studies that used similar codes on the literature. Using it helped me with the reduction of bias because the codes used are pre-determined. On the other hand, using inductive thematic analysis helped me to develop deep understanding of the data on the teachers' perceptions on performance in History public examinations. It also helped me with the identification of the new emerging themes that were not considered initially.

As identified by Brawn and Clarke (2006) and Caulfield (2023), thematic analysis comprises six stages namely: familiarisation, coding, generating themes, reviewing themes, defining and naming themes and writing up. In this study, I used and followed these processes manually to analyse data on teachers' perceptions on performance in History public examinations. Familiarisation is the initial stage of data analysis where I thoroughly understood and engaged with the data through active and frequent reading and transcription of verbal data into written form. The second stage is coding data whereby I highlighted sections of texts such as phrases or sentences and come up with written labels to describe their content.

Additionally, the third stage is generating themes which consist of sorting the varied codes into prospective themes and arranging all the relevant coded data extracts in the identified themes (Braun & Clarke, 2012). Here, I started analysing codes and examining how different they could combine to form a main single theme by examining the relationship between codes, themes, and different levels of themes. Here the initial codes may form main themes, whereas others may form sub-theme and others may be discarded (Caulfield, 2023).

The fourth stage of data thematic analysis involves reviewing themes, which involves re-reading the entire dataset for two purposes (Braun & Clark, 2012), firstly, for ascertaining whether the themes work in relation to the dataset or not and secondly, in order to code any additional data that

has been missed in earlier stages within themes. Therefore, here I reviewed identified themes by re-reading the entire data in order to see whether the themes are in line with the entire data or not and to code any missed data. Equally important, another phase is defining and naming themes in which the researcher identifies the core of each theme and the specific feature of the data that each theme captures (Braun & Clark, 2006). At this stage, I identified whether a theme contains any subtheme or not. The final stage is writing up which involves the final analysis and write up of the report (Caulfield, 2023). This analysis should provide a conscience, coherent, logical, non-repetitive and interesting account of the story the data tell within and across themes. This is where I wrote a report of what I found in the field about teachers' perceptions on performance in History public examinations.

Thematic analysis is appropriate for any study that seeks to discover using interpretations in order to provide a researcher with analytic tools to make sense of data (Maguire & Delahunt, 2017). Thematic analysis was employed in this study to discover in-depth understanding of teachers' perceptions on performance in History public examinations through careful reading and re-reading as well as interpretation of transcribed data from the aforementioned research questions in chapter one.

Principally, Crosley (2021) emphasises the importance of thematic analysis in examining participants' experiences, views and opinions using data derived from interviews, conversations, open-ended survey responses and social media posts. Therefore, in the context of this study, thematic analysis was used to explore teachers' views and opinions on the status of History in national curriculum, factors influencing History learners' performance in public examinations and how performance of History learners in public examinations should be mitigated. This was performed using semi-structured interviews and open-ended questionnaires. Principally, by using this analysis, I had an opportunity to link generated data from field notes with the opinions of teachers through semi-structured interviews and questionnaires about teachers' perceptions on performance in History public examinations specifically factors influencing performance of History learners. This supports the ideas of Ibrahim (2012) when he indicates that thematic analysis is capable of detecting and identifying factors that influence any issue generated by the researcher.

Moreover, thematic analysis was used because it offers flexibility in data interpretation and easy approach to large datasets by sorting them into broad themes (Caulfield, 2023). Despite its

flexibility, thematic analysis was time consuming and monotonous due to sifting through large volumes of text and manually assigning codes. However, I was urged to be patient and exercise caution in using thematic analysis to interpret teachers' responses on performance in History public examinations, as it is often subjective and based on the researcher's judgment. As a result, if I could not thoroughly consider data, I could miss its inference. For that reason, if not reflected very carefully, I may miss the inference in data. Contrary to this, I risked missing inference in data as thematic analysis enabled me to gain contextual understanding and in-depth exploration on performance in public examinations which enabled me to gain accurate interpretation and meaningful insights.

3.8 Trustworthiness of the research

As identified by Ghafouri and Ofoghi (2016), trustworthiness of the research refers to the degree of confidence in data, interpretation and methods used to ensure the quality of the study. As stated by Middleton (2023), trustworthiness is a very important aspect which is used to measure the quality of the research and how well the method and techniques will measure something. In this study, trustworthiness was used to determine how well open-ended questionnaires, semi-structured interviews and field-notes instruments measured perceptions of teachers on performance in History public examinations. In qualitative research like this one, trustworthiness consists of credibility, transferability, dependability and confirmability of the data and interpretations. Below are the detailed descriptions of four types of trustworthiness and how they were applied in this study.

3.8.1 Credibility

Ghafourri and Ofoghi (2016) define credibility as the extent to which the findings accurately represent the participants' experiences or views. Moreover, Connelly (2016) adds that credibility is the true value of qualitative research, indicating the accuracy and the correctness of the study's findings. Considering that, credibility can be enhanced through several strategies such as prolonged engagement, persistent observation, peer debriefing, member checking and triangulation (Ghafouri & Ofighi, 2016). Based on the ideas of Connelly (2016), prolonged engagement and persistent observation involve spending a significant amount of time in the field to allow the researcher to develop a deep understanding of the research context. In this study, I enhanced credibility by visiting the selected schools a month before I interviewed the participants

just to be familiar with them together with their working environment and build my trustworthiness with them. Then after that, I went to the field to generate data in the following month.

Peer debriefing consists of consulting and seeking feedback from fellow researchers who will review and assess transcripts, methodology and findings to gain different perspectives and reduce bias (Delve & Limpaecher, 2021). In order to do peer debriefing, I consulted my supervisor with my entire manuscript to check and review research methodology, and findings of the study and correct me. Not only that, I also consulted my classmates to check everything in methodology and findings and discussed with them. This helped me to gain different perspectives from different people and made a comprehensive conclusion in research methodology and findings and this helped me to reduce bias.

As per McKim (2023), member checking involves participants' evaluation of the researcher's interpretation to promote participants' validation whereas triangulation entails employing using several sources of data (data triangulation) or multiple researchers (investigator triangulation) to confirm findings. For member checking, I returned to the participants after transcribing data so that they could see whether I interpreted data as exactly as they answered the interviews. I even showed participants the findings of the study to see whether I interpreted their opinions on performance of History learners in public examinations as intended. In this study, to ensure trustworthiness through triangulation, I used several sources from the literature about performance of History learners in public examinations to see whether my findings would agree or disagree with the literature. The results were interpreted accurately and transparently without bias with the help of member triangulation, peer debriefing, and checking whereby the participants reviewed the researcher's interpretations. I also ensured credibility by sharing my ideas with other colleagues to gain different perspectives and I addressed any limitations and implications of this study.

3.8.2 Transferability

Transferability refers to how much the results of the study are applicable in other contexts, circumstances and settings (Connelly, 2016). In alignment with this, Ghafouri and Ofoghi (2016) assert that transferability symbolises that findings of the present research will be the same as findings in the same situations hence their suitability for use in future. That is, it provides sufficient details on the site, participants and methods or procedures used to collect data and this may be

achieved through comprehensive description of the findings from multiple data generation techniques (The Farnsworth Group, 2023). In this study, transferability was ensured by comprehensive description of the findings from questionnaires, semi-structured interviews, and the field notes on the teachers' perceptions on performance in History public examinations. I also ensured transferability by carefully and contextually defining the research questions before generating data, chose data generation methods and tools and piloted the research instruments to ensure their effectiveness.

3.8.3 Dependability

According to Elo et al., (2014), dependability refers to the stability of data over time and under various circumstances. Additionally, the Farnsworth Group (2023) adds that dependability is the measure of the consistency and reliability of the results of the study. In line with the above authors, Nguyen et al. (2021) confirm that dependability refers to the degree to which data generation and analysis are consistent. According to The Farnsworth Group (2023), dependability can be guaranteed by documenting the exact procedures followed for data generation, analysis, and interpretation as well as by providing enough relevant information to enable other researchers to repeat the study in an academic setting and produce consistent results. In order to ensure dependability, I wrote precise procedures which I followed in generating data in this study. I started with the paradigm, research approach, design, population sampling procedure, research instruments, data analysis procedures through six manual step processes. Trustworthiness of the research and research ethics protocols followed.

3.8.4 Confirmability

Shenton (2004) defines confirmability as the degree to which the research findings are objective and not influenced by the bias and beliefs of the researcher. Similarly, the Farnsworth Group (2023) affirms that conformability is a proof that qualitative research is neutral without being influenced by the assumptions or prejudices of the researchers. Bearing that in mind, confirmability in this study was ensured through my justification of theoretical, methodological and analytical choices throughout the study to provide an understanding of how and why decisions were taken (Alele & Malau-Aduli, 2023). In other words, I ensured conformability of this study by providing justification of why I used Expectancy Theory, interpretivism paradigm, qualitative approach, case study design, open-ended questioners, semi-structured interviews, filed notes and

thematic analysis in this study. Furthermore, justification of trustworthiness and research ethics also ensured confirmability of the study.

Additionally, Applied Doctoral Centrer (2024) concurs that confirmability is assured through audit trail, triangulation, reflexivity, member checking, data checking and re-checking during data generation and analysis. An audit trail is a clear coding schema that defines the codes and patterns found in analysis. I also ensured confirmability through submission of data to my participants to check it and see whether I presented it as exactly as what they had said. I also checked and rechecked data during the data generation process and I ultimately ensured confirmability of this study by providing clear coding schema through detailed explanation of codes and patterns found during thematic analysis process.

3. 9 Research ethics

Research ethics are the set of principles and guidelines that shape and guide researchers on how scientific research should be designed, conducted, managed, used and disseminated (Wagle, 2020). In addition, Davis and Lachlan (2017) define research ethics as the scientific principles, rules, guidelines and norms of research-related behaviour that a research community has decided are proper, fair and appropriate. Therefore, I followed all the ethical protocols in order to enhance the trustworthiness of the research, maintain scientific academic integrity and protect the rights of the History teachers as the participants of this study and to conduct a valid, reliable, legitimate and representative research (Bhandari, 2021). Following all the research protocols helped me to expand knowledge on the main objective of this research which is exploring teachers' perceptions on performance in History public examinations in three high schools in Maseru.

First and foremost, I collected an introductory letter from the Faculty of Education and submitted it to the management of the targeted schools in order to seek permission to be allowed to generate data from those targeted schools. I also designed my own consent forms and gave them to participants before participating in this study. After that, ethical issues such as voluntary participation, informed consent, anonymity, confidentiality, potential for harm and result communication were considered (Cohen et al., 2018). Below is the detail explanation of each ethical principle and how they were used in this study.

3.9.1 Voluntary participation

According to Bhandari (2023), voluntary participation implies that all participants choose to participate without any pressure and they are free to withdraw from the participation at any point without feeling an obligation to continue to participate. Putting that into consideration, I respected the decisions of the participants by not pressurising them to participate in the study and by allowing them to withdraw from participating at any time without providing any valid reason if they wish to withdraw. As part of ensuring voluntary participation, I also conscientised the participants that there are no negative consequences to their refusal to participate in the study.

3.9.2 Informed consent

Informed consent is defined as a condition whereby all prospective participants obtain and understand all the information needed to decide whether they want to participate or not (Bhandari, 2023). According to Fleming (2018), there must be oral explanation of the informed consent to the participants before they participate whereby they are fully informed of what will be asked of them, how the data will be used, benefits of the study, risks, funding, institutional approval and what could be the consequences of their participation. As a result, after teachers agreed to participate in this study, they were allowed to willingly sign a consent form before participating and this informed consent was used as a contract between me and the participants. After that, I explained to the participants the purpose of the study, benefits of the study, made them aware of the risks that may occur during data generation, I conscientised them that there would be no monetary benefits for their participation in this study.

3.9.3 Anonymity

Based on the ideas of Fleming (2018), anonymity in research refers to the idea that participants in the study should not be able to be identified personally when their data is analysed, and that researchers should not record any direct or indirect personal identifiers that can be connected to any research participant. In this study, I ensured the anonymity by not collecting any personally identifying teacher participants' information such as names, phone numbers, email addresses, physical characteristics, photos, and videos (Bhandari, 2023). Instead, I kept the names of all the participants anonymous and I replaced their real names with pseudonyms whereby I named them participant 1, 2, 3, 4, 5, 6, and 7.

3.9.4 Confidentiality

According to Evergreen State College (2024), confidentiality refers to the condition in which the researcher knows who the participants are but removes all the identifying information from his report to protect their rights of privacy from others. Oxfam International (2020) recognises that confidentiality entails careful management of data such as storing data securely and separating identifiable information from other data. In this study, the researcher ensured confidentiality by keeping the records of generated data secure through the use of password protected files. The information that was discussed between the research and participants was kept confidential by not allowing anybody besides the participants to access generated data. I also clearly elaborated to the participants how generated data would be used and I stored it in my personal computer which is locked with a password and nobody could access it.

3.9.5 Potential for harm

This is whereby the researcher considers every possible source of harm to participants and concrete ways to mitigate them (Fleming, 2018). In the context of this study, I involved my supervisor to discuss steps that can be taken to reduce harm such as counselling services. According to Fleming (2018), harm can be in many different ways such as psychological, social, physical, legal harms, reputational, and resource loss (time). Psychological harm may occur when sensitive questions or tasks may initiate negative emotions such as shame or anxiety. Harm can also be social harm whereby participants can involve social risks and public embarrassment (stigma). Another essential harm to be considered is physical harm whereby pain or injury can result from the study procedures (Bhandari, 2023). In this study, I encountered psychological harm with one participant who became emotionally affected during interviews and had to provide counselling for her. She was bothered more about her working environment which seemed unconducive for her due to the principal ill-treating her while at the same time she was the only History teacher in that school.

3.9.6 Results communication

It involves good science communication as being honest, reliable, and credible in order to make the results transparent (Bhandari, 2023). This could be done through the avoidance of plagiarism and research misconduct. According to Bhandari (2023), plagiarism is defined as copying someone else's work without proper citation and reference while research misconduct refers to falsifying data, manipulating data analysis or misrepresenting results in research reports. Therefore, in this

study, I made the results to be transparent by ensuring that the study is free from plagiarism and by avoiding research misconduct through provision of proper citations and references.

3.10 Conclusion

In this chapter, the research methodology employed was elaborated on throughout the entire study. Firstly, the chapter explained the research paradigm, research approach, research design, and population sampling. The interpretivism paradigm was used as a philosophical lens, a qualitative approach was employed, and a case study design was used. Basically, the research approach and design used in the study were judged pertinent to the purpose of the study because they effectively addressed the research questions, providing a deeper understanding of History teachers' perceptions of their learners on performance in History public examinations. The chapter further explained how open-ended questionnaires, semi-structured interviews, and field notes as research instruments for this study were used in generating data. Apart from that, the study explained how the data was analysed using both inductive and deductive thematic analysis. Finally, the chapter explained the trustworthiness of this study and the research ethics addressed in it.

CHAPTER FOUR DATA PRESENTATION AND ANALYSIS

4.1 Introduction

This chapter reports the findings on the teachers' perceptions on performance in History public examinations in five high schools in Maseru. The findings are presented in three broad themes according to the three research questions. The themes are status of History in national curriculum, factors influencing performance of History learners in public examinations and mitigation of History learners' poor performance in public examinations. Seven teachers provided their views through questionnaires on the status of History in the national curriculum considering learners' performance in public examinations. The same teachers were interviewed and they provided their perspectives on the factors influencing the performance of History learners in public examinations and how History learners' performance can be enhanced. The chapter starts with the demographic profile of participants. The data is presented using the tables from Microsoft Excel and Microsoft Word.

4.2 Demographic profile of participants

This section presents the demographic profile of seven interviewed teachers. This includes data about the characteristics of each participant. This might help the reader to understand what kinds of participants were dealt with. The demographic profile could also help to shed light on particular characteristics that may help me to be able to analyse data by relating those unique qualities with the findings of the participants. In this study, demographic profile includes the age of the participants, their highest teaching qualification, teaching experiences and any other position besides teaching. More importantly, teaching qualifications and experiences of teachers were used as part of data analysis on factors influencing performance of History learners in public examinations.

Table 1: Demographic Profile of Participants

Demographic variables	Participants	Frequency
Age	25-30	2
	31-35	1
	36-40	1
	41-45	1
	46-50	2
Total		7
Highest teaching qualifications	Bachelor's degree	5
	Postgraduate diploma	1
	Diploma	1
Total		7
Teaching experiences	1 year	1
	2 years	1
	3 years	1
	4 years	1
	10 years	1
	15 years	2
Total		7
Other position besides teaching	None	4
	Marker	1
	Administrative committee	1
	Examiner	1
Total		7

The data presentation in Table 1 above demonstrates the seven participants' responses regarding the benefits of History in Lesotho's national curriculum. The participants were asked the following question, "What do you think could be some benefits of History in National curriculum of Lesotho?" The above data shows that out of 7 participants, 5 were below 45 years old, whereas 2 were above 45 years old. The young participants' age ranged between 25 and 40, whereas the old participants' age ranged between 46 and 50.

Apart from that, five participants had the highest qualification of a bachelor's degree, while the other two had postgraduate diplomas and diplomas, respectively. Therefore, the findings of the study showed that the majority of the interviewed teachers were young and qualified. This could affect the performance of History learners positively because young teachers are more energetic and effective than old teachers. Besides, all the teachers were qualified to teach History to high school learners, although their qualifications differed, as the majority of teachers had the highest

qualification of degree, while one teacher had a postgraduate diploma and the remaining one had a diploma.

These various qualifications of History teachers may affect the performance of History learners differently, as postgraduate diploma holders are more advanced than bachelor's degree holders, whereas bachelor's degree holders are more advanced than diploma holders regarding knowledge impartation and implementation in the process of teaching and learning. The qualifications of all these teachers could have allowed them to improve the performance of History learners.

Further, four participants had less than five years of teaching experience, whereas two teachers had 15 years of teaching experience and one teacher has 10 years of teaching experience. The findings further revealed that among the interviewed teachers, there was a marker, an examiner, and a school committee administrator. The study demonstrated that the majority of the interviewed teachers had less than 5 years of teaching experience. On account of this, little teaching experience may influence the performance of high school History learners negatively because teachers with little teaching experience may not deliver History content as effectively as experienced teachers, and they still need to be monitored by the principals and experienced teachers.

On the other hand, more experienced History teachers may be able to help their learners perform better on History public examinations because they are more knowledgeable and competent in teaching historical content and are able to effectively and confidently impart it to learners. In doing so, they can improve the performance of History learners. Contrary to this, it may not always be the case, as in some schools, that less experienced teachers may affect the performance of History learners more positively than experienced teachers because they may work very hard. Some experienced History teachers may not put in enough preparation since they believe they already know everything and do not need to put in extra effort. Some may even miss classes frequently, which could have a detrimental impact on how well History learners perform in their public examinations.

4.3 Presentation of findings on the objectives

Data presentation and analysis on the objectives is categorised into two. The first part of analysis is from research question one and the data is retrieved from the open-ended questionnaires. The second part of data presentation and analysis is from research questions two and three as retrieved from semi-structured interviews.

4.4 Data presentation and analysis on open-ended questionnaires

This section presents the findings regarding the current status of History in the national curriculum of Lesotho considering learners' performance in public examinations. To put it in simpler words, this section presents the presentation and analysis on the views and opinions of participants from the questionnaires of research question 1. The research question read as follows: "What are the History teachers' views on the status of History in national curriculum considering learners' performance in public examinations?" The section starts with the benefits of History in national curriculum of Lesotho, the current number of History learners, general performance of History learners in 2023 and reasons for the poor performance of History learners in their public examinations.

4.4.1 Status of History in the national curriculum of Lesotho

This section presents the status of History in the national curriculum regarding learners' performance in public examinations. It is better to put the status of History in national curriculum into consideration because it is affected by the performance in History learners in public examinations. Since the subject seems to be performed poorly, this may result into bad and neglected status of the subject which may ultimately affect the attitude of both teachers and learners negatively. This negative attitude may further influence the performance of History learners hence the importance of highlighting the benefits of History subject in national curriculum as explained in detail in the following paragraph.

4.4.2 Benefits of History in national curriculum

Table 2:Benefits of History in National Curriculum

Participants	Category	Participant Quote
Participant 1	Moral values (empathy, and forgiveness), identity (genealogy), citizenship values, patriotism (nationalism), and community awareness.	"It teaches moral values, identity, citizenship values, patriotism and empathy, develops unity amongst the learners."
Participant 2	Patriotism (nationalism, peace, unity, and nation development), and moral values (empathy, and forgiveness).	"Learners will develop patriotism, empathy and forgiveness values and this will help the country to develop peacefully."
Participant 3	Patriotism (nationalism and nation development).	"it could bring back our love for the country that will help learners to think on their country's development."
Participant 4	Patriotism	"It could potentially develop very much needed patriotism in Lesotho"
Participant 5	Global awareness (community, and national awareness), development of learners' critical thinking skills, patriotism (collaboration, unity), moral values (appreciation of different cultures, values and norms).	"Learners know their History and History of other people, it teaches learners to know important current events in other countries such as the ongoing wars in Ukraine and Russia Learners learn and appreciate different cultures, values and norms"
Participant 6	Patriotism	"It instils patriotism to the kids."
Participant 7	Global awareness of events, learners' critical thinking skills, reasoning skills (learners' argumentative skills, and learners' informed decisions).	"It teaches learners to be aware of the world-wide events It helps learners to develop reasoning skills" "It helps learners to have their own views without being easily manipulated."

From the two participants, as indicated in Table 2 above, it shows that History instils moral values like empathy, forgiveness, and appreciation for diverse cultures, norms, and values in learners. Participant 1 said, "It teaches moral values, identity, citizenship values, patriotism, empathy, and unity among the learners as they grow up knowing where they come from." The other two participants mentioned that History improves the critical thinking skills of learners. Participant 5 emphasised, "Learners will know their History and the History of others... develop critical

thinking skills through essay writing... Learn and appreciate different cultures, values, and norms. develop values such as patriotism, collaboration, and unity."

Participants further said that studying History is advantageous because it instils in learners' citizenship values, identity awareness, enhances their reasoning and arguing skills, ability to make well-informed decisions, and awareness of community, national, and global events. Participant 7 further claimed:

History teaches learners to be aware of world-wide events and to note important current events in other countries, such as the ongoing wars in Ukraine and Russia. It also helps learners develop reasoning skills because the questions they are asked involve the argumentative essay section, where learners are asked to what extent a certain thing is beneficial to Basotho, for example.

Therefore, the majority of the participants showed the benefits of History in the Lesotho national curriculum, with patriotism receiving the most attention as they discussed and elaborated on it, with the exception of one participant who spoke about reasoning skills, critical thinking skills, and global awareness. It is for this reason that this study asserts that History is beneficial in the national curriculum of Lesotho as it helps learners develop moral values, citizenship values, patriotism, identity awareness, global awareness, critical thinking skills, and reasoning skills. This being the case, considering the benefits of History in the national curriculum of Lesotho can improve the status of the subject. Improving the status of History may result in a positive attitude of teachers and learners towards the subject; hence, teachers may work tirelessly to ensure that learners perform it well. Simultaneously, learners may have more time towards the subject; they may have an interest in learning; and they may work hard to ensure that they perform it well in their public examinations.

4.4.3 Current number of History learners

Table 3:Current Number of History learners

Participants	Number of learners	Decline/increase
Participant 1	20	Declining
Participant 2	15	Increasing
Participant 3	16	Declining
Participant 4	Less than 65	Increasing
Participant 5	22	Declining
Participant 6	20	Declining

The above data from Table 3 displays the number of grade 11 History learners who are expected to write their public examinations in 2024. Participants 1 and 6 were from the same school and participants 4 and 7 were also from the same school, while all the remaining participants were from different schools. Participants were asked the following question: "In comparison to prior years, what do you think could be the cause of the decreasing or increasing of the number of grade 11 learners?"

The data showed a decrease in grade 11 History learners expected to take the 2024 examinations, except for one school from participant 4 and participant 7, with less than 65 learners. The other schools have 20, 15, 16, and 22 grade 11 History learners, respectively. Therefore, the data and presentation from Table 3 show that, while the participants from two schools revealed an increase in grade 11 History learners, the majority of participants in three schools reported that the number of grade 11 History learners was declining in comparison to prior years. Therefore, the decline in the number of History learners implies that there are few learners who opt for history. This may be because of the consistently unsatisfactory performance of History learners and the false belief that History is a failing subject. This decrease may further contribute to the unsatisfactory status of History in the national curriculum of Lesotho.

On the question of the reasons that might have led to the decline in the number of History learners, participant 3 said, "Learners are declining because there is a claim that the subject is not performing well." On a similar issue, participant 1 answered: "To be honest, the numbers are decreasing year after year because the subject has not received any justice from the Ministry of Education and Training or government, so learners become reluctant to take the subject due to its poor performance." The injustice the participant is talking about may be the lack of employed qualified History teachers by the government, the lack of provision of History textbooks by the

government, and the exclusion of History teachers in setting History examination questions. Consequently, this injustice may affect the performance of History learners negatively in their public examinations if not considered.

On the other hand, participant 4 said, "The number is increasing because History is compulsory in this institution I am teaching." It would seem that in institutions where History is compulsory, the number of learners who are doing History is higher than in schools where it is not compulsory. This is because in schools where History is compulsory, all grade 11 learners will have to do it whether they like it or not, while in schools where History is elective, most learners will not select it due to several reasons, such as its already unsatisfactory status due to its poor performance in public examinations. This implies that History being compulsory in schools can make the subject more popular and recognized in the national curriculum of Lesotho.

4.4.4 General performance of History learners in public examinations of 2023

Table 4:General Performance of History learners in their Public Examinations of 2023

Participant	Performance	Teachers'	of History
S		feelings towards	
		performance	
Participant	Poor	Not satisfied	"This subject would be eliminated from the curriculum and learners will
1			continue to hate the subject."
Participant 2	Poor	Not satisfied	"It is completely going to be removed from the school curriculum."
Participant 3	Poor	Not satisfied	"It shall lack learners."
Participant 4	Poor	Not satisfied	"As is already happening, phasing out of the subject in secondary schools."
Participant 5	Good	Satisfied	"Schools will exempt it from their curriculum since it is an elective subject."
Participant 6	Poor	Not satisfied	"Schools will cancel the subject."

Participant	Poor	Not satisfied	"It is already happening that many schools have deleted history, and it
7			will end up being removed from the curriculum. I witnessed this happen
			while serving as president of the Lesotho History Teachers Association
			and from 2007 History topped being compulsory in schools."

The data presentation from Table 4 above shows the participants' views on the general performance of History learners in their public examinations in 2023. Firstly, the participants were asked the following question, based on your opinions, is the performance of History in public examinations poor or good? Are you satisfied with your performance, and why? Six participants from four schools, as indicated by the above table, reported poor performance by History learners in the 2023 final examinations as compared to previous years, while one participant rated their performance as good. Consequently, the analysis revealed that History learners consistently perform poorly in public examinations. Therefore, it seems that it is this consistent poor performance that might have contributed to the unsatisfactory status of History, as indicated by Ntabeni (2006) and Fru (2015a).

Concerning the questionnaires asking participants about their satisfaction with learners' performance in their public examinations and the implication of the continuous poor performance of History learners in public examinations, the analysis revealed that six respondents expressed dissatisfaction, with the exception of one participant in a school that performed well. The majority of participants were therefore not satisfied with the performance, except for one participant, as mentioned above. Thus, it appears that the participant who was satisfied was motivated by the learners' performance in his school, while the participants who were not satisfied were demotivated by the learners' performance in their schools. This highlights the significance of Expectancy Theory as employed in this study. For the demotivation of History teachers, the theory guides them to encourage their learners to perform better in public examinations by making the best decisions that can inspire them as well as their learners even in times of unsatisfactory performance (Vroom, 1964).

Four participants expressed dissatisfaction with the way History learners perform in public examinations, fearing the removal of the subject from the curriculum. Participant 1 said, "I am not satisfied because this may lead to the reluctance of learners to take the subject, and the poor performance puts the subject under the pressure of being eliminated from the curriculum."

Another participant expressed dissatisfaction, citing the demotivation that comes with poor performance by History learners in public examinations. Moreover, participant 3 said, "I am not satisfied because most of the learners are failing, and this hinders my interest in teaching History." The demotivation of History teachers may affect their learners as well as their performance in History public examinations, hence the importance of motivation for both History teachers and their learners.

Additionally, participant 6 expressed dissatisfactions with the new curriculum. She claimed, "I am not satisfied with the new curriculum because learners fail to handle the content delivered because they come to high school level young and not well prepared." Due to the introduction of automatic promotion in primary schools in Lesotho, it seems that most of the learners come to high school unprepared and without knowing how to write, and this affects the performance of History learners negatively in public examinations. This is because it may be difficult and time consuming for high school History teachers to train learners on how to write, and it may ultimately lead to teachers not covering historical content very well, hence the negative influence on History public examinations.

On the other hand, participant 5 said, "I am good since History teachers receive content, methodology, and assessment training from the Lesotho History Teachers Association and Ministry of Education and Training annually." The implication here is that; this training might have resulted in the good performance of History learners in public examinations. However, this was in contradiction with most of the participants, who claimed a lack of training for History teachers. Although this participant expressed satisfaction with the performance of History learners in public examinations, he warned against removing the subject from the syllabus if it continued to be poorly performed. Concerning the implications of History if it consistently performs unsatisfactorily, this study revealed that all participants feared that the subject may be removed from the curriculum. It is understood that History was compulsory in some schools in previous years, but due to its higher failure rate, most of the schools started removing it from the curriculum, as indicated by participant 7 in the above table and Ntabeni's (2007) study.

4.4.5 The reasons for the poor performance of History learners in public examinations

Table 5:Reasons for the Poor Performance of History learners in Public Examinations

Participant Categories

Participants Quote

S

Participant	Poor teaching strategies, the limited time slots allocated,	"the poor teaching strategies employed by the
1	lack of textbooks and Information and Communication	teachers, absence of resources such as
	Technology (ICT) gargets.	textbooks and ICT gargets to teach the subject."
Participant	Shortage of History teachers, demotivation of History	"There is only one teacher teaching the subject,
2	teachers, lack of technology, lack of content knowledge by	teachers are demotivated, lack of History
	History teachers, frequent shift in marking of history, lack	textbooks, lack of technology in the teaching of
	of History textbooks, miscommunication between	History, lack of content knowledge by other
	Examination Council of Lesotho (ECOL) and History	History teachers and change in marking by
	teachers.	ECOL."
Participant	Learners' lack of interest in History.	"Learners' lack of interest in History and they
3		complain that it teaches only about the past
		which does not participate in the economic
		aspects."
Participant	Incompetent teachers, new ways of marking History by	"Incompetent teachers, learners would struggle
4	ECOL, learners' struggle in answering the questions, the	to maintain balance in responses, vague
	difficult introduction of answering History questions by	mannerism of marking ECOL-based exams. "
	ECOL	
Participant	Problem in essay writing, lack of relevant textbooks, false	"Essay writing due to poor linguistic ability in
5	belief that History is a failing subject.	English Language, lack of relevant
		textbooksfalse belief that History is a failing
		subject."
Participant	Misunderstanding of English Language by History	"They do not understand English Language and
6	learners, miscommunication between ECOL and History	the new way of answering the questions is quiet
	teachers, lack of History teachers training on marking by	tricky. ECOL does not inform us before they
	ECOL, the difficult introduction of answering History	introduce their new ways of markingECOL is
	questions by ECOL.	not even sure that we are doing it right because
		of lack of training given on marking."
Participant	The type of learners, the shift in marking by ECOL.	
7		"I think is the issue of the type of learners we are
		teaching. Another issue was the shift in marking
		by ECOL which resulted in new ways of
		4

teaching."

The above presentation in Table 5 demonstrates the opinions of teachers on the reasons for the poor performance of History learners in public examinations. The question that was asked was, "In your opinion, what could be this kind of performance?" Participants 2 and 6 mentioned miscommunications between ECOL and History teachers. Miscommunication between ECOL and History teachers, as well as evolving marking standard by ECOL, may result in a misunderstanding of consensus in setting the History examinations based on the taught content by History teachers. This may result in learners being asked questions on topics they are not taught, resulting in their failure to answer those questions correctly, which may negatively affect their performance in public examinations.

Participant 2 also mentioned the demotivation of History teachers, the lack of History teachers, and relevant History textbooks. It would appear that a shortage of History teachers may affect the performance of History learners in public examinations because the few available teachers may be overloaded with the work and fail to deliver the historical content effectively to their learners.

Participant 3 further mentioned a lack of content knowledge and the demotivation of History teachers. History teachers with a lack of content knowledge mean that they cannot deliver historical content to their learners effectively; hence, this may affect performance in History public examinations. Besides, demotivated History teachers may also affect the performance of their learners negatively because they may not interact and relate well with their learners, other teachers, and the principals.

Participant 1 mentioned poor teaching strategies used by other teachers and limited time slots allocated for history. Participant 1 and participant 2 also mentioned a lack of ICT. Lack of technology and relevant History textbooks in the teaching of History may affect the performance of History learners in that teaching and learning without them may not be effective or engaging. Instead, it may become demotivating. Therefore, demotivated students may not work hard to achieve better performance in their public examinations. Further, participants 1, 2, and 5 mentioned a lack of relevant History textbooks. Aside from that, a lack of relevant History textbooks may result in a lack of instructional materials for the teaching and learning of History, leading to failure in retention and understanding by History learners, as emphasised by Sengai and Mokhele (2022).

Participants 2, 6, and 7 mentioned shifts in marking as one of the reasons for the poor performance of History learners. It would seem that this consistent shift in marking by ECOL may not only confuse History teachers; it may also affect the performance of History learners negatively in their public examinations because teachers are not even provided with proper training for these new ways of marking.

Additionally, participants 4 and 6 mentioned the difficult introduction of answering the questions by ECOL. This new way of answering the questions implies a shift in marking and teaching processes. It may shift the attention of History teachers in their teaching, and this may even demotivate them due to their exclusion. Participant 4 further echoed incompetent History teachers and learners' struggles in answering the questions. It is inferred that, incompetent History teachers in the teaching of History may not deliver historical content well to their learners, leading to failure for History learners and ultimately influencing their performance negatively in public examinations. Therefore, such kinds of teachers require training through several workshops. Learners who struggle with answering the questions may not be able to answer the questions correctly in their public examinations, hence the need for several provision of assessments for them.

Participant 5 further stated the problem of essay writing. Learners who struggle with writing essays in History may not perform well in their public examinations, and this requires serious attention because public examinations are written, not oral. Participant 6 stated misunderstandings of the English language by History learners and the lack of History teacher training on marking by ECOL. Misunderstanding of English by History learners means that they cannot understand the examination questions that are written in English; hence, this may result in their failure to understand the questions clearly. This may ultimately contribute negatively to their performance in public examinations.

Participant 7 also stated that the type of learners is one of the reasons for the poor performance of History learners in public examinations. By type of learners, participant was referring to slow and fast learners. It seems like nowadays learners are slow and lazy in doing their work. Therefore, these learners may not do well in their public examinations because they are always pushed to do their work. Even though these factors appear to be dominant than others, they were nevertheless responsible for the low performance of History learners in their public examinations.

4.5 Data presentation and analysis on semi-structured interviews

This section present data from semi-structured interviews for research questions two and three. The research question 2 was "How do History teachers perceive the factors influencing performance of History learners in public examinations?" The research question 3 was: "What are History teachers' suggestions on mitigation of History learners' poor performance in public examinations?" For research question two, the section presents data from the perspectives of teachers about the factors influencing performance of History learners in public examinations. Concerning research question three, the perceptions of teachers on the mitigation of History learners' poor performance in public examinations was presented and analysed.

In this study, the several factors influencing the performance of History learners were presented and analysed. Such factors include motivation of History teachers, History teachers' relationship with principals, other teachers and learners, lesson preparation (lesson planning and scheming), use of teaching aids, use of teaching strategies, assessment methods and immediate feedback. Apart from that, the section discussed the challenges encountered by History learners at school and in learning as well as their solutions. Finally, the opinions of teachers on the mitigation on the poor performance of History learners in public examinations were discussed.

4.5.1 Motivation of History teachers in the teaching of history *Table 6:Motivation of History teachers in the Teaching of History*

Participants	Feelings	Reasons
	Towards	
	Teaching of	
	History	
Participant	Motivated	"I am motivated teaching learners about their identity, teaching them about where they
1		come from and who they are is the interesting thing to do."
Participant	Motivated	"In teaching History, I learn the stories of the different people from different countries
2		and if I pass that knowledge to my learners I become motivated. I am also motivated
		when we take educational trips like school History fair because we meet other teachers
		and learners and we explore historical places."
Participant	Motivated	"I am motivated when dealing with Lesotho History where we will be gaining knowledge
3		on Moshoeshoe I's background and how he built Basotho nation and that bring
		brilliance in us."

Participant	Motivated	"I am motivated because I love History and I am urging to instil needed patriotism in
4		learners."
Participant	Motivated	"Passion for the subject and values such as peace, resilience, patriotism History
5		inculcates."
Participant	Motivated	"I am motivated by patriotism."
6		
Participant	Motivated	"I enjoy teaching of History because my learners also enjoy learning the subject."
7		

Table 6 exhibits data on the motivation of participants in the teaching of history, indicating whether they are motivated or not. All seven participants expressed their motivation for teaching history, but what motivated them varied. Participant 1 claimed that he is motivated by teaching learners their identity, which is where they come from and who they are. Therefore, if the teacher is motivated by that topic, it means some other topics demotivate him and may not be able to impart historical content on such topics very well to learners, hence affecting their performance in public examinations.

Secondly, participant 2 is motivated by learning and sharing with learners' diverse stories from various countries and participating in educational trips like the History fair, where they interact with other History teachers and learners. This participant is motivated by learning historical concepts and sharing them with her learners, and this may enhance her content knowledge and ultimately contribute to the positive performance of History learners in their public examinations. Apart from that, a teacher is motivated by interacting with other teachers and learners from other schools during educational trips. This may improve the performance of History learners in public examinations because education trips provide practical knowledge not only to the teachers but also to the learners as they learn through experience, where they perceive what they were taught as reality.

Participant 3 expressed motivation for understanding Lesotho history, particularly Moshoeshoe I's background and his role in building the Basotho nation. This participant is also motivated while teaching the topic of Moshoeshoe I, and this means that while he is teaching other topics such as the Cold War, he is not motivated. This implies that he may master and deliver content effectively for only one topic, while the ones that demotivate him may not teach them well, and this may ultimately affect their performance negatively in public examinations. Participant 4 expressed

motivation due to his love for History and the desire to instil necessary patriotism in his learners. Motivation that emanates from love may act as a transforming motivation that may have impacts on the performance of History in public examinations. This is because teachers with such a kind of motivation may teach the History subject with love and may select what is best for them and their learners for effective teaching and learning and this may eventually lead to good performance in public examinations.

Similarly, participant 6 stated that he is motivated by the love for History and values such as peace, patriotism, and resilience that History inculcates. Participant 7 expressed his motivation to teach history, as his learners also enjoy learning history. This implies that his motivation stems from learners' motivation to learn history, not from his enthusiasm for the subject. It follows that if the learners were unable to enjoy studying History, he would also become demotivated. This could have a detrimental impact on History learners' performance because it is not always the case that learners would always love studying History and there may be instances when they have a negative attitude towards learning it. Generally, the study revealed that three teachers are motivated by their love for the subject, while the remaining three are driven by specific History topics, with one being driven by learners' motivation. Therefore, the analysis reveals that all teachers were motivated to teach history.

4.5.2 Relationship with principals, other teachers and learners

Table 7:Relationships with Principals, other Teachers and Learners

Participants	Categories	Participant Quotes
Participant 1	Not friendly with principal, better with other teachers and cordial with learners.	"Our principals tend to be our smaller godsthe relations will not be friendly as such. With the other teacher, the relationship is good and we tend to collaborate more often. Also, with my learners the relationship is very cordial."
Participant 2	Good with principal, other teachers and learners.	"With my principal and other teachers, I make sure that our relationship is good at all times. Also with my learners because it makes work to become easier because good relations make me happy with my work."

Participant 3	Smooth with a principal and	"My relationship with my learners is very smooth except for other
	learners but not smooth with	teachers who teach different subjects from History because they have a
	other teachers.	lot of discriminating attitude that History is a storytelling, old-fashion
		and biased subject. The relationship with the principal is very smooth
		because he is interested in my study."
Participant 4	Good with a principal and	"My relationship is good with teachers and principal for
	teachers. Friendly with learners.	professionalism. Friendly with learners to reinforce conducive learning
		for learners."
Participant 5	Healthy and collaborative with	"Healthy and collaborative because being societal focus and people-
	a principal, teachers and	focused is one of the benefits of teaching history."
	learners.	
Participant 6	Good with a principal, teachers	"Good because I am a friendly person who knows how to deal with
	and learners.	different people of different characters."
	~	
Participant 7	Good with a principal, teachers	"My relationship is good with them and I do not have problem with
	and learners.	them."

Table 7 shows data on the relationships of participants with their principals, other teachers, and their learners, indicating whether their relationships are friendly or not. The question that was asked was, "How are your relationships with other teachers, your learners, and the principal, and why do you have such relationships?" The findings revealed that five participants indicated that their relationships with their principals, other teachers, and their learners are smooth, collaborative, and cordial. They indicated that such relationships motivate them, help them make their work easier, and enhance a conducive environment, as emphasised by participant 4. Therefore, these kinds of relationships may motivate teachers to teach their learners well in order to enhance their performance in public examinations.

However, participant 1 reported unfriendly relationships with principals due to their perceived smaller god status, while his relationships with other teachers are collaborative and cordial with learners. Unlike friendly, smooth, and collaborative relationships, the unfriendly relationship of the principals to History teachers may demotivate them as well as their learners because a bad relationship between teachers and principals may also affect learners' emotions negatively, which

may ultimately affect their performance in public examinations negatively. Apart from that, teachers who have unfriendly relationships with the principals may not teach learners effectively due to stress, hence affecting learners negatively in their learning process and their performance in public examinations.

Besides, participant 3 said that his relationships with his principal and his learners are good. However, he said his relationships are not good with other teachers from other subjects due to their discriminatory attitudes towards history. Teaching requires collaboration among all the teachers, and failure to collaborate may result in conflicts that may affect teachers emotionally. This may also affect learners because emotionally-affected History teachers may not impart historical content well to History learners; hence, this may affect the performance of learners negatively. The instability itself may affect learners emotionally, and eventually, this may also affect the performance of History learners negatively.

4.5.3 Lesson preparationTable 8:Lesson Planning and Scheming

Participants	Categories	Quote of Participant
Participant	Do not plan frequently	"Lesson planning helps me to focus, follow the steps and achieve what
1	but when needed.	I want to achieve. My learners are also able to follow and connect
		with everything. It helps me with a blueprint of where I will start and
		finish."
Participant	Do not plan frequently	"Lesson planning helps me to teach the planned topics with the
2	because of overloading	allocated times."
	work but scheme	
	quarterly	
Participant	Not interested in	"Lesson planning does not benefit me. What benefit me is scheming
3	planning but scheming	because it shows me what aspects I have to teach and how and record
		of work done that helps me to see what I have achieved."
Participant	Frequently practice	"They benefit me in establishing plan of execution of work, hence
4	both planning and	logical teaching which likely benefit learners."
	scheming	
Participant	Frequently practice	"They help me to break content, easy teaching and planning, work
5	both planning and	done and easy supervision by school management and inspectorate."
	scheming	

Participant	Frequently practice	"They help me to be to be on the track e.g. what to teach and how to
6	both planning and	teach it."
	scheming	
Participant	Frequently practice	"You cannot go to class without planning. Record of scheme helps me
7	both planning and	to see what I have taught."
	scheming.	

Table 8 above presents data on lesson preparation, specifically lesson planning and scheming. The question that was asked was, "As a History teacher, do you frequently practise lesson planning and scheming? Can you explain how they benefit you and your learners?" The findings revealed that four teachers frequently practice lesson planning and scheming, while three teachers specified that they do not often practice lesson planning. Therefore, the findings of the study showed that the majority of the participants do not practice lesson planning and scheming frequently.

When the participants were asked whether they often practice lesson planning and scheming, they provided different answers. Participant 1 said, "I do not practise lesson planning frequently, but I do it when it is needed." This implies that teaching without lesson planning may be haphazard, and this kind of teaching may be ineffective and may not promote teaching and learning; rather, it may act as a barrier to effective teaching and learning as well as negatively affect the performance of History learners in their public examinations. On top of that, participant 2 claimed, "But as for lesson planning, I do not follow lesson planning frequently due to overloading work." It seems that some of the teachers fail to do lesson planning properly due to the overloading work of one teacher teaching learners from grade 8 to grade 11, hence the need for the government to hire qualified History teachers. Besides, participant 3 believes that scheming is more beneficial than lesson planning as it helps in determining the aspects to be taught, how to teach them, and the work to be done. He said, "Lesson planning does not benefit me. What benefits me is scheming because it shows me what aspects I have to teach and how, and a record of work done that helps me see what I have achieved."

However, all the participants, even the ones who said they do not practice lesson planning and scheming frequently, agreed that these strategies are crucial because they keep teachers on task, help them stay focused, plan ahead, follow instructions, break down content, facilitate teaching and learning, and provide guidance on what to teach and how to teach it. If that is the case, lesson planning and scheming may help teachers set their targets in the process of teaching and learning

and achieve them if done properly, and this may enable them to simply deliver historical content effectively to their learners.

Further, participants indicated that lesson plans and scheming aid in determining teacher work and simplifying supervision by school management and inspectorates. Therefore, lesson planning and scheming may not only act as guidelines for History teachers and their learners but may also act as guidelines for school management for easy supervision to see whether teachers are doing their work effectively or not, and if they are doing their work, how effective they are in doing their work. As a result, the data showed that although some of the participants did not practise lesson planning and scheming frequently, there are crucial tools for teachers to effectively teach and achieve their desired outcomes.

4.5.4 Use of teaching aids *Table 9:Use of Teaching Aids*

Participants	Times of Using Teaching Aids	Participant Quotes
Participant 1	Not often	"It is not often that we bring the teaching aids to the classrooms but we are forced by the topic we are teaching to bring something to the classroom that will help learners to understand.
Participant 2	Rarely	"I use simplified materials from History Association because currently History does not have textbooks. I also use my phone as my resource."
Participant 3	Rarely	"I rarely use teaching aids because there is lack of resources such as mere textbooks and internet. We do not have support of the issues of teaching resources and for me such resources need a lot of money."
Participant 4	Rarely	"I rarely use teaching aids because they are rarely limited."
Participant 5	Weekly	"Weekly because they enhance learning by learners' participation, concentration and class management."
Participant 6	Rarely	"I rarely use them because they are very limited."

Participant 7	Use them often	"I often use teaching aids because they help me to transmit information I want to
		transmit. For example, use of maps in demonstrating how European countries are
		located during my teaching of European Map. This simplifies the work when they
		see the countries on the map than when you teach them by mouth that Belgium is
		near Germany."

Table 9 above presents data on the use of teaching aids. The participants were asked whether they often use teaching aids or not and why. Five participants indicated that they rarely use teaching aids because they are very limited, some are very costly, and they are not able to afford them. However, one of the participants said she used simplified materials from the Lesotho History Teachers Association and her phone to solve this problem. Teaching aids attract learners' attention in the process of teaching and learning. For that reason, if History teachers are not using them due to their shortages, this may affect the learners negatively as they may not actively be engaged in the process of teaching and learning. As a consequence, this may not only affect their effective teaching but also their performance in public examinations since they will respond to questions on concepts which they failed to effectively grasp. Therefore, the data showed that the majority of teachers do not use teaching aids in their teaching due to their scarcity. Participant 4 said, "I rarely use teaching aids because they are rarely limited." To add on to that, participant 3 said, "I rarely use teaching aids because there is a lack of resources, such as mere textbooks and the internet."

On the other hand, participant 1 said that he uses teaching aids depending on the topic he is going to teach. Furthermore, participant 5 indicated that he uses teaching aids weekly because they enhance learners' participation, concentration, and class management. This means that through the use of teaching aids, History learners may concentrate on the topic being taught without making noise; hence, it may be easy for teachers to manage their learners in their classrooms. Participant 7 indicated that he frequently uses teaching aids to effectively convey desired information. Teaching aids are also important in helping teachers effectively transmit the planned historical knowledge they want to impart to learners, and this may enhance the performance of History learners because they may gain the required knowledge from their effective teachers.

4.5.5 Use of Teaching Strategies *Table 10:Use of Teaching Strategies*

Participants	Kinds of teaching strategies used	Participant quotes

Teacher 1	Explanation, discovery and question and answers.	"I use variety of teaching strategies depending on the topic that I am teaching. I use both teacher centred and learner centred depending on what I am going to teach and the nature of the topic."
Teacher 2	Lecturing, role-play, presentations, field trips and group discussion.	"Teaching methods for rural schools like mine is lecturing because there are no textbooks, no internet for them to search information for themselves. So, a teacher does all the work. Apart from that I use role-play, group discussion, presentations and field trip like History fair."
Teacher 3	Lecturing, discovery, discussion and probing.	In history, there is a lot of lecturing methods because learners are not aware of some information but I intertwine learner-centred ones whereby I give learners work to go and discover information from their elders particularly on both European and Lesotho History as some of their parents fought in the First-World war."
Teacher 4	Learner-centred and teacher-centred.	"I use learner-centred to enhance independence. I also use teacher centered as to enhance required speed needed to complete syllabus by end of the year."
Teacher 5	Question and answer, discussion, debate, presentation, information finding and group work.	"I use question and answer, discussion, debate, presentation, information finding and group discussion because they are learner-centtred."
Teacher 6	Discussion, explanation, question and answer, role-play and discovery.	"I use discussion, explanation, question and answer, role-play and discovery."
Teacher 7	Question and answer and lecturing.	"I use question and answer method and lecturing."

Table 10 above presents data on the teaching strategies used by participants in the teaching of history. The participants were asked the following question, "What kinds of teaching strategies are you using in the process of teaching? Why are you using such teaching strategies?" The findings revealed that most of the History teachers use learner-centred methods of teaching, such as question and answer, discovery, and discussion, while a few use role-play, field trips, presentations, debate, and probing. It is important to use these kinds of teaching strategies as they engage learners in learning and motivate them to learn. On account of this, effective use of these strategies may enhance the performance of History learners in their public examinations.

On the other hand, it seems that teacher-centred methods of teaching such as explanation and lecturing are also mostly used. Participant 4 indicated that they use teacher-centred methods to

enhance the required speed to complete the syllabus on time. Participant 2 added that she uses the lecturing method because there are no textbooks and no internet for learners to search for information for themselves. Therefore, it might be that History teachers are more teacher-centred than learner-centred in order to complete the syllabus on time because History has a lot of content. They may also use teacher-centred methods due to a shortage of teaching and learning resources. Hence, this may affect the performance of History learners negatively because teacher-centred instruction does not engage them in learning. Unlike learner-centred, which enables them to learn by doing, teacher-centred makes them passive receivers of knowledge and memorise information instead of understanding it profoundly.

Participant 4 indicated that he uses learner-centred methods to enhance the independent learning of learners, while participant 3 indicated that she uses them for learners to discover knowledge for themselves. Using teaching strategies with the purpose of instilling knowledge in learners through discovery may enhance the performance of History learners in public examinations. However, this requires closer supervision of History teachers and their facilitation.

Nonetheless, participant 6 indicated that she uses questions and answers, discussion, information-finding, debate, and presentation because they are learner-centred. This implies that some teachers use learner-centred methods of teaching just because they were encouraged to use them. They lack a clear understanding of how to use such methods to enhance the participation of learners and impart knowledge to them with the goal of achieving good performance in public examinations. This may affect the performance of History learners in their public examinations because such teachers may just teach because it is their duty without a clear vision of the outcomes they want to achieve with their learners.

4.5.6 Assessment methods

Table 11:Assignments and Weekly Tests

Participants Times learners are Participant Quote

given assignments and weekly tests

Teacher 1 Often but with weekly test not often.

"I often give my learners assignments but in terms of weekly tests, I do not. Weekly tests demand a lot of work for me because I am teaching grade 8 to 11."

Teacher 2	Once a week for assignments and monthly for tests.	"I give learners assignments once a week because History is overloaded in the syllabus and if I focus on giving learners assignments and weekly tests is challenging because marking History is not easy. I may be left behind by focusing on the assignments and tests. I give them tests monthly."
Teacher 3	No longer give learners written assignments and weekly tests but give them oral ones.	"I used to give them the assignments and weekly tests but I no longer do that because most of them fail to do their assignments because they live far from school and arrive at their homes at night-time. It is better to give them oral assignments where they should just discover the stories and narrate them in the class rather than writing because learners have a lot of challenges in writing. They like oral assignments and hate writing."
Teacher 4	Repeatedly for both assignments and weekly tests.	"Yes, I do repeatedly give my learners assignments and weekly tests to familiarise them with mannerism of questioning and encourage and enhance their studying frequency."
Teacher 5	Fortnightly for both assignments and weekly tests.	"Fortnightly to assess learners' knowledge on the topic and learners' progress in the learning and teaching of history."
Teacher 6	Once a week both assignments and weekly tests.	"Yes, once a week."
Teacher 7	One test per week but assignment is for daily.	"I give learners one test per week but assignment is for daily."

Table 11 presents the times participants give their learners assignments and weekly tests. Participants were asked whether they were always giving their learners assignments and weekly tests or not. The findings revealed that 3 participants give learners their assignments repeatedly, while the other 3 participants indicated that they give learners assignments once a week, except one teacher, who indicated that he does not give learners written assignments. Participant 5 stated that he gives learners assignments fortnightly, which is once every two weeks. Assessment may play a very significant role in enhancing the performance of History learners in their public examinations because it helps learners become familiar with the structure of the questions that are asked in public examinations. Regarding the participant who does not give learners a written assessment, this is a very great challenge that may hinder learners' progress in their public

examinations because public examinations are written and not oral. Therefore, the analysis indicates that the majority of teachers do not assign assignments repeatedly to their learners, with one teacher not giving them due to challenges such as overloading work.

Concerning the weekly tests, one participant indicated that he gives learners weekly tests repeatedly, while three participants indicated that they give their learners weekly tests once a week, including the one who gives them test once every two weeks. Providing History learners with weekly tests may benefit them in their public examinations because it enables them to identify their mistakes in learning and correct them before they write the examinations. Contrary to this, not providing History learners with weekly tests may result in poor performance in public examinations because learners may not even be familiar with the structure of examination questions. Participant 1, as quoted in the table above, specified that he does not give learners weekly tests often because it is demanding, while participant 3, as quoted in the table above, indicated that he no longer gives learners written weekly tests but provides them with surprise oral and presentation quizzes because learners hate writing.

Participant 6 indicated that she gives learners tests monthly, not weekly. Participant 4 mentioned that he assigns assignments and administers tests every week to help learners become familiar with the style of questioning and to promote and increase their frequency of learning. Participant 5 expressed that he administers weekly tests to evaluate learners' progress in teaching History and their comprehension of the subject matter. Therefore, this may help teachers to see whether learners are following or not in the process of teaching and learning, and if they are not following, they may be re-taught a certain topic that they do not understand. In doing so, repeatedly and passionately, this may improve the performance of History learners in their public examinations.

4.5.7 Immediate feedbackTable 12:Immediate Feedback on Assignments and Weekly Tests

Participants	Times of feedback	Participant quotes
Teacher 1	Same day they were given	"In terms of assignment, I give learners feedback the very same day I
	assignment or the following day	gave them assignment or the following day. In terms of test, I give them
	and after a week for test.	feedback after a week."

Teacher 2	In the following lesson or after week for both assignments and weekly tests.	"During low workload, I give feedback in the following lesson for assignments to ensure no content is left behind, as History is difficult to understand due to its chronology of events."
Teacher 3	Within three days or two days for both assignments and weekly tests.	"I give them feedback within three days or two days so that we can engage in errors they have committed."
Teacher 4	After three to ten days for both assignments and tests.	"I give my learners feedback after 3-10 days depending on my schedule. The quicker the better so that I will be able to do corrections with them."
Teacher 5	The following day for assignment and after two days for the tests.	"The following day for assignment and after two days for test in order for learners to know their progress in the learning of history."
Teacher 6	After marking.	"As soon as I am done with marking."
Teacher 7	Friday test, its feedback is given every Monday.	"I often give learners tests Friday and give them feedback on Monday."

Table 12 presents the status of participants' provision of immediate feedback on assignments and weekly tests to their learners. It shows whether participants provide their learners with immediate feedback or not. The findings showed that the majority of teachers provide their learners with immediate feedback on both assignments and weekly tests.

On the question of whether participants give learners immediate feedback or not, participant 1 indicated that he gives learners feedback the very same day he gave them an assignment or the following day, while he gives learners test feedback after a week. This shows that the participant gives learners immediate feedback so that they can work on the learners' mistakes and correct them on time in order to improve their performance. In addition, participant 5 stated that he gives learners feedback the following day for assignments and after two days for the test in order for learners to know their progress in the learning of history. Knowing the progress of learning for History learners may be beneficial in preparing them for their final examinations, as the identified mistakes may be corrected and learners may not repeat them while they are in their public examinations.

Concerning the weekly test, participant 7 indicated that he often gives learners tests on Friday and gives them feedback on Monday. Giving learners tests on Friday may benefit teachers as they may mark well during the weekend without being under pressure, and this may reduce the risks that

may occur during the week when there is too much work to do. Apart from that, the participant indicated that he gives learners feedback after 3–10 days, depending on his schedule. On the other hand, participant 6 indicated that she gives learners feedback as soon as she has finished marking. Ultimately, all participants agreed that the purpose of giving learners immediate feedback is to help them make corrections to the mistakes they make. However, all the details of the reasons for the provision of immediate feedback are given in Table 14 below.

4.5.8 Correction from feedbackTable 13:Ways in Which Corrections from Feedback can Improve the Performance of History learners

Participants	Categories	Quote from participants
Teacher 1	Help them to see where they were	"They help learners see where they were wrong, how they should
	wrong and avoid the very similar	have answered the questions. This helps to avoid very same
	mistakes they did.	mistakes they did in future."
Teacher 2	Help learners to correct their	"They help learners to correct their mistakes. They help learners to
	mistakes and understand the topic	understand the topic being taught."
	being taught.	
Teacher 3	Help learners to study further but it	"This engages them to study further and look what they have missed
	depends on the motives of learners.	but in some cases, it depends on how learners are motivated and
		their desires in history. There are some learners who are still
		demotivated and get the same results even after doing corrections."
Teacher 4	Help learners to identify challenges.	"They help learners to identify challenges and address them during
		revision."
Teacher 5	Prune learners' weaknesses and	"Learners' weaknesses and boost their self-confidence and self-
	build their self-confidence.	awareness in the learning and performance of history."
Teacher 6	Awareness of mistakes.	"They help learners to be aware of their mistakes."
Teacher 7	Help learners to know where they	"They help learners to know where they failed the questions."
	failed.	

Table 13 above presents data on ways in which corrections from feedback can improve the performance of History learners. Participants were asked how corrections from feedback can benefit learners. According to the analysis of this study, the majority of participants said feedback corrections assist learners in identifying their challenges, recognising their errors, and avoiding them in the upcoming tests, assignments, and final examinations. This may improve the

performance of History learners because recognising learners' errors through formative assessments and correcting them is part of their preparation for final examinations.

The findings further showed that feedback corrections strengthen learners' weaknesses and build their self-confidence and self-awareness. This implies that corrections from constructive feedback may act as a motivational tool for History learners, as they may make them aware of their mistakes and correct them. Participant 5 indicated that learners are able to identify challenges where they were wrong, prune learners' weaknesses, and boost their self-confidence and self-awareness in the learning and performance of History.

On the other hand, participant 3 argued that feedback corrections help learners to study further, depending on their motives. He claimed, "This engages them to study further and look for what they have missed, but in some cases, it depends on how motivated learners are and their desires in history. There are some learners who are still demotivated and get the same results even after making corrections." This may seem true, as some learners may be demotivated by personal factors they encounter, and sometimes even if they are provided with corrections, they are still getting the answers wrong, hence the importance of counselling sessions in high schools for the personal issues of learners to reduce any stress for better performance in public examinations. Participant 2 stated that corrections help learners understand the topic being taught. By providing learners with constructive corrections, the more they learn the taught topic.

4.5.9 Challenges encountered by History learners at school and in learning

Table 14:Challenges Encountered by History learners at School and in Learning

Participants	Challenges	Participant quotes
Teacher 1	Learners are mocked for studying history, demotivation of learners, lack of textbooks, lack of support from the Ministry of Education and Training, lack of ICT resources, lack of regular training in teaching of History and traditional ways of teaching.	"Learners are being mocked in doing History. It demotivates them from studying historythere are no teaching resources such as textbooks and ICT resourcesthe response from the Ministry of Education and Training is very weak in terms of providing our schools with the resourcesthe teachers are also not given regular training in teaching of the subject. History relies on traditional ways of teaching."
Teacher 2	Lack of support from the school management, lack of textbooks, no technology.	"There are no textbooks and no technology. Learners do not have enough materials and resources and learners become tired of listening to the teachers onlyIt is impossible to make them copies due to lack of money because school management do not help."
Teacher 3	Demotivation of teachers, failure of learners to conceptualise content, interpret and express their knowledge.	"Learners fail to put the taught content according to their understandingI become demotivated by the poor results of my learners"

Teacher 4	Lack of resources, overloaded syllabus and struggle of learners in answering the questions.	"Avoiding bias by learners, lack of resources and overloaded syllabus."
Teacher 5	Problem in essay writing, English Language proficiency, lack of relevant textbooks and false belief that History is a failing subject.	"Essay writing due to poor linguistic ability in English Language acquisition, lack of relevant textbooksfalse belief that History is a failing subject."
Teacher 6	Learners' difficult understanding of English, difficult in reading skills and lack of text books.	"Difficulty of understanding in English Language, reading skills and lack resources such as textbooks."
Teacher 7	Learners forget the years.	"Because History is about chronology and sequence of events, learners sometimes forget the years."

Table 14 above shows the challenges encountered by History learners at school and in their learning. As found by this study, the majority of teachers highlighted the most challenges as a lack of teaching resources such as textbooks and technology, criticism towards History learners' traditional ways of teaching, and learners' lack of English proficiency. It is in this regard that participant 1 highlighted the lack of teaching resources such as textbooks and ICT. On the similar issue, participant 2 claimed that there is no technology or teaching resources. Teaching resources such as technology and textbooks are needed for instructional guides and effective transmission of content, and their unavailability may result in ineffective teaching and ultimately affect the performance of History learners negatively.

In regard to criticism towards History and traditional ways of teaching, participant 1 stated that learners are being mocked for doing history. It seems that this mocking developed a bad attitude in learners towards History. Traditional ways of teaching seem to be employed by History teachers to finish the syllabus on time and to address the challenge of lack of resources and this may ultimately affect the performance of History learners in their public examinations. To add on this point, participant 5 claimed that the false belief that History is a failing subject is a challenge. participant 6 further stated reading skills as a major challenge. Participant 5 further stated that essay writing due to poor linguistic ability in English and acquisition is a challenge. The challenge of English proficiency may affect the performance of History learners in public examinations because they fail to understand what is being taught in the classrooms.

The study further discovered a lack of support from the Ministry of Education and Training, learners' failure to conceptualise content, and a lack of History teachers' regular training. It seems

that lack of support from the Ministry of Education and Training may emanate from failure to provide schools with teaching resources and failure to provide teachers with regular training and this may affect the performance of History learners negatively.

Some identified challenges include a lack of school management support, difficulty in learners' reading skills, demotivation of teachers and learners, an overloaded syllabus, and the struggle of learners in answering the questions. Failure for school management to help History teachers with teaching and learning resources, failure for school management to motivate History teachers and learners, and autocratic management may affect the school environment negatively and ultimately affect the performance of History learners negatively. An overloaded syllabus also appears to limit History teachers' ability to deliver historical content effectively, affecting the performance of History learners negatively. However, in Table 16 below are how these challenges could be mitigated in order to improve the performance of History learners in their public examinations.

4.5.10 Mitigation on performance of History learners in public examinations *Table 15:Solutions to the Challenges*

Participants	Categories	Participant Quote
Participant1	The Ministry of Education and Training must provide schools with History teaching resources, Teachers should be provided with regular training on how to teach history, and learners must be allowed to come at schools with their cellphones.	"The Ministry of Education and Training must provide the schools with History teaching resourcesteachers need regular training because we cannot expect the teachers who were trained 20 years ago to be as effective as now as there are changes in the teaching and learningthe schools must eradicate the policies whereby learners are prohibited from coming with their cellphones at schools."
Participant2	Learners should take school trips, incorporation of technology in the teaching of history, regular training of teachers and the government should employ History teachers.	"To confirm the reality of historical events requires technology, they need the books and pictures. They need to take the trips to visit historical places. Government should really employ History teachers."
Participant3	Incorporating technology into the classrooms (Wi-Fi and allowing learners to use their cellphones in schools).	"if we bring technology in the classroom, let us allow learners to come with their cellphones at schools, provide Wi-Fi for them so that they can use those cellphones for their productivity"
Participant4	ECOL must provide clear marking system.	"vivid ECOL marking scheme."

Participant5	Provision of relevant History textbooks by Ministry of Education and Training, making History core subject, including teachers in ECOL marking panel, using online learning, using technology in the classrooms and employing qualified History examiners.	"The Ministry of Education and Training must provide relevant textbooks to stop learners' reliance on teachers as their mobile textbooks. Making History core subjecthaving online learning through the utilisation of technology to enhance research skills. Teachers should be included in the ECOL setting panel as the current examiners are not class practitioners"
Participant6	Provision of textbooks by the government, Daily use of English Language by learners and they must read to understand, questions should be put to the standard of learners.	"Learners must make English their daily language, read to understand and the government must provide the schools with the textbooks."
Participant7	The principals must support history. Learners must be given more assignments, practice of discovery learning and the school's quick response to the shortage of History textbooks.	"Learners must be given more assignments and tests so that they will be used to historical contentalso learn through discoveryin other subjects when the teachers report about the shortage of textbooks, the principal respond quickly and such teachers get the books. This is not the case with History where it is not easy to get books."

Table 15 above presents solutions to the challenges encountered by History learners at school and during teaching and learning. These solutions provide ways in which performance of History learners could be improved in their public examinations. Based on the findings of this study, the majority of participants specified that the Ministry of Education and Training should provide the schools with textbooks. This could help in enhancing the performance of History learners as textbooks may help learners to read on their own to develop their prior knowledge and this may ease their understanding.

The findings also showed that Examination Council of Lesotho could provide clear marking scheme. In this marking scheme, History teachers should be included so as to enable them in providing proper marking for their learners. The data also reflected that technology should be incorporated in the teaching (by allowing learners to come to schools with their cellphones, and providing Wi-Fi at schools) and providing History teachers with regular training. Participant 1 claimed, "The schools must also eradicate the policies whereby learners are prohibited from coming with their cellphones at schools."

Moreover, participant 3 added that learners should be allowed to come with their cellphones at their respective schools. Use of these cellphones may enable History learners to do their discovery effectively and this may enhance their performance in their public examinations. Participant 5 suggested that History must be made a core subject not an elective subject as it is now. Making History a core subject may improve its status which may result into more learners. In regard to marking, the Examination Council of Lesotho should reconsider its marking process, by informing History teachers about the evolving marking before they are implemented, including History teachers in marking and implementing a clear marking system. This could help teachers as the implementers of curriculum to mark learners according to the standard of ECOL. ECOL should also train History teachers how to mark according to their standard in order to enhance the performance of History learners.

Participant 5 further suggested that the government should hire many History teachers and qualified examiners who are class practitioners. This was seconded by participant 2, who stated that the government should really hire History teachers. Hiring more qualified History teachers may improve the performance of History learners because the more the History teachers, the more effective their teaching and learning will be. In addition, as shown in the table above under the participant quote column, participant 7 suggested the need of regularly assessing learners in order to expose them to historical content and knowledge and provision of support from school management. Participant 6 further recommended that learners should use the English Language daily and should read to understand. Generally, most of the participants suggested the provision of History textbooks at schools, allowing learners to use their phones at school, regular training of teachers, and clear marking by ECOL, which were identified as the major challenges.

4.6 Conclusion

Based on the findings of this study, the performance in History public examinations is a major concern for educational stakeholders. The findings of this study provided evidence of poor performance by History learners in the 2023 final examinations, where the majority of the schools performed poorly. This study demonstrated several factors contributing to this performance, not excluding a lack of resources such as textbooks and technological resources, improper training of History teachers, poor preparation by History teachers, incompetency of History teachers, and

improper use of assessment strategies by History teachers. This implies that the status of History is unsatisfactory due to the high failure rate of History learners in public examinations.

Bearing that in mind, despite the benefits provided by History to the national curriculum and the entire nation, the subject is in danger of being removed from the curriculum because of its poor performance in the public examinations. Hence, the study established that there is a need for involved educational stakeholders, such as the Ministry of Education and Training, to support the schools through the provision of funding for teaching and learning resources. Not only that, but History teachers also need to be properly trained on how to teach the subject and mark it.

CHAPTER FIVE

SUMMARY OF FINDINGS, DISCUSSIONS, CONCLUSIONS AND RECOMMENDATIONS

5.1 Introduction

This study explores teachers' perceptions on performance in History public examinations in five high schools in Maseru. This chapter is a presentation of the summary of the findings, discussion of the findings, conclusions of the findings, and recommendations based on the findings of the study. The findings are discussed based on the research questions of the study. The chapter further discusses the findings of the study based on the application of ideas from the theoretical perspectives discussed in Chapter One and the relevant literature.

5.2 Summary of the findings

This section gives a brief summary of the research problem, the methodology used in this study, the limitations of the study, and the implications of the findings. The purpose of this study was to explore teachers' perceptions on performance in History public examinations in five high schools in Maseru district. The highlighted research problem in Chapter One reveals that History is underperformed in Lesotho's public examinations, particularly among grade 11 History learners from many schools. This claim was supported by the Examination Council of Lesotho reports and History Examiners' reports (ECOL, 2020; ECOL, 2021; ECOL, 2022; ECOL, 2023). Despite this problem, it appears that the studies on teachers' perceptions on why the majority of schools and grade 11 History learners perform poorly in public examinations seem to be limited, hence the significance of this study.

In addition, the study employed the interpretivism paradigm as the philosophical lens of the study to understand the subjective understanding of the teachers about their perceptions on performance in History public examinations. Besides, the study utilised the qualitative approach to understand, describe, and explain the various interpretations of teachers' perceptions on the status of History in the national curriculum with regard to learners' performance, as well as the factors influencing History learners' performance and mitigating their performance in public examinations.

In addition, the study used the study design of five different schools in order to obtain a detailed and tangible comprehension of teachers' perceptions on performance in History public examinations. The study further employed purposive sampling to select five schools, including both well-performing and unsatisfactory schools, and seven History teachers. Aside from that, field notes, open-ended questionnaires, and semi-structured interviews were used to gather valuable and in-depth data from teachers regarding performance of learners in History public examinations.

More importantly, the researcher applied credibility, transferability, dependability, and confirmability to the data and interpretations to ensure the trustworthiness and quality of the study. This was done through member checking, peer debriefing, tracking the precise methods used for data generation, analysis, and interpretation, and providing sufficient related information from the literature. Also, the study ensured trustworthiness by justifying theoretical, methodological, and analytical choices, including Expectancy Theory, research paradigm, approach, design, instruments, and data analysis, respectively. In other words, the researcher provided the reasons for the selection of these choices. Apart from that, the researcher thoroughly reviewed and revised the data during the data generation process. The researcher also adhered to research protocols, including voluntary participation, informed consent, anonymity, confidentiality, potential harm, and result communication.

The limitation of this study lies in its focus on History teachers' perceptions, excluding their learners due to its focus on public examination performance; hence, there is a need to make future studies explore the areas left out of this study. Therefore, a similar study may focus on learners' perceptions in the future to enhance their understanding. However, in the future, the focus may also be on learners' perceptions on performance in History internal examinations. Another constraint was the lack of information regarding the effectiveness of History teachers' marking, as most of them expressed dissatisfaction with the regular changes in marking and the new approaches to examination-based History marking as major causes of the poor performance of History learners. Consequently, another future study may assess the effectiveness of History teachers marking.

The study revealed that History is consistently underperformed in public exams due to factors like insufficient resources, incompetent teachers, inadequate training, inadequate content coverage, and poor teacher preparation. Apart from that, poor performance of History learners in public examinations is also attributed to other factors, including lack of support from the Ministry of Education and Training, lack of support from school management, miscommunication between

History teachers and the Examination Council of Lesotho, learners' English proficiency, and frequent marking changes.

Equally important, the findings of this study further revealed that regular unsatisfactory performance of History could lead to its annulment from schools and complete removal from the syllabus. As a result, this highlights the significance of the study to all relevant educational stakeholders and helps them understand the advantages of having History included in the national curriculum of Lesotho. Therefore, there is a need to enhance the performance of History learners, which is why the current study aims to understand and inform policies and strategies to improve the performance of learners in public examinations of grade 11 History learners.

5.3 Discussion of findings

This section provides a discussion of the findings. The findings of this study were discussed per the research objective. The findings were also discussed according to the three main themes, which are classified according to the three objectives of the study. Aside from that, the discussion utilises the themes identified in Chapter 4. The main themes are followed by the sub-themes, which are the subtopics that appear under each sub-objective. The themes discussed are as follows:

5.3.1 Status of History in national curriculum

The performance of History in public examinations does not only affect History learners; it also affects its status in the national curriculum. The major finding was that History is beneficial in the national curriculum of Lesotho as it imparts moral values such as patriotism values, citizenship values, peace, unity, identity forgiveness, empathy, and collaboration. This performance contributed to its unsatisfactory status in public examinations due to the fact that History is perceived as a failing subject. These findings are in agreement with the findings of Wasserman (2010), who discovered that History introduces learners to their heritage by teaching them about their society's origins, development, functioning, and future state.

Apart from that, History may also instill learners with 21st-century skills such as critical thinking, arguing skills, and reasoning skills. This aligns with the findings of Eeden (2012), who discovered a similar situation. Even though History seems beneficial in the national curriculum, it is performed poorly in public examinations, contributing further to its unsatisfactory status. For that reason, the continuous poor performance of the subject is putting the subject at the risk of being

removed from the national curriculum. This supports the findings of Ntabeni (2006) and Fru (2015a), who found that History is performed poorly due to improper training of History learners.

5.3.1.1 The reasons for the poor performance of in public examinations

The major findings of the study were poor teaching strategies employed by History teachers. Apart from hearing from the participants, I observed demonstrations of incompetent teaching strategies during my interaction with teachers through interviews when most of them stated that they use teacher-centred strategies in order to complete the syllabus and when there is a shortage of textbooks. This means that teacher-centredness is often used because History has a lot of content to be covered and a shortage of resources in schools. Therefore, a learner-centred way of teaching, if used more often, may result in a higher failure rate for History learners in public examinations. This is because it does promote effective learning; hence, it may be difficult for History learners to grasp and understand the historical content due to high memorisation in the passive teacher-centered method of teaching, whereby a teacher dominates the classroom while learners are recipients of knowledge.

Furthermore, lack of textbooks, lack of technology, shortage of History teachers, demotivation of History teachers, and their learners are among the major findings of the study. It is apparent that a lack of textbooks, technology, and the demotivation of both teachers and learners are barriers to effective teaching and learning; hence, they may lead to a higher failure rate among History learners. The study further showed incompetent History teachers with a lack of content knowledge. Therefore, it would appear that it would be difficult for such teachers to deliver and impart historical content effectively to learners.

Frequent shift in the marking of History by ECOL, miscommunication between ECOL and History teachers, learners lack interest in history, and problems in essay writing are among the major findings of the study. Therefore, it seems that this frequent shift and miscommunication between ECOL and History teachers in marking of History may not only affect History teachers negatively but it may also affect the performance of History negatively in their public examinations. Not only that, learners lack of interest in History affect the status of History in national curriculum hence this may affect the enrolment of History learners negatively. Apart from that, the findings revealed incompetent teachers, learners struggling to answer the questions, the difficult introduction to

answering the History questions by ECOL, and a lack of History teacher training on marking by ECOL.

${\bf 5.3.2\;Factors\;influencing\;performance\;of\;History\;learners\;in\;public\;examinations}$

5.3.2.1 Motivation of Teachers in the Teaching of History

The major findings of the study revealed that all interviewed teachers were motivated to teach history. Therefore, their motivation may affect the performance of History learners positively in their public examinations. This is because if teachers are motivated, their learners may also be motivated. These findings support the findings of Thoka (2020), Wilkesmann et al. (2021), and Liu et al. (2022), who discovered similar situations about the motivation of teachers and learners.

It seems that motivated teachers are innovative, and they could work hard to find useful ways to foster effective teaching and learning in order to improve the performance of their learners in public examinations. In other words, they may even use locally available resources in the absence of textbooks and technological tools in the teaching of history. Consequently, the findings of this study are also in concurrence with the Expectancy Theory (Vroom, 1964) underpinning this study, which highlights that an individual will always select the behaviour that may maximise their satisfaction and minimise their stress. It is for this reason that the motivation of teachers and their learners may minimise their stress in the process of teaching and learning, resulting in good performance by learners in their public examinations.

However, those teachers indicated that they are motivated when teaching certain topics. This means that they are demotivated while teaching some of the topics. Although the teachers claimed that they are motivated in the teaching of history, probing them and interacting with them helped to observe that they are not motivated at all due to challenges such as a lack of resources. Therefore, this implies that the teaching environment, job satisfaction, and career development may be the roots of the demotivation of History teachers. For that reason, this study is in line with the findings of Mruma (2013) and Mudongo (2017). Apart from their partial expression, it also showed me that they are not motivated. Therefore, their demotivation may affect the performance of History learners negatively in their public examinations because teaching with a discouragement motive may result in many errors, such as being angry at learners, incompetent teaching, and ineffective delivery of knowledge to learners by History teachers. This is because teachers have a significant influence on motivating learners.

5.3.2.2 Relationship with principals, other teachers, and learners

The findings of the study were that the majority of History teachers have smooth, good, and friendly relationships with principals, other teachers, and learners, while a few teachers indicated that their relationship is only cordial and friendly with the learners but not friendly with the principals and other teachers. Therefore, findings revealed that a good relationship between History teachers and the principals, other teachers, and learners may enhance the performance of learners since the relationship motivates teachers to do their work and reinforces conducive learning for History learners. The findings of this study are in line with the findings of Edgerson (2006), Emosivwe (2023), and Edgbo (2024), which highlighted the importance of good relationships between teachers and principals, other teachers, and learners in enhancing the performance of learners due to a conducive working environment, while unfriendly relationships may affect the performance of History learners negatively.

5.3.2.3 Lesson preparation

The major findings of the study are that the majority of History teachers frequently practise lesson planning and scheming, while the other few teachers do not frequently practice lesson planning. Those who do not practise lesson planning and scheming claimed that they are overloaded with their school work due to the fact that they are the only History teachers in their schools. One of the teachers indicated that he is not interested in lesson planning because it does not benefit him, but what benefits him is scheming. Therefore, it may seem that teachers who practice lesson planning well may contribute to the positive performance of History learners in their public examinations because they may teach effectively due to their proper organisation of work, unlike the ones who do not make lesson preparation.

However, the findings of the study revealed that seven teachers, including those who do not plan and scheme, frequently listed the benefits of lesson planning and scheming. The findings revealed that lesson planning and scheming help teachers to plan well and easily, focus, logically teach, achieve what was planned, teach the planned topics, stay on track, break content, receive easy supervision by school management and inspectorate, and identify what they have taught. The findings are in concurrence with the findings of Ntabeni (2006), Straessel (2014), Mhonyiwa (2014), Mumba (2019), Aada (2020), Strong (2021), and Khan et al. (2024), who found similar findings on effective and ineffective lesson preparations that may affect the performance of History

learners positively and negatively, respectively. This is because this may result in poor content delivery from teachers to their learners, and some of the important topics may not be covered well.

5.3.3 Use of teaching aids and resources in the teaching of History

The findings of this study revealed that the majority of the interviewed History teachers rarely use teaching aids because they are limited. They also indicated that most of their teaching aids are limited History textbooks. For example, one of the teachers said, "I rarely use teaching aids because there is a lack of resources, such as textbooks and the internet. We do not have support for the issues of the teaching resources, and for me to look for such resources, it takes a lot of money." Another teacher said, "I rarely use teaching aids because they are very limited."

Therefore, lack of teaching aids which act as instructional guides in the process of teaching and learning may affect the performance of History learners in their public examinations. This is because their availability and effective usage may motivate History learners to learn and comprehend the content being taught hence enhancing their performance in public examinations. Even though the teaching aids are limited, the findings of this study discovered that teaching aids are very beneficial in the teaching and learning of History because they enhance learning through learners' participation, concentration, and class management. The findings of this study support the findings from the reviewed literature (Munyaradzi, 2013; Limene, 2018; Ada, 2020; Sengai & Mokhele, 2022; Muhoza et al., 2023).

5.3.4 Use of teaching strategies in the teaching of History

This study shows that History teachers use both teacher-centred and learner-centred methods of teaching, depending on the nature of the topic they are going to teach. Although some teachers mentioned that they use other, learner-centered methods of teaching, they indicated that they use lecturing methods due to a lack of teaching resources and in order for them to complete the syllabus. For example, one of the teachers said, "Teaching methods for rural schools like mine are lecturing because there are no textbooks and no internet for them to search for information for themselves." Apart from that, another teacher said, "In History, there are a lot of lecturing methods because learners are not aware of the information."

Some of the teachers use learner-centred methods without understanding why they are using them. In other words, they use them just because they were told to use them. This shows that some History teachers may teach without clearly understanding what they are really doing, and this may

lead to ineffective delivery and impartation of content from History teachers to their learners. For example, one of the teachers said, "I use question and answer, group, discussion, debate, presentation, and information finding because they are learner-centred." Therefore, it seems that the lecturing method is used to complete the syllabus, and when there is a lack of resources, this means that it is used mostly in History because there is a lack of resources and there is too much content to be covered. If that is the case, the poor performance of History learners may also be attributed to rote learning, as indicated by the findings of Nyamwembe, Ondigi, and Kiio (2013), Chong and Yeo (2016), Samuelsson (2018), and Juliana (2020).

5.3.2.6 Assessment methods

The findings of the study showed that the majority of History teachers assess their learners frequently for assignments while the majority of teachers do not give their learners weekly tests frequently. Frequent assessments may be important in promoting the performance of History learners in public examinations. For that reason, the findings of this study support the findings of Atabek et al. (2014), Shin et al. (2017) and Haq et al. (2020). Teachers who do not regularly give their learners assessments claimed that they are overloaded with work as assessments demand a lot of work for them as they are the only History teachers in their schools. Therefore, this implies that although teachers claimed that they give their learners frequent assessment, it seems that this process is not operated properly due to challenges such as lack of History teachers.

5.3.2.7 Immediate feedback

Findings of the study showed that that most of the History teachers provide learners with the immediate feedback for both assignments and weekly tests. The teacher participants indicated that they provide their learners with the immediate feedback so that they can identify their errors, correct them, and know their progress in the in their learning of history. In doing so, teachers are doing the right thing by providing their learners with the immediate feedback. However, this feedback should not come as plain as it is but it should be accompanied by the constructive comments as recommended by the Lesotho educational frameworks of 2009 and 2021, respectively (MOET, 2009; MOET, 2021). Therefore, it would appear that the findings of the study align with findings of the literature (Moghaddam, 2013; Atabek et al., 2014; Djarfour, 2018; Haq et al., 2020).

5.3.2.8 Correction from feedback

The findings of this study revealed that corrections based on the feedback of learners are very beneficial and contribute positively to the performance of History learners in their public examinations. The findings discovered that correction from feedback helps learners to identify their mistakes, correct them, avoid repeating similar mistakes, understand the topic being taught, study further, prune learners' weaknesses, and build learners' confidence. Therefore, this may enable them to perform well in their public examinations. These findings align with the findings of Djarfour (2018) and Haq et al. (2020), who found similar results from the reviewed literature.

However, one of the participants contradicted the findings of the literature by emphasising that correction from the feedback may improve the performance of learners, depending on their motivation. For instance, he said, "This engages learners to study further and look for what they have missed, but in some cases, it depends on how motivated learners are and their desires in History." Therefore, this highlighted the importance of Expectancy Theory in this study, which may play a very significant role in the motivation of both teachers and learners through the stimulation of their expectations regarding the performance of learners in public examinations (Day, 2020). In this regard, good expectations may motivate teachers to encourage their learners to learn, and they may also motivate learners to learn, hence their successful learning which may result in good performance in their public examinations. This implies that the findings of this study are still aligned with Expectancy Theory underpinning this study.

5.3.2.9 Challenges encountered by History learners at school and in learning

Based on the findings of this study, the major challenges encountered by History learners are demotivation, a lack of textbooks, and a lack of technology in the teaching of history. It seems that History learners are demotivated to study History due to its status, which results from consistent poor performance and the false belief that History is a failing subject. This may result in the deterioration of the status of History in the national curriculum. Apart from that, a lack of textbooks and technology in schools appears to be a challenge for History learners. These challenges appear to be the main causes of History learners' poor performance and obstacles to their ability to learn effectively. The study also established that History learners struggle to communicate in English, thereby affecting their chances of doing well in public examinations that are set in English.

Other major challenges that affect History learners' performance in public examinations are their failure to conceptualise content and their struggle to answer the questions. If left unaddressed,

learners' inability to comprehend the historical material they were given in class could potentially prevent them from performing well on their public examinations. Overburdened curriculum is another challenge that might make it difficult for learners studying History to perform well in their public examinations due to the fact that a single teacher may be responsible for a large class and multiple grades. Teaching grades 8 to 11 by one History teacher leads to ineffective teaching due to an unmanageable large class and many grades.

The findings of the study also established that History learners struggle with writing essays, which could eventually have a negative impact on their performance in public examinations where the majority of the questions require an essay-style response. The findings of the study further proven that a lack of support from school management for the use of textbooks and technological tools is a foremost obstacle that could impede efficient teaching and learning and ultimately lead to poor performance in History in public examinations. Therefore, the findings of this study are in agreement with the findings from the literature, which discovered similar results (Nathen, 2000; Ntabeni, 2007; Boadu et al., 2014; Fru, 215b; Annua-Sekyi, 2015; Monserate, 2018; Thoka, 2020; Rey, 2023). These challenges motivated the researcher to find solutions from the perspective of the participant. This resulted in the mitigation of the poor performance of History learners in public examinations, which will be discussed in Section 5.3.3 below.

5.3.3 Mitigation on performance of History learners in public examinations

The findings of this study established that History teachers should be provided with regular training on how to teach History as a way of improving performance of History learners in public examinations. The study also established the incorporation of technology in the teaching of History in order to enhance performance of History learners in their public examinations. Apart from that, the findings of the study revealed that it is important to allow learners to come with their smart phones at school to use them in doing their assignments as a way to improve the performance of History learners in public examinations. Moreover, the study revealed that it is beneficial for ECOL to provide History teachers with a clear marking system and train them on how to mark learners properly. The findings of the study also established that schools should be provided with History textbooks in order to improve the performance of History learners in their public examinations. Therefore, the findings of this study support the findings of the reviewed literature that discovered similar results (Ntabeni, 2006; Boadu et al., 2014; Johnson et al., 2014; Mhonyiwa, 2014; Fru, 2015a; Musau & Abebe, 2015; Meij & Mex, 2018; Monserate, 2018; Zeibell & Clarke,

2018; Lysaght et al., 2019; Dar, 2020; Johnson et al., 2021; Moea, 2022, Sengai & Mokhele, 2022; Peter et al., 2023).

5.4 Conclusions

The purpose of the study was to explore teachers' perceptions on performance in History public examinations in five high schools in Maseru district. The study employed a qualitative approach and a case study design. The study was based on three objectives, which are to: explore the views of teachers on the status of History in the national curriculum considering learners' performance in public examinations; investigate how History teachers perceive the factors influencing the performance of History learners in public examinations; and explore History teachers' suggestions on mitigation of History learners' performance in public examinations.

Based on the findings of this study, it could be concluded that History is very beneficial in the national curriculum of Lesotho as it instils moral values and citizenship values in learners. Apart from that, it seems that History is also beneficial in promoting patriotism, identity, critical thinking skills, reasoning skills, and educating learners about awareness of world-wide events. Although History seems beneficial in the national curriculum, it has been discovered that History is performed poorly in public examinations. Therefore, it is better to rectify this problem because, if not, the subject may be in danger of being completely removed from the curriculum, no matter its importance.

Looking at the findings of this study, the performance of learners in public examinations is not attributed to one factor; instead, it is attributed to several factors, including Improper training of History teachers, poor teaching strategies employed by History teachers, a lack of textbooks lack of technology, a shortage of History teachers, the demotivation of both History teachers and their learners, curriculum misalignment between curriculum designers and examiners, and miscommunication between examiners and History teachers. Although the participants claimed their motivation in the teaching of history, the researcher observed their demotivation due to a lack of resources and the higher failure rate of History learners. Therefore, it is important for teachers to be trained on how to teach history, including technology use, and how to mark it. It is also important for History teachers to be motivated at school and in their teaching in order to enhance the performance of their learners in public examinations.

5.5 Limitations of the study

The aim of the study was to explore teachers' perceptions on performance in History public examinations in five high schools in Maseru. The limitation of this study is that the data was generated from History teachers only, while learners were excluded. The problem with this is that teachers might have provided biased information regarding the performance of learners. A more balanced strategy would have been to generate data from both teachers and History learners, but the problem was that those learners were currently not at their respective schools, as some are at tertiaries while others did not manage to go to tertiaries.

Another limitation is that the data was generated using open-ended questionnaires, interviews, and field notes. Using these techniques with teachers and classroom observation was inadequate. It is believed that including classroom observation, whereby the researcher would have been observing the teachers while they were teaching, would have provided a researcher with a clearer insight into the perceptions of teachers on performance in History public examinations, especially on the factors influencing the performance of learners in their public examinations.

Another limitation is that in generating data, participants were not free to provide me with the information. Despite the clear explanation in the consent letters that the study was for educational purposes only and that the participants' responses would be kept confidential, some of the teachers still believed that their information might be exposed and believed that they had to respond in an unacceptable way. For that reason, they might have provided biased information. Another limitation is that the researcher delayed generating data due to several shifts in interviews with the participants. However, the researcher managed to generate data despite these challenges, including not being allowed to generate data for some of the schools and having to look for other schools.

5.6 Implication of the study

The results of this study on teachers' views of learners' performance in public examinations in history, which appear to be failing at a higher rate than previous years, suggest that learners in History are not gaining enough knowledge during the learning process. Apart from that, a shortage of resources such as textbooks and technology imply that effective teaching and learning may be hindered. Improper training of History teachers on how to teach the subject implies that History teachers may be incompetent in their teaching. Miscommunication between History teachers and examiners implies that there may be incompatibility in the public examination set, which may

ultimately affect the performance of History learners negatively. The benefits of including History in the national curriculum, as stated above, suggest that, given appropriate consideration and mitigation, learners' lives can be transformed by the knowledge they acquire from the subject.

5.7 Recommendations

Based on the above conclusions and the findings of the study, the following recommendations are made: Firstly, it may be important to provide History teachers with regular training on how to teach, assess, and mark History learners. Secondly, the government should provide the schools with History textbooks with lower subsidies. Thirdly, technology should be integrated into the teaching of history, and History teachers should be trained about the importance of technology and how to use technology in their teachings. Besides, the policy that allows learners to come to school with their cellphones must be introduced. Additionally, it may be important for the government to employ several professional and qualified History teachers in order to reduce the overload on inservice History teachers. There should be curriculum alignment between planned, implemented, and intended curriculum.

Further, it may be important for the Ministry of Education and Training, school management, and parents to support the education of History learners with relevant resources, guidance, and advice. Apart from that, there is a need for school management to motivate teachers through rewards and complements so that they can be effective in their work of teaching. Teachers should also be intrinsically motivated by their work, develop a passion for it, be patient with learners, and provide them with rewards and compliments in order to motivate them in the teaching and learning of history. Finally, it may also be important for History teachers to give their learners more assessments to familiarise them with examination questions and immediate feedback to identify their mistakes and correct them as some of the ways to improve their performance in public examinations.

5.8 Recommendation for future research

The fact that the study was based on the performance in History public examinations, it was difficult to access the learners to get their views. That is why the study ultimately focused on the teachers' perceptions on performance in History public examinations. However, in future, the researcher will conduct a study on the learners' perceptions on performance in History internal examinations so that History learners' views will be accessed. Apart from that, the findings of this

study revealed that the evolving marking by Examination Council of Lesotho plays a very significant role in poor performance of History learners in their public examinations due to the fact that History teachers are not even trained on how to mark History. Having considered that, another limitation was the difficulty in evaluating the effectiveness of History teachers in marking the subject hence there will be another future study to be conducted on this matter.

REFERENCES

- Aada, K. "Insight on planning and assessing the teaching-learning process." International Journal of Social Science and Education Sciences 2, no. 2 (2020): 88-96.
- Adebayo, O. O. and Adigun, S. O. (2018). Impact of instructional aids on students' academic performance in physics in secondary schools in Federal Capital Territory (FCT) Abuja, Nigeria. European Scientific Journal, 14(4), 366-376.
- Adedoyin, O. O. (2013). Public examination and its influence on the Botswana educational system:

 Vies of undergraduate education students at the University of Botswana. Asian Journal of

 Humanities and Social Sciences, 1(2), 124-134. www.ajhss.org
- Agbo, I. S. (2015). Factors influencing the use of Information and Communication Technology (ICT) in teaching and learning computer studies in Onaukwu local government area of Ebonyi State-Nigeria. Journal of Education and Practice, 6(7), 71-86. https://www.iiste.org
- Aini, N., Khoyimah, N. and Santoso, I. (2020). Improving students speaking ability through repetition drill. Professional Journal of English Education, 3(1), 26-35.
- Ajala, E. M. (2003). The influence of peace education on labour management relations in selected industries in Nigeria. [Doctoral dissertation, University of Ibadan].
- Akessa, G. M. and Dhufera, A. G. (2015). Factors that influence students' academic performance:

 A case of Rift Valley University, Jimma, Ethiopia. Journal of Education and Practice,
 6(22), 55-63. https://www.iiste.org/book/
- Alele, F. and Malau-Aduli, B. (2023). An introduction to research methods for undergraduate health profession students: Data collection methods. https://jcu.pressbooks.pub/intro-res-methods-heath/chapter/4-5-data-collection-methods/
- Alkalaki, E. (2021). Pedagogical content knowledge: A comparative study of Greek heritage language teachers in Sweden. [Masters' thesis, Stockholm University].
- Al-Mubireek, S. (2021). The effects of cooperative learning versus traditional teaching on students' achievement: A case study. TESOL International Journal, 16(2), 31-56.

- Al-Zoubi, S. M. (2015). Low academic achievement: Causes and results. Theory and Practice in Language Studies, 5(11), 2262-2268. https://dx.doi.org/10.17507/tpls.o511.09
- Amodu, L. O. (2006). Perception: A determinant for effective communication. An African Journal of Philosophy, 9(1), 148-153.
- Ampofo, E. T. and Osei-Owusu, B. (2015). Students' academic performance as mediated by students' academic ambition and effort in the public senior high schools in Ashanti Mampong Municipality of Ghana. International Journal of Academic Research and Reflection, 3(5), 19-35.
- Annua-Sekyi, E. T. and Nti, S. K. (2015). Factors affecting students' performance in English at Colleges of Education in Ghana. International Journal in Research in Humanities, Arts and Literature, 3(10), 30-43. www.impactjournal.us
- Applied Doctoral Centrer (2024). Trustworthiness of the data.

 https://resources.nu.edu/c.php?g=1013606&p=8394398
- Asghar, J. (2013). Critical paradigm: A preassemble for novice researchers. Life Sciences Journal, 10(4), 3121-3127.
- Asian Development Bank. (2017). Innovative strategies for accelerated human resource development in South Asia: Student assessment and examination. Asian Development Bank.
- Asif, F., Safdar, H. and Ali, S. (2020). Factors affecting the performance of school students: Case study of Bahala walpur. Ilkogretimonline-Elementary Education Online, 19(3), 3650-3660. https://ilkgretim-online.org.doi.10.17051/ilkonline.2020.03.258
- Atabek, Y. E., Balkan, K. F. and Cetinkaya, G. (2014). Evaluating the testing effect in the classroom: An effective way to retrieve learned information. Eurasian Journal of Educational Research, 54, 99-116. Back, U. D. (2010). Parental involvement practices in formalised home-school cooperation. Scandinavian Journal of Educational Research, 54(6), 549-563.

- Badar, Y., Bala, T., Mammam, M. A. and Isa, S. G. (2022). The impact of teaching methods on academic performance of secondary school students in Nigeria. International Journal of Development Research, 10(7), 37382-37385.
- Baloglu, N. (2009). Negative behaviour of teachers with regard to high school students in classroom settings. Journal of Institutional Psychology, 36(1), 69-77. https://www.researchgate.net/publication/274567625
- Basol, G. and Johnson, G. (2009). Effectiveness of frequent testing over achievement: A metaanalysis study. International Journal of Human Sciences, 6(2), 99-120.
- Baxter, P. E. and Jack, M. S. (2010). Qualitative case study methodology: Study design and implementation for novice researchers. The Qualitative Report, 13(4), 544-559. https://www.nova.edu/ssss/QR/QR13-4/baxter.pdf
- Bernard, J. (2009). Learning counts: An overview of approaches to understanding, assessing and improving the quality of education for all. UNESCO.
- Bhandari, P. (2023). Ethical considerations in research: Types and examples. https://www.scribbr.com/methodology/research-ethics/
- Bingimlas, K. A. (2009). Barriers to the successful integration of ICT in teaching and learning environments: A review of the literature. Eurasia Journal of Mathematics, Science and Technology Education, 5(3), 235-245.
- Bizimana, A. and Andala, H. O. (2023). English language proficiency factors and students' academic performance in Historysubject in secondary Rwanda: A case of Kayonza district. Global Scientific Journal, 11(6), 1847-1858.
- Boadu, G., Awnah, M., Ababio, A. M. and Ednaquah, S. (2014). An examination of the use of technology in the teaching of history: A study of selected senior high schools in the Cape Coast Metropolis, Ghana. International Journal of Learning, Teaching and Educational Research, 8(1), 187-214.
- Bolapiwa, D. A., Kolawole, A. O., Ayodoe, O.V., Fakunle, A. F. and Adetule, O. (2020). Teachers' teaching experience and educational qualification as correlates of academic performance

- of students in public secondary schools in Ekiti state, Nigeria. Journal of Education and Practice, 11(2), 108-110. https://doi.org/10.176/JEP/11-2-12
- Braun, V. and Clarke, V. (2006). Using thematic analysis in psychology. Qualitative Research in Psychology, 3(2), 77-101. https://dx.doi.org/10.1191/1478088706qp0630a
- Braun, V. and Clarke, V. (Eds). (2012). APA handbook of research methods in psychology, Vol2: Research designs, quantitative, qualitative, neuropsychological and biological. American Psychological Association.
- Briones, S. K. F., Dagamac, R. J. R., David, J. D. and Landerio, C. A. B. (2021). Factors affecting the students' scholastic performance: A survey study. Indonesian Journal of Educational Research and Technology, 2(1), 97-102. https://ejournal.upi.edu/index.php/IJERT/
- Cansiz, M., Ozbaylanli, B. and Colakoglu, M. H. (2019). Impact of school type on student academic achievement. Education and Science, 44(197), 275-314. https://doi.org/10.1539/EB.209.7378
- Caulfield, J. (2023). How to do thematic analysis: Step-by-step guide and examples. https://www.scribbr.com/methodology/thematic-analysis/
- Charlesworth. (2023). Purpose and benefits of maintaining a research journal.

 https://www.cwauthors.com/article/purpose-and-benefits-of-maintaining-a-research-journa
- Chatterjee, M. (2020). Historyand historiography: A study of the nature of history. International Journal of Creative Research Thoughts, 8(10), 818-826. www.ijcvt.org
- Cho, H. (2018). Handbook of research on pedagogies and cultural considerations for youth English Language learners. University of Kansas.
- Chong, C. and Yeo, K. (2016). Why students fail in history: A minor case in Malaysia and solutions

 from cognitive psychology perspective. Mediterranean Journal of Social Sciences, 7(1),

 517-526.
- Cohen, L., Manion, L. and Morrison, K. (2018). Research methods in education. Routledge.
- Connelly, L. M. (2016). Trustworthiness in qualitative research. *Medsurg Nursing Pitman*, 25(6), 435-436. https://www.proquest.com/docriew/1849700459?sourcetype=scholarlt%20Journal

- Creswell, J. & Clark, L. P. (2018). Designing and conducting mixed methods research (3rd Ed.). SAGE Publications Inc.
- Creswell, J. W. (2013). Steps in conducting a scholarly mixed methods study. DBER Speaker Series.
- Croseley, J. (2021). What exactly is thematic analysis: Plain language explanation and definition (with examples)? https://gradcoach.com/what-is-thematic-analysis/
- Cross, S., Whitelock, D. and Mittembeir, J. (2016). Does the quality and quantity of exam revision impact on student satisfaction and performance in the exam itself: Perspectives from undergraduate distance learners? https://oro.open.ac.uk/46937/
- Dagogo, B. H. (2020). The student behaviour and its relationship on academic achievement: A study of Nigeria high schools. International Journal of Social Science and Humanities Research, 8(4), 93-107.
- Dar, A. (2022). Effect of teachers' behaviour on students' academic performance and personality. Global Scientific Journal, 10(10), 2491-2508.
- Davis, C. and Lachlan, K. (2017). Understanding research methods from straight talk about communication research methods (3rd Ed.). Kendall Hunt Publishing.
- Day, C. T. (2020). Expectancy Value Theory as a tool to explore teacher beliefs and motivations in elementary Mathematics instruction. International Electronic Journal of Elementary Education, 13(2), 169-182.
- Deggs, D. and Hernandez, F. (2018). Enhancing the value of qualitative field notes through purposeful reflection. The Qualitative Report, 23(10), 2552-2560. https://doi.org/10.46743/2160-3715/2018.3569
- Delve, H. L. and Limpaecher, A. (2021). What is peer debriefing in qualitative research?

 https://develop.com/blog/peerdebriefing
- Denzin, N. K. and Lincoln, Y. S. (2011). The sage handbook of qualitative research. Sage Publications.
- DeSimone, L. M. (2009). Improving impact studies of teachers' professional development: Toward better conceptualisations and measures. Educational Researcher, 38(3), 181-199.

- Dewed, S. (2020). Thematic analysis approach: A step by step guide for ELT research practitioners. Journal of Nepal English language Teachers Association, 25(1-2), 62-71.
- Djarfour, F. (2018). The effects of teachers' assessment on the learners' performance case study:

 Cheiksh Touhami and Abdelhamid Ben Badissmiddle school in Tamansrasset. [Masters'

 Dissertation, University Ahmed Draia-Adbar].
- Donkoh, R. (2021). Motivation and Job performance among teachers in La-Kwantanang Madina Education Directorate. Afriibary. https://scholar.afribary.com/works/motivation-and-job-performance-among-teachers-in-la-nkwantanang-madina-education-directorate
- Drugs.Com. (2024). What is poor school performance? https://www.drugs.com/cg/poor-school-performance-in-children.html
- Dube, B. A., Nhame, E. and Magonde, S. (2018). Factors affecting integration in the teaching and learning of physical education in South Africa: A case of Johannesburg East Cluster primary schools in the Gauteng province. International Journal of sport, Exercise and Health Research, 2(1), 88-92.
- Duke University Libraries. (2023). Qualitative research: Observation. https://guides.library.duke.edu/c.phpig=289813&p=1934020
- Durisic, M. and Bunijevac, M. (2017). Parental involvement as an important factor for success in education. Centrer for Educational Policy Studies Journal, 7(3), 137-153.
- Edgerson, D. E., Kritsons, W. A. and Herrington, D. (2006). The critical role of the teacherprincipal relationship in the improvement of student achievement in public schools of the United States. The Lamar University Electronic Journal of Student Research, 3, 1-5.
- Edigbo, M. N. (2024). Teacher-students' relationship and its effects on the academic performance of students in secondary schools in Abakaliki local government area, Ebonyi state, Nigeria. Journal of Educational Research, 10(2), 1-17.
- Educational Policy on Ballotpedia (2023). Academic performance.

 https://ballopedia.org/Academic_Performance
- Ehsanipour, T. and Zaccarelli, F. G. (2017). Exploring coaching for powerful technology use in education. Stanford University.

- Elo, S., Kaariainen, M., Kanste, O., Polkki, T., Utrianen, K. and Kyngas, H. (2014). Qualitative content analysis: A focus on trustworthiness. SAGE Publication.
- Emosivwe, S. P., Philomina, A. N., Patience, I. F. and Martins, O. U. (2023). Student-teacher relationship and its effect on the performance of students in secondary schools in Udu local government area. International Journal of Arts and Humanities, 11(5), 69-84. https://doi.org/ijah/10.54978
- Ene, C. U., Anyiam, L. I. and Onoja, A. E. (2022). Influence of teachers' years of teaching experience on students' academic achievement in Basic Certificate examination in basic science. African Journal of Science, Technology and Mathematics Education, 8(1), 488-494. https://www.ajstme.com.ng
- Examination Council of Lesotho (2020). Lesotho General Certificate of Secondary Education/International General Certificate of Secondary Education O LEVEL school's performance. Examination Council of Lesotho.
- Examination Council of Lesotho (2021). Lesotho General Certificate of Secondary Education/International General Certificate of Secondary Education school's performance. Examination council of Lesotho.
- Examination Council of Lesotho (2022). 2022 examiner's report, Historypaper 1 (0184/01). Examination Council of Lesotho.
- Examination Council of Lesotho (2022). 2022 examiner's report, Historypaper 2(0184/02). Examination Council of Lesotho.
- Examination Council of Lesotho (2022). Lesotho General Certificate of Secondary Education school's performance. Examination council of Lesotho.
- Flay, O. O. (2023). Impact of parental involvement on the academic performance of African American elementary school students: A multiple regions analysis. [Doctoral Dissertation, Liberty University].
- Fisher, T. N. (2023). Teachers' perceptions of student performance on the Advanced Placement English Language and composition exam in Department of Defense Schools. [Doctoral Dissertation, Walden University].

- Fleming, J. (2018). Methodologies, methods and ethical considerations for conducting research in work-integrated learning. International Journal of Work-Integrated Learning, 19(3), 205-213.
- Fru, R. N. (2015a). Pedagogic dilemma for Historyeducation: Voices of student teachers in Lesotho. World Journal of Education Research, 2(2), 182-192. www.scholink.org/ojs/index.php/wjer
- Fru, R. N. (2015b). Historyeducation at the crossroads: Challenges and prospects in a Lesotho context. Yesterday and Today, 13, 67-82.
- Gemma, R. (2018). Introduction to positivism, interpretivism and critical theory. Nurse Research, 25(4), 41-49. https://dx.doi.org/doi.10.77481
- George, T. (2023). Semi-structured interview: Definition, guide and examples. https://www.scribbr.com/methodology/semi-structured-inteview/
- Gharehbagh, Z. A., Massourzadeh, A., Khadem, A. M. and Saedi, M. (2022). Reflections on using open-ended questions. Medical Education Bulletin, 3(2), 483-490. https://medicaleducation.bulletin.ir
- Ghofouri, R. and Ofoghi, S. (2016). Trust worth and rigor in qualitative research. International Journal of Advanced Biotechnology and Research (IJBR), 7(4), 1914-1922. https://www.bipublication.com
- Government of Lesotho (2011). National Assembly Electoral Act, 2011. Government of Lesotho.

 AntDatagain (2023). Advantages and disadvantages of open-ended questions.

 https://antdatagain.com/blog/advantages-and-disadvantages-of-open-ended-questions
- Hahn, S., Kim, T. and Seo, B. (2014). Effects of public and private schools on academic achievement. Seoul Journal of Economics, 27(2), 138-147.
- Hammersely, M. (2013). The myth of research-based policy and practice. Sage Publications.
- Hammersley, M. (2018). What is qualitative research? Bloomsbury.
- Hanafi, H., Gistituati, N., Rusdinal, R. and Aimon, H. (2020). The effect of principal policy on teachers' professional commitments and its impact on learning effectiveness in vocational

- schools. Advances in Social Science Education and Humanities Research, 504, 21-26. https://creativecommons.org/licenses/by-in/4.0/
- Haq, M. N. U., Shakil, A. F. and Din, M. N U. (2020). Impact of homework on the student academic performance at secondary school level. Global Social Sciences Review, 5(1), 586-595. https://dx.doi.org/10.31/gssr.2020(v-1).59
- Haradhan, M. (217). Two criteria for good measurements in research: Validity and reliability.

 Annals of Spiru Haret University, 17, (3), 58-82. https://mpra.ub.uni-muenchen.de/83458/
- Harry, L. (2016). The effects of school type on academic performance-evidence from the secondary entrance assessment Exam in Trinidad. New York University.
- Hawes, P. (2021). School facilities and academic achievement of secondary school agricultural science in Ekiti state, Nigeria. Asian Social Science, 7(7), 64-74.
- Heeralal, P. J. H. and Dhurumraj, M. (2017). Factors that contribute to poor learner performance in physical sciences in Kwazulu-Natal province with special reference to school in the Pinetown district. International Journal of Educational Sciences, 14(3),304-317. https://doi.org/10.3190/24566322.2016/14.03.15
- HistoryToday. (2024). What is History? https://www.Historytoday.com/archive/head-head/what-History
- Ho, E. S. (2009). Educational leadership for parental involvement in an Asian context: Insights from Bourdieu's theory of practice. The School Community Journal, 19(2), 101-122. https://www.researchgate.net/publication/365300784
- Ian, X., Zhang, G. and Park, K. H. (2022). The influence of English on high school students' academic achievement. International Journal of Learning and Educational Research, 21(9), 477-493. https://doi.org/10.26803/ijlter.21.9.26
- Ibrahim, A. M. (2012). Thematic analysis: A critical review of its process and evaluation. West East Journal of Social Sciences, 1(1), 39-47.
- Ihan, S., Siraj, D. and Ilyas, Z. (2024). Effect of lesson planning on academic performance: Evidence from the elementary level classroom. Pakistan Social Sciences Review, 8(1), 169-177. https://doi.org/10.35484/pssr.2024

- Insights, E. (2022). What is research instrument? https://www.editage.com/insights/what-is-a-research-instrument
- Iquabal, N. M., Rahaman, S., Chunawala, S., Ahmed, G., Alzoubi, H. M., Alshuride, M. and Alhamad, A. Q. M. (2022). Perceived factors affecting students' academic performance. Academy of Strategic Management Journal, 21(4), 1-15.
- Ishmaru, A. M. (2019). From family engagement to equitable collaboration. Educational Policy, 32(2), 350-385. https://doi.org/10.1177/08959048/769/841
- Ismai, M. M. A. S., Jogezai, N. and Baloch, F. (2020). Hindering and enabling factors towards ICT integration in schools. A developing country perspective. Ilkogretim Online-Elementary Education Online, 19(3), 1537-1547. https://ilkogretim-online.org.t.doi.org/10.17051/ilkoonline.2020.733175
- Ismail, S. M., Rahul, D. R., Patra, I. and Rezvani, E. (2022). Formative vs. summative assessment: Impacts on academic motivation, attitude towards learning, test anxiety, and self-regualtion skill. Language Testing in Asia, 12(40), 1-23. https://doi.org/0.1186/s40468-022-00191-4
- Issar, K. (2021). Students' attitude towards studying Historyand teaching practices. Education Quarterly Reviews, 4(3), 45-50. https://www.asianinstitudeofresearch.org/
- Jackson, R. M. and Andala, H. O. (2022). School management strategies and students: Academic performance in twelve years' basic education in Rwanda. Journal of Education, 5(1), 84-96. https://doi.org/10.53819/8101810t5053
- James, T. (2017). The effect of automatic promotion on performance of learners in primary schools in Mubende district. [Bachelors' degree dissertation, Kampala International University].
- Jeynes, W. H. (2018). A practical model for school leaders to encourage parental involvement and parental engagement. School Leadership and Management, 38(2), 147-163. https://doi.org/10.1080/13032434.2018.1434767
- Johnson, B. S., Johnson, A. M. and Johnson, W. L. (2014). Strategies for improving school performance. Conference Presentation, 2014 Curriculum Conference June 11-12, 2014, 7, Tyler.

- Johnson, C., Dunn, M., Bodlani, L., Hougaari, C. and Bester, H. (2021). Expanding access to education, energy and health services through digital enabled delivery in Lesotho. Finmark Trust.
- Johnson, D. (2017). The role of teachers in motivating students to learn. Journal of Gratitude Studies in Education, 9(1), 46-49.
- <u>Juliana, M. (2020). Lack of resources in classrooms. English Department Research for Change-Wicked Problems in Our World, 38, 2-11.</u>

 https://research.library.kutztown.edu/wickedproblems/38
- Juliana, M. (2020). Lack of resources in classrooms. English Department: Research for change-wicked problems in our world, 38.

 https://research.library.kutztown.edu/wickedproblems/38
- Kahveci, H. (2023). The positive and negative effects of teacher attitudes and behaviours on student progress. Journal of Pedagogical Research, 7(1), 290-306. https://doi.org/10.33902/JPR.202319128
- Kaso, N., Aswar, N., Firman, F. and Ilham, D. (2019). The relationship between principal leadership and teacher performance with student characteristics based on local culture senior high schools. Scientific Journal of Management, 7(1), 87-98.
- Kaur, S. (2019). Role of a teacher in student learning progress. International Journal of Business and Management Invention (IJBMI), 8(12), 41-45. www.ijbmi.org
- Keennwe, J., Mbae, J. G., Mbae, J. G. and Onchwari, G. (2016). Handbook of research on global issues in next-generation teacher education. https://www.igi-global.com/book/handbook-research-global-issues-next/14143#table-of-contents
- Keizer, R., Van Steensel, R., Jongerling, J., Stam, T., Godor, B.P. and Lucassen, N. (2022). Collaborative learning intervention associated with small increases in home-based school involvement for lower SES families in deprived neighbourhoods. Educational Studies, 1-21. https://doi.org/10.1080/0355698.2022.2058320

- Kennedy, G. (2023). Challenges of ICT integration in teachers' education: A case study of College of Education, University of Liberia. International Journal of Social Science and Education Research Studies, 3(5), 860-870. https://doi.org/10.55677/issers/v03I5y2023
- Kenny, D. T. and Faunce, G. (2004). Effects of academic coaching on elementary and secondary school students. The Journal of Educational Research, 98(2), 115-126. https://doi.org/10.3200/JOER.98.2.115
- Kivunja, C. and Kuyini, A. B. (2017). Understanding and applying research paradigms in educational contexts. International Journal of Higher Education, 6(5), 26-41.
- Kokoro, L. L. (2022). The influence of teacher motivation on learner achievement: A case of primary schools in Serowe, Botswana. [Masters' Dissertation, University of South Africa].
- Kraft, M. A., Blazar, D. and Hogan, D. (2018). The effect of teaching coaching on instruction and achievement: A meta-analysis of the causal evidence. Review of Educational Research, 88(1), 52-80.
- Lal Das, D. K. and Bhaskaran, V. (Eds.). (2008). Research methods for social work. New Delhi.
- Lamas, H. A. (2015). School performance. Propositos y Re-presentactiones, 3(1), 313-386. https://dx.doi.org/10.20511/pyr2015.v3nl.74
- Le Courtois, S. (2024). The dark side of rewards and punishment in the classroom part II.

 https://my.chartered.college/research-hub/the-dark-side-of-rewards-and-punishments-in-the-classroom-part-ii/
- Lebata, M. C. (2014). An investigation of performance in the Biology 5090 at selected high schools in Lesotho. [Master of Education Dissertation, University of South Africa].
- Leed, P. D. and Ormod, J. E. (2016). Practical research: Planning and design (11th Ed.). Pearson Education Limited.
- Lekhetho, M. (2013). Stakeholder perspectives on strategies that can improve student performance in the school leaving examinations in Lesotho. Mediterranean Journal of Social Sciences, 4(3), 89-94. https://www.researchgate.net/publication/272709197

- Lekhetho, M. (2021). Exploring factors influencing the performance of high-achieving secondary schools in Lesotho. Issues in Educational Research, 31(3), 871-90. https://www.iier.org/iir31/lekhetho.pdf
- Lesotho News Agency. (2024). Rapapa announces Lesotho General Certificate of Secondary Education results. https://www.lena.gov.ls/13457-2/
- Limene, I., Tanga, M. and Mushorina, T. (2019). The influence of learner-related factors on the academic performance of grade 12 English (L2) learners in the Oshikoto region of Namibia. Journal for Language Teaching, 53(1), 37-57. https://dx.doi.org/10.4314/jlt.rs3il.2
- Limene, T. (2018). Examining factors influencing the academic performance of grade 12 learners in English Language: A study of four schools in the Oshikoto of Namibia. [Masters dissertation, University of Fort Harare].
- Limo, L. P. (2015). The impact of school inspection on students' academic achievement. [Masters' degree dissertation, University of Tanzania].
- Lioyd, C. (2010). Handbook of researcher on discourse behaviour and digital communications: Language structures and social interaction. University of Kansas.
- Lokman, A., Hassan, F., Ustadi, Y. A., Rahman, F. A. A., Zain, M. and Rahmat, N. H. (2022).

 Investigating motivation for learning via Vroom's Theory. International Journal of Academic Research in Business and Social Sciences, 12(1), 504-530.
- Lopz, P., Torrance, M., Rijlaarsdam, G. and Fildago, R. (2021). Evaluating effects of different forms of revision instruction in upper-primary students. Reading and Writing Quarterly, 34(7), 1741-1767. https://doi/10.1007/51145-021-10156-3
- Lysaght, Z., Scully, D., Murchan, D., O'Leary, M. and Shiel, G. (2019). Aligning assessment, learning and teaching in curriculum reform and implementation. Dublin City University.
- Machila, N., Chipindi, F. M., Chiputa, E. and Kaani, B. (2022). A critical analysis of learner performance content-based and outcome-based in the secondary school Historycurriculum. International Journal of Research and Innovation in Social Science, 6(10), 381-390.

- Maguire, M. and Delahunt, B. (2017). Doing a thematic analysis: A practical, step-by-step guide for learning and teaching scholars. All Ireland Journal of Teaching and Learning in Higher Education, 8(3), 3352-33514.
- Maicibi, N. A. (2013). Pertinent issues in employees' management: Human resource and educational management. Net Media Monitor Publishers.
- Makori, A. and Onderi, H. (2014). Examining the teaching and learning resources related challenges facing small and medium-sized public secondary schools in Kenya: A comparative analysis. African Educational Research Journal, 2(2), 72-84.
- Man, E. K. F., Mnsaray, M. F., Amady, F. and Konneh, K. (2020). Assessing the factors affecting students' performance in secondary schools in Kenema city. International Journal of Scientific Research and Engineering Development, 3(6), 696-710. www.ijsred.com
- Mankgele, K. P. (2023). The consequence of classrooms in the teaching and learning of entrepreneurial subjects in secondary schools. Asian Journal of Management Entrepreneurship and Social Science, 3(4), 1533-1545.

 https://ajmesch.com/index.php/ajmesc
- Martirosyan, N., Huang, E. and Wanjohi, R. (2015). Impact of English proficiency on academic performance of international students. Journal of International Studies, 5(1), 60-71. https://jistudnts.org
- Masayile, Q., Nyoni, T., Nyoni, M., Bonga, W. G. and Mussinyani, B. J. (2017). An evaluation of the use of marking schemes in the teaching and learning of commerce at ordinary level: A case study of five secondary schools in Chiefs Sayi and Tiri areas in Gokwe South district, Midlands, Zimbabwe. Dynamic Research Journals of Economics and Finance, 2(6), 1-19.
- Mayasari, R. (2020). The effect of drilling technique towards students' speaking achievement at grade VIEI of SMPN Kota Jambi. [Bachelors' Degree Dissertation, Batanghari University].
- McCombes, S. (2023). Sampling methods/types, techniques and examples.

 https://www.scribbr.com/methodology/sampling-methods/

- McCombes, S. (2023). What is a case study: Definition, examples and methods? https://www.scribbr.com/methodology/case-study/
- Mcdonald, S. M. (2021). Perception: A concept analysis. [Doctoral Dissertations, The University of Texas at Tyler].
- McKim, C. (2023). Meaningful member-checking: A structured approach to member-checking. American Journal of Qualitative Research, 7(3), 41-52. https://doi.org/10.29333/ajqr.org
- Meij, L. W. and Merx, S. (2018). Improving curriculum alignment and achieving learning goals by making the curriculum visible. International Journal for Academic Development, 23(3), 219-231. https://doi.org/10.1080/1360144x.2018.1462187
- Mhonyiwa, J. M. (2014). Factors influencing examination performance in commercial subjects in Tanzania Ordinary Level Secondary Schools: A case of Ilala Municipality. [Master's degree dissertation, The Open University of Tanzania].
- Michael, I. O. and Wumi, O. A. (2016). Causes and remedies to low academic performance of students in public secondary schools: A study of Ijero local government area of Ekiti state.

 Research on Humanities and Social Sciences, 6(15), 66-71. www.iist.org
- Middleton, F. (2023). The 4 types of validity in research: Definitions and examples.

 https://www.scribbr.com/author/fionamiddleton/
- Mikie, M. A. and Warner, C. H. (2011). Classroom learning environments and the mental health of first grade children. Journal of Health and Social Behaviour, 1, 1-5.
- Mlay, N. (2010). The influence of language of instruction on students' academic performance in secondary schools: A comparative study of urban and rural schools in Ausha-Tanzania. [Masters' Dissertation, University of Osto].
- Moea, K. S. (2022). Problems faced by Lesotho piloting high school teachers in implementing the 2009 curriculum and assessment policy. Merit Research Journal of Education and Review, 10(10), 069-073.
- Moghaddam, M. M. (2013). The effect of weekly quizzes on students' final achievement score.

 International Journal of Modern Education and Computer Science, 376-41.

 https://www.mecs.press.org/

- Monserate, C. A. (2018). Impact of technology on the academic performance of students and teaching effectiveness. International Journal of Interdisciplinary Research and Innovations, 6(1), 47-87. www.researchpublish.com
- Mpeirwe, N., Twebaze, R. and Kyayemagye, F. (2024). Teachers' preparedness and syllabus content coverage of Historyand geography subjects in secondary schools of Bukanga County, Isingiro district. American Research Journal of Humanities Social Science, 7(5), 19-26. www.arjhss.com
- Mruma, J. M. (2013). Effect of motivation factors on teachers' performance in Tanzania Education Institution: A case of public secondary schools in Mamaagana district, Mwanza. Open University of Tanzania.
- Mudongo, N. (2017). The influence of school management teams on pupil academic performance in Tutume North Inspectorate schools Munich, GRIN Verlag. https://www.grin.com/document/43871/
- Muhoza, R. N. (Ymaniriho, D. Nshimiyimana, A. Mkunzabo, E., Ntenzimana, N. J. and Bafera, P. N. (2023). The effect of teaching aids on students' performance in biology subject in O'level secondary schools in Kayonza District. Journal of Research Innovation and Implications in Education, 7(2), 124-133.
- Mukherjee, A. (2014). The open-ended questions playbook: Strategies for engaging communication. https://www.olyvy.co/blog/open-ended-questions/
- Mumba, M. (2019). Contribution of lesson preparations to the academic performance of evening class students: A case of Fibobe primary school, Ndola, Zambia. [Masters' Dissertation, The University of Zambia].
- Munyaradzi, G. E. (2013). Teaching methods and students' academic performance. International Journal of Humanities and Social Science Invention, 2(9), 29-35.
- Musau, L. M. and Abebe, M. J. (2015). Teacher qualification and students' academic performance in science mathematics and technology subjects in Kenya. International Journal of Education Administration and Policy Studies, 7(3), 84-89. https://doi.org/10.5897/IJEAPS2014.0386

- Mwale, C. G. (2018). Teachers' experience in innovative teaching and learning of Historyin secondary school. International Journal of Engineering Science and Management, 1(1), 1-8.
- Mwatwana, M. I., Mungai, C., Gathumbi, A. W. and George, G. E. (2014). The impact of Historyteaching/learning resources on student performance in KCSE Historyexaminations: A case of Tigania and Tembe districts Meru County, Kenya. Journal of Education and Practice, 5(3), 64-72.
- Nakhanu, S. B. (2012). Effective syllabus coverage on secondary school students; performance in mathematics in Kenya. International Journal of Educational Science, 4(1), 31-34. https://doi.org/10.39901/24566322.2012/04.01.05
- Narad, A. and Bilkees, A. (2016). Academic performance of senior secondary school students:

 Influence of parental encouragement and school environment. Rupkatha Journal on
 Interdisciplinary Studies in Humanities, 8(2), 13-19.

 https://dx.doi.org/10.216559/rupkatha.v8n2.02
- Nazir, M. I. J., Rahaman, S., Chunawala, S., Ahmed, G., Alsoubi, H. M., Alshuride, M. and Alhamad, A. Q. M. (2021). Perceived factors affecting students' academic performance. Academy of Strategic Management Journal, 21(4), 1-15.
- Neuman, L. W. (2011). Social research methods: Qualitative and quantitative approaches (7th ed.). Allyn Bacon.
- Nguringa, P. and Cheprasov, A. (2023). Historyas a discipline: Overview, definition and example. https://academy/lesson/History-as-an-academic-discipline.html
- Nguyen, H., Ahn, J., Belgrave, A., Lee, J., Cawelti, L., Kim, H. E., Prado, Y. Santagata, R. and Villavicencio, A. (2021). Establishing trustworthiness through algorithmic approaches to qualitative research. Spring Nature Swizerland AG. https://doi.org/10.1007/978-3-0306788-6-4
- Nnganga, P. M. (2014). Effect of marking scheme on the consistency of scoring mathematics examinations. [Masters' Dissertation, University of Nairobi].

- Ntabeni, M. (2006). The status of Historyteaching, learning and examination results in Lesotho, 2000-2004: Implications for teaching education. Yesterday and Today, 100-117.
- Ntabeni, M. (2007). Historyteaching, learning and Junior Certificate (JC) examination results in Lesotho, 2000-2006: Implications for teacher education. Yesterday and Today, 1, 218-238.
- Ntekane, A. (2018). Parental involvement in education. Research Gate, 1(1), 1-9. https://doi.org/10.3389/fpsyg.2019.01384
- Nyamwembe, E. O., Ondigi, S. and Kiio, M. (2013). Attitudes of students towards Historyand government in some selected secondary schools in Mosocho Division, Kisii County, Kenya.

 Journal of Education and Practice, 4(11), 17-26.
- Nyarko, J. A. (2023). Development of Historyas an academic disciple. https://dx/doi.org/10.2139/ssrn.4543233
- Nzoka, J. T. and Orodho, J. A. (2014). School management and students' academic performance:

 How effective are strategies being employed by school managers in secondary schools in

 Embu North district, Embu County Kenya? International Journal of Humanities and Social

 Science, 4(9), 86-99. www.ijhssnet.com
- Oguta, P. A. (2022). Teacher experience factor impetus on student academic performance. Journal of Research, Innovation and Implications in Education, 6(1)141-148.
- Okeke, C., Thomas, A. and Botha, A. (2023). The impact of learners' disruptive behaviour on teachers' well-being: A qualitative approach. International Journal of Research in Business and Social Science, 12(8), 505-511. https://www.ssbfnet.com/ois/index.php/ijrbs
- Okettch, M., Mutisya, M., Sague, J. L, Musyoka, P. and Ngware, M. W. (2013). The effect of active teaching and subject content coverage on students' achievement: Evidence from primary schools in Kenya. London Review of Education, 10(1), 19-33. https://dx.doi.org/10.1080/14748460.2012
- Okongo, R. B., Ngao, G., Rop, N. K. and Nyongesa, W. J. (2015). Effect of availability of teaching and learning resources on the implementation of inclusive education in pre-school centrers in Nyamira North Sub-County, Nyamira County, Kenya. Journal of Education and Practice, 6(35), 132-141. www.iiste.org

- Oliver, V. (2010). 301 Smart answers to tough business etiquette questions. Skyhorse Publishing.
- Ololube, N. P. and Major, N. (2014). School inspection and educational supervision teachers' productivity and effective education programs in Nigeria. International Journal of Scientific Research in Education, 7(1), 91-104. https://www.iisre.com
- Organisation for Economic Cooperation and Development. (2020). Material resources at school in PISA 2018 results (volume v): Effective policies, successful schools. Organisation for Economic Cooperation and Development Publishing. https://doi.org/10.1787/za420765
- Oscan, M. (2021). Factors affecting students' academic achievement according to the teachers' opinion. Education Reform Journal, 6(1), 1-18. https://dx.doi.org/10.22.596/erj2021.06.01.1.18
- Oyinloye, O. M. and Imeda, S. N. (2019). The impact of assessment for learning on learner performance in life science. Eurasia Journal of Mathematica, Science and Technology Education, 15(11), 1-8. https://doi.org/10.2933/ejmste/108689
- Patrick, M. M. (2020). Effects of language of instruction on students' academic performance in community secondary schools in Bunda district. [Masters' Dissertation, The Open University of Tanzania].
- Pertersen, C. I., Baepler, P., Beitz, A., Ching, P., Gorman, K. S. and Neudauer, D. (2020). The tyranny of content: Content coverage as a barrier to evidence-based teaching approaches and ways to overcome it. Life Science Education, 19(17), 1-10. https://doi.org/10.1187/cbe.19-04-0079
- Peter, L., Julius, A. and Deus, T. (2023). The impact of teachers' teaching experience and academic performance in Mathematics in selected schools of Katabi town council.

 Metropolitan Journal of Business and Economics, 2(6), 318-326.
- Phaeton, M. J. (2017). Exploring the alignment of the intended and implemented curriculum through teachers' interpretation: A case study of A-Level Biology practical work. Journal of Mathematics Science and Technology Education, 13(3), 723-740.
- Phillippi, J. and Lauderdale, J. (2018). A guide to field notes for qualitative research: Context and conversation. Qualitative Health Research, 28(3), 381-388.

- PubGenius Inc. (2024). What is good and bad academic performance? https://typeset.io/questions/what-is-good-&-bad-academic-performance-4p69tx5623
- QuestionPro. (2023). Sampling methods: Types with examples. <u>https://www.questionpro.com/blog/types-of-sampling-for-social-research/</u>
- QuestionPro Survey Software. (2022). Open-ended questions: Examples and advantages. https://www.questionpro.com/blog/what-are-open-ended-questions/
- Radha, R. (2021). Introduction to history-part 1. Alagappa University.
- Rehman, S. A., Sehar, S. and Afzal, M. (2019). Performance appraisal: Application of Victor Vroom Expectancy Theory. Saudi Journal of Nursing and Health Care, 2(12), 431-434. https://doi.org/10.36348/sjnhc.2019.v02i12.008
- Rind, I. A. and Mari, M. A. (2019). Analysing the impact of external examination on teaching and learning of English at the secondary level education. Cogent Education, 6, 1-14. https://doi.org/10.1080/2331186X.2019.1574947.
- Rizada, C. S. and Rey, P. P. (2023). Effects of using technology on the academic performance in mathematics of the College Millennial learners. East Asian Journal of Multidisciplinary Research, 2(6), 2495-2508. https://doi.org/10.55927/eajmr.r2i6.4055
- Rofiah, Q. (2022). Students perception towards drilling technique in speaking skill at Mathesa Institute Pare Kediri. [Bachelors' Degree Dissertation, State Islamic University].
- Roland Christensenet –al. (2013). Teacher incentives and student achievement: Evidence from New York city public schools. Harvest University and NBER.
- Rono, D. (2015). Constraints on effective Historyand government instruction in secondary schools in Bomet district, Rift valler province in Kenya. [Masters' Dissertation, Moi University].
- Ruslin, R., Alhabsyi, F., Mashuri, S., Rasak, M. S. A. and Syam, H. (2022). Semi-structured interview: A methodological reflection on the development of a quality research instrument in educational studies. Journal of Research and Method in Evaluation, 12(1), 22-29.
- Sabir, R. I., Majid, M. and Masood, K. (2021). Impact of students' wellness on academic performance with mediating role of learning environment. Academic Journal of Social Sciences, 5(4), 419-335.

- Scotland, J. (2012). Exploring the philosophical underpinnings of research: Relating ontology and epistemology to the methodology and methods of the scientific, interpretive and critical research paradigms. English Language Teaching, 5(9), 9-15. https://dx.doi.org/10.5539/elt.v5n9p9
- Sebbowa, D. K. and Ng'ambi, D. (2020). Teaching Historyin ways C21st students learning: A designed-based research perspective. International Journal of Learning Teaching and Educational Research, 19(9), 259-280. https://doi.org/10.2680/ijlter.19.914
- Sengai, W. (2021). Heads of department's role in implementation of Historysyllabi at selected Zimbabwean secondary schools: An instructional leadership perspective. Yesterday and today, (25), 1-30. https://dx.doi.org/10.17159/2223-0386/2021/n25a6
- Sengai, W. and Mokhele, M. L. (2021). Teachers' perceptions on development and implementation of History2167 syllabus in Zimbabwe. Cypriot Journal of Educational Science, 16(3), 916-927. https://doi.org/10.18844/cjes.v16c3.5763
- Sengai, W. and Mokhele, M. L. (2022). Examining the role of instructional resources in the implementation of the History2167 syllabus in Zimbabwe: A case study of 5 schools in the Glen View/Mufakose district. African Perspectives of Research in Teaching and Learning, 6(1), 119-131.
- Senguo, E. A. and Ilomo, O. O. (2020). Effect of school management on students' perceived academic achievement among Seventh-Day Adventist secondary schools in North-East Tanzania. East African Journal of Education and Social Sciences, 1(3), 105-110. https://doi.org/10.46606/eajiesszozordi03.0048
- Septibera, S. (2016). The use of drilling teaching to improving the sixth-grade students' speaking ability and active participation at Rung Arun Suka school Krabi, Thailand in the 2015/2016 academic year. University of Muhammadiyah.
- Shenton, A. (2004). Strategies for ensuring trustworthiness in qualitative research projects. Education for Information, 22, 63-75.
- Shin, J., Lee, Y. and Seo, E. (2017). The effects of feedback on students' achievement goals:

 Interaction between reference of comparison and regulatory focus. Learning and
 Instruction, 49, 21-31. https://doi.org/10.1016/j.learninstruc.2016.11.008

- Shing, C. L. (2015). The knowledge of teaching-pedagogical content knowledge (PCK). The Malaysian Journal of Educational Science, 3(3),40-55.
- Shout. (2020). Open-ended questions: Definition, example and tips. https://shout.com/survey-design/open-ended-questions/
- Simba, N. O. and Kabuka, E. (2016). Impact of discipline on academic performance of pupils in public primary schools in Muhoroni Sub-County, Kenya. Journal of Education and Practice, 7(6), 164-173. www.iiste.org
- Singh, S. K. (2021). Defining and measuring academic performance of Hei Students-A critical review. Turkish Journal of Computer and Mathematics Education, 12(6), 3091-3105. https://www.researchgate.net/publication/3590022367
- Straessele, J. M. W. (2014). Teaches' perspectives of effective lesson planning: A comparative analysis. [Masters' Dissertation, The College of William and Mary in Virginia].
- Strong, M. (2021). The art of lesson planning: Strategies for effective teaching. Routledge.
- StudySmarter. (2024). Perception research. https://www.studysmarter.uk/explanations/psychology/cognition/perceptio-research/
- Sulain, K. (2024). The impact of school management styles on students' performance: A case study of Mpanga secondary school Kabarole district. Metropolitan Journal of Social Educational Research, 3(2), 343-351.
- Sumner, K. Y. (2011). An explanatory mixed-methods study of instructional coaching practices and their relationship to students' achievement. [Doctoral Dissertation, Graduate School of Western Carolina University].
- Taherdoost, H. (2020). Sampling methods in research methodology: How to choose a sampling technique for research. International Journal of Academic Research in Management (IJARM), 5(2), 18-27.
- The Evergreen State College (2024). Understanding confidentiality and anonymity. https://.evergreed.edu/offices/institutional-review-board/confidentiality

- Thoka, B. M. (2020). The effects of motivation on student academic performance in Capricorn College for technical and vocational education and training. [Master' Dissertation, University of Limpopo].
- Tirado, A. (2021). Review of literature: Impact of coaching on student academic performance.

 Information Capsule Research, 2002, 1-6.
- Ullah, K., Badshah, S. and Qamar, H. (2018). Impact of teachers' attitudes on academic achievement of students in mathematics. Liberal Arts and Social Sciences International Journal, 2(2), 22-28. www.lassij.org
- *UNESCO.* (2017). *Television for learning: Our foremost tool in the* 21st *century. UNESCO.*
- University of Southern California. (2024). Research Guides: Organising your social sciences research assignments, writing field notes. https://libguides.usc.edu/writngguide/assignments/fieldnotes
- *Uwezo, K.* (2011). Are our children learning annual assessment report? Sage.
- Van Eeden, E. S. (2012). The youth and school History-learning from some of the thinking of yesterday in South Africa. Yesterday and Today, 8, 23-46.
- Vella, Y. and Caruana, R. (2015). Raising performance in Historyteaching while improving secondary students essay writing skills. International Journal of Historical Learning, Teaching and Research, 13.1(1), 118-131.
- Verrah, A. (2011). The effect of teaching aid on students' academic performance in English in selected schools in Rongo Division Rongo County. [Degree Dissertation, Kampala International University].
- Verrah, A. (2011). The effect of teaching and students' academic performance in English selected schools in Rongo Division Rongo County. [Bachelors' degree dissertation, Kampala International University].
- Vroom, V. H. (1964). Work and Motivation. Wiley.
- Wachira, E. (2021). A case study on the causes of underperformance in Historysubject in secondary schools in Meru County, Tigania West Sub County. [Bachelor's degree dissertation, Gretsa University].

- Wafulal, E. N., Kisilu, A. S., and Mukwa, C. W. (2019). Influence of score and coverage of the syllabus for physics for secondary schools in Kenya. International Journal of Scientific and Research Publications, 9(5), 866-873. https://dx.doi.org/10.29322/IJSRP.9.05.2019.p8xx
- Wagle, K. (2020). Research ethics: Definition, principles and advantages.

 https://www.publichealthnotes.com/research-ethics-definition-principles-and-advantages/
- Walker, R. J. (2018). Twelve characteristics of an effective teacher: A longitudinal, qualitative, quasi-research study of in-service and pre-service teachers' opinions. Educational Horizon, 61-68.
- Wangdi, N. and Zangmo, D. (2023). Inculcating the habit of revision to enhance students' learning outcomes. Universe International Journal of Interdisciplinary Research, 2(6), 66-76. https://www.doi-ds.org/doilink/12.2021-17341444UIJIR
- Wanyama, M. (2013). School-based factors influencing students' performance at Kenya Certificate Secondary Education in Narok-North District, Kenya. [Masters dissertation, University of Nairobi].
- Waseka, E. L. and Simatwa, E. M. (2016). Students' factors influencing academic performance of students in secondary education in Kenya: A case study of Kakamega County. International Research Journals, 7(3), 072-087. https://dx.doi.org/10.14303/er.2016.138
- Wasserman, J. (2010). Historyand culture, identity and heritage. The South African Society for HistoryTeaching (SASHT) Newsletter, 3, 17-20.
- West, J. and Meier, C. (2020). Overcrowded classrooms-the Achilles heel of South Africa education? South Africa Journal of Childhood Education, 10(1), 1-20, https://sajce.co.za/index.php/sajce/article/view/617/412
- Wilkesmann, Z. N., Steinmayr, A. and Fischer, K. (2021). Influence of motivation on academic performance of students in Germany. Journal of Education, 4(6), 1-9. https://doi.org/10.53819/810181025018

- Yulifar, L. and Aman, A. (2023). Resources of Historylearning in conventional and modern continuum lines. Cakrawala Pendidikan Journal Iimiah Pendidikan, 42(3), 586-6000. https://doi.org/10.21831/cp.v42i3.63713
- Zhang, S. and Huang, A. (2022). The long-term effects of automatic grade promotion on child development. China Economic Review, 74, 1-38. https://doi.org/10.1016/j.chieco
- Ziebell, N. and Clarke, D. (2018). Curriculum alignment: Performance types in the intended, enacted, and assessed curriculum in primary mathematics and science classrooms. Studia Paedagogica, 23(2), 175-204. https://doic.org/10.5817/SP2018-2-10

APPENDIX A

Open-ended questionnaires (research question 1)

1.	Background/general information
a)	How old are
	you?
b)	For how long have you been teaching History in this school or any other school?
c)	What is your highest teaching
	qualification?
d)	What are your subject majors?
e)	Do you have another position besides your teaching role? If yes, state it.
2.	How do you feel towards teaching of History and why?
3.	How many grade 11 History learners do you have this year?
	year:
4.	In comparison to prior years, what do you think could be the cause of decreasing or increasing of the number of grade 11
	learners?

5.	(a) Based on your opinions, is the performance of History poor or good?
	(b) Are you satisfied with the performance and why?
6.	In your opinions, what could be the reasons for this kind of performance?
7.	What do you think could happen with the subject if it is consistently performed poorly in public examinations?
8.	What do you think could be some benefits of History in national curriculum of Lesotho?

APPENDIX B

Semi-structured interviews (research questions 2 and 3)

- 1. What motivates you in the teaching of history, and why?
- 2. As a History teacher, do you practise lesson planning and scheming frequently? Can you explain how they benefit you and your learners?
- 3. How is your relationship with the principal, other teachers, and your learners? Why do you have such a relationship with them?
- 4. What kinds of teaching strategies are you using in the process of teaching, and why are you using such teaching strategies?
- 5. How often do you use teaching aids and teaching resources during teaching and learning? Can you explain why?
- 6. (a) Do you repeatedly give your learners assignments for weekly tests, and why? How often per week?
- (b) After how long do you give your learners feedback, and why?
- (c) Do you normally make corrections with your learners after giving them feedback? How do those corrections improve their performance?
 - 7. (a) What are the challenges your learners encounter in the learning of History and at school?
- (b) What do you think could be the solutions for the above challenges?
 - 8. (a) As a History teacher, what are the challenges you encounter in the teaching of History?
- (b) In your opinion, what do you think could be possible solutions for the above-mentioned challenges?
 - 9. What do you think could be done to improve the performance of History learners in public examinations?
 - 10. Is there any information you would like to add concerning the performance of History learners in public examinations?

APPENDIX C

Plagiarism report

NONYANA'S FINAL DISSERTATION.docx Thesis **Education** National University of Lesotho **Document Details** Submission ID 150 Pages trn:oid:::1:2995239803 48,485 Words Submission Date Sep 1, 2024, 4:42 PM GMT+2 292,603 Characters Download Date Sep 1, 2024, 4:48 PM GMT+2 NONYANA_S_FINAL_DISSERTATION.docx File Size 292.2 KB Turnitin Page 1 of 169 - Cover Page Submission ID trn:old::1:2995239803 Turnitin Page 2 of 169 - Integrity Overview Submission ID trn:old::1:2995239803 15% Overall Similarity The combined total of all matches, including overlapping sources, for each database. Filtered from the Report Bibliography Quoted Text **Match Groups Top Sources** 455Not Cited or Quoted 10% 14% @ Internet sources Matches with neither in-text citation nor quotation marks 5% III Publications 197Missing Quotations 5%

Matches that are still very similar to source material 5% __ Submitted works (Student Papers) Missing Citation 0%
 Matches that have quotation marks, but no in-text citation O Cited and Quoted 0%
 Matches with in-text citation present, but no quotation marks